Draft ETSI EN 301 908-13 V13.3.0 (2024-07)



IMT cellular networks;

Harmonised Standard for access to radio spectrum; Part 13: Evolved Universal Terrestrial Radio Access (E-UTRA) User Equipment (UE) 2

Reference

REN/MSG-TFES-1504

Keywords

3G, 3GPP, cellular, digital, E-UTRA, IMT, IMT-2000, IMT-Advanced, LTE, LTE-Advanced, mobile, radio, regulation, UMTS

ETSI

650 Route des Lucioles F-06921 Sophia Antipolis Cedex - FRANCE

Tel.: +33 4 92 94 42 00 Fax: +33 4 93 65 47 16

Siret N° 348 623 562 00017 - APE 7112B Association à but non lucratif enregistrée à la Sous-Préfecture de Grasse (06) N° w061004871

Important notice

The present document can be downloaded from: <u>https://www.etsi.org/standards-search</u>

The present document may be made available in electronic versions and/or in print. The content of any electronic and/or print versions of the present document shall not be modified without the prior written authorization of ETSI. In case of any existing or perceived difference in contents between such versions and/or in print, the prevailing version of an ETSI deliverable is the one made publicly available in PDF format at www.etsi.org/deliver.

Users of the present document should be aware that the document may be subject to revision or change of status. Information on the current status of this and other ETSI documents is available at <u>https://portal.etsi.org/TB/ETSIDeliverableStatus.aspx</u>

If you find errors in the present document, please send your comment to one of the following services: <u>https://portal.etsi.org/People/CommiteeSupportStaff.aspx</u>

If you find a security vulnerability in the present document, please report it through our Coordinated Vulnerability Disclosure Program: https://www.etsi.org/standards/coordinated-vulnerability-disclosure

Notice of disclaimer & limitation of liability

The information provided in the present deliverable is directed solely to professionals who have the appropriate degree of experience to understand and interpret its content in accordance with generally accepted engineering or other professional standard and applicable regulations.

No recommendation as to products and services or vendors is made or should be implied.

In no event shall ETSI be held liable for loss of profits or any other incidental or consequential damages.

Any software contained in this deliverable is provided "AS IS" with no warranties, express or implied, including but not limited to, the warranties of merchantability, fitness for a particular purpose and non-infringement of intellectual property rights and ETSI shall not be held liable in any event for any damages whatsoever (including, without limitation, damages for loss of profits, business interruption, loss of information, or any other pecuniary loss) arising out of or related to the use of or inability to use the software.

Copyright Notification

No part may be reproduced or utilized in any form or by any means, electronic or mechanical, including photocopying and microfilm except as authorized by written permission of ETSI.

The content of the PDF version shall not be modified without the written authorization of ETSI.

The copyright and the foregoing restriction extend to reproduction in all media.

© ETSI 2024. All rights reserved.

Contents

Intelle	ectual Property Rights	11
Forew	ord	11
Modal	l verbs terminology	12
Introd	uction	12
1	Scope	13
2	References	16
2.1	Normative references	16
2.2	Informative references	16
3	Definition of terms, symbols and abbreviations	18
3.1	Terms	18
3.2	Symbols	21
3.3	Abbreviations	23
4	Technical requirements specifications	25
4.1	Environmental profile	25
4.2	Conformance requirements	25
4.2.0	General	25
4.2.1	Introduction	25
4.2.2	Transmitter Maximum Output Power	
4.2.2.1	Transmitter maximum output power for Single Carrier	
4.2.2.1	1 Definition	
4.2.2.1	2 Limits	
4.2.2.1	3 Conformance	
4222	Transmitter output nower for Carrier Aggregation (DL CA and UL CA)	20 26
4222	1 Definition	20 26
4.2.2.2	2 Limits	20 27
4.2.2.2	2 Conformance	21 28
4.2.2.2	Transmitter output power for UL MIMO	20 20
4.2.2.3	1 Definition	20 20
4.2.2.3		20
4.2.2.3		
4.2.2.3	.3 Conformance	
4.2.2.4	Transmitter output power for category NB1	29
4.2.2.4	.1 Definition	29
4.2.2.4	.2 Limits	29
4.2.2.4	.3 Conformance	29
4.2.2.5	Transmitter output power for UE category M1	29
4.2.2.5	.1 Definition	29
4.2.2.5	.2 Limits	29
4.2.2.5	.3 Conformance	30
4.2.3	Transmitter Spectrum Emission Mask	30
4.2.3.1	Transmitter spectrum emission mask for Single Carrier	30
4.2.3.1	.1 Definition	30
4.2.3.1	.2 Limits	
4.2.3.1	.3 Conformance	
4.2.3.2	Transmitter spectrum emission mask for Carrier Aggregation (DL CA and UL CA)	
4.2.3.2	.1 Definition	
4.2.3.2	.2 Limits	
4.2.3 2	.3 Conformance	33
4.2.3.3	Transmitter spectrum emission mask for UL-MIMO	33
4233	1 Definition	
4722	2 Limits	2 <i>۸</i>
4929	2 Conformance	
+.2.3.3	Transmitter apostrum amiggion mode for Multi Cluster DUCCU within a commence to an	
4.2.3.4	1 Definition	
4.2.3.4		
4.2.3.4	.2 Limits	34

42242	Conformance	24
4.2.3.4.3	Comornance	
4.2.3.3	Definition	
4.2.3.3.1		
4.2.5.5.2	Carformana.	
4.2.3.3.3	ComornitanCe	
4.2.4	Transmitter spurious emissions for Single Continu	
4.2.4.1	Definition	
4.2.4.1.1		
4.2.4.1.2		
4.2.4.1.3	Conformance	
4.2.4.2	Transmitter spurious emissions for Carrier Aggregation (DL CA and UL CA)	40
4.2.4.2.1	Definition	40
4.2.4.2.2	Limits	40 45
4.2.4.2.3	Conformance	45
4.2.4.3	Transmitter spurious emissions for UL-MIMO	45
4.2.4.3.1	Definition	
4.2.4.3.2	Limits	
4.2.4.3.3	Conformance	45
4.2.4.4	Definition	45
4.2.4.4.1		45
4.2.4.4.2	Limits	45
4.2.4.4.3	Conformance	40
4.2.4.5	Transmitter spurious emissions for category NBT	40
4.2.4.5.1	Definition	40
4.2.4.5.2	Limits	46
4.2.4.5.3	Conformance	40
4.2.5	Transmitter Minimum Output Power.	40
4.2.5.1	I ransmitter minimum output power for Single Carrier	
4.2.5.1.1	Definition	40
4.2.5.1.2	Limits	40
4.2.5.1.3	Conformance	46
4.2.5.2	Transmitter minimum output power for Carrier Aggregation (DL CA and UL CA)	4/
4.2.5.2.1	Definition	4/
4.2.5.2.2	Limits	4/
4.2.5.2.3	Conformance	4/
4.2.5.3	I ransmitter minimum output power for UL-MIMO	47
4.2.5.3.1	Definition	47
4.2.5.3.2	Limits	4/
4.2.5.3.3	Conformance	48
4.2.5.4	Transmitter minimum output power for category NB1	48
4.2.5.4.1	Definition	48
4.2.5.4.2	Limits	48
4.2.5.4.3	Conformance	48
4.2.6	Receiver Adjacent Channel Selectivity (ACS)	48
4.2.0.1	Receiver Aujacent Unannel Selectivity (AUS) for Single Uarrier	
4.2.0.1.1		
4.2.6.1.2		
4.2.6.1.3		
4.2.6.2	Receiver Adjacent Channel Selectivity (ACS) for Carrier Aggregation in DL-only bands	
4.2.6.2.1	Definition	
4.2.6.2.2	Limits	
4.2.6.2.3	Conformance	
4.2.0.3	Receiver Aujacent Unannel Selectivity (AUS) for category NB1	
4.2.0.3.1		
4.2.6.3.2		
4.2.0.3.3	Conformance	
4.2.7	Receiver Blocking Characteristics for Single Constant	
4.2.7.1	Receiver Blocking Unaracteristics for Single Carrier	
4.2.7.1.1		
4.2.7.1.2		
4.2.7.1.3	Conformance	
4.2.1.2	Keceiver Blocking Unaracteristics for Carrier Aggregation in DL-only bands	

4.2.7.2.1	Definition	53
4.2.7.2.2	Limits	54
4.2.7.2.3	Conformance	56
4.2.7.3	Receiver Blocking Characteristics for category NB1	57
4.2.7.3.1	Definition	57
4.2.7.3.2	Limits	57
4.2.7.3.3	Conformance	58
4.2.8	Receiver Spurious Response	58
4281	Receiver Spurious Response for Single Carrier	58
1.2.0.1	Definition	58
42812	L'imits	59
4.2.8.1.2	Conformanca	50
4.2.8.1.5	Descriver Spurious Desponse for Corrier Aggregation in DL only hands	50
4.2.0.2	Definition	50
4.2.8.2.1		
4.2.8.2.2	Limits	
4.2.8.2.3	Conformance	59
4.2.8.3	Receiver Spurious Response for category NB1	59
4.2.8.3.1	Definition	59
4.2.8.3.2	Limits	60
4.2.8.3.3	Conformance	60
4.2.9	Receiver Intermodulation Characteristic	60
4.2.9.1	Receiver Intermodulation Characteristics for Single Carrier	60
4.2.9.1.1	Definition	60
4.2.9.1.2	Limits	60
4.2.9.1.3	Conformance	61
4.2.9.2	Receiver Intermodulation Characteristics for Carrier Aggregation in DL-only bands	61
4.2.9.2.1	Definition	61
4.2.9.2.2	Limits	61
4.2.9.2.3	Conformance	62
4293	Receiver Intermodulation Characteristics for category NB1	62
42931	Definition	62
42932	Limits	62
42933	Conformance	02 62
4 2 10	Receiver Spurious Emissions	63
4.2.10	Receiver Spurious Emissions for Single Carrier	63
4.2.10.1	Definition	05
4.2.10.1.1	Definition	05
4.2.10.1.2	Limits	05
4.2.10.1.5	Conformance	03
4.2.10.2	Receiver Spurious Emissions in DL-only bands	63
4.2.10.2.1		63
4.2.10.2.2	Limits	63
4.2.10.2.3	Conformance	63
4.2.11	Transmitter Adjacent Channel Leakage Power Ratio	64
4.2.11.1	Transmitter adjacent channel leakage power ratio for Single Carrier	64
4.2.11.1.1	Definition	64
4.2.11.1.2	Limits	64
4.2.11.1.3	Conformance	65
4.2.11.2	Transmitter adjacent channel leakage power ratio for Carrier Aggregation (DL CA and UL CA)	65
4.2.11.2.1	Definition	65
4.2.11.2.2	Limits for CA UTRA	66
4.2.11.2.3	Limits for CA EUTRA	67
4.2.11.2.4	Conformance	68
4.2.11.3	Transmitter adjacent channel leakage power ratio for UL-MIMO	68
4.2.11.3.1	Definition.	68
4.2.11.3.2	Limits	68
4.2.11.3.3	Conformance	69
4.2.11.4	Transmitter adjacent channel leakage power ratio for Multi-Cluster PUSCH within a component	
	carrier	
421141	Definition	69
421142	Limits	70
4 2 11 4 3	Conformance	70
4 2 11 5	Transmitter adjacent channel leakage power ratio for category NR1	70 70
7.2.11.J	Transmuter aujacent chamber leakage power ratio for category NDT	/ 0

4.2.11.5.1	Definition	70
4.2.11.5.2	Limits	70
4.2.11.5.3	Conformance	70
4.2.12	Receiver Reference Sensitivity Level	71
4.2.12.0	General	71
4.2.12.1	Receiver Reference Sensitivity Level for Single Carrier	71
4.2.12.1.1	Definition	
4 2 1 2 1 2	Limits	71
421212	Conformance	
4.2.12.1.	Desciver Deference Sensitivity Level for Corrier Aggregation in DL only hands	
4.2.12.2	Definition	12
4.2.12.2.1		12
4.2.12.2.2	Limits	72
4.2.12.2.3	Conformance	73
4.2.12.3	Receiver Reference Sensitivity Level for category NB1	73
4.2.12.3.1	Definition	73
4.2.12.3.2	Limits	73
4.2.12.3.3	Conformance	73
4.2.12.4	Receiver Reference Sensitivity Level for UE category 0	73
4.2.12.4.1	Definition	73
4.2.12.4.2	Limits	73
421243	Conformance	74
4 2 1 2 5	Receiver Reference Sensitivity Level for LIE category M1	74
4212.5	Definition	74 7/
4.2.12.5.1	Limito	
4.2.12.3.2	Canformence	14
4.2.12.5.3	Device of Text 1 Device in the Company of the Compa	15
4.2.13	Receiver Total Radiated Sensitivity (TRS)	/0
4.2.13.0	Applicability	/6
4.2.13.1	Definition	76
4.2.13.2	Limits	77
4.2.13.3	Conformance	77
4.2.14	Total Radiated Power (TRP)	77
4.2.14.0	Applicability	77
4.2.14.1	Definition	77
4.2.14.2	Limits	78
- -		-
5 Te	esting for compliance with technical requirements	79
5.1	Environmental conditions for testing	79
5.2	Void	79
5.3	Essential radio test suites	79
5.3.0	General	79
5.3.1	Transmitter Maximum Output Power	80
5.3.1.1	Transmitter maximum output power for Single Carrier	80
5.3.1.1.1	Method of test	80
53112	Test requirements	80
5312	Transmitter maximum output nower for intra-hand contiguous Carrier Aggregation (DL CA and	
5.5.1.2	III CA)	81
53121	Mathod of tast	01 Q1
5 2 1 2 2	Tast requirements	01 01
5.2.1.2.2	Test requirements	001
5.5.1.2A	Fransmitter maximum output power for inter-band Carrier Aggregation (DL CA and UL CA)	82
5.3.1.2A.	I Method of test	82
5.3.1.2A.	2 Test requirements	82
5.3.1.3	Transmitter maximum output power for UL-MIMO	83
5.3.1.3.1	Method of test	83
5.3.1.3.2	Test requirements	83
5.3.1.4	Transmitter maximum output power for category NB1	83
5.3.1.4.1	Method of Test	83
5.3.1.4.2	Test requirements	84
5.3.1.5	Transmitter maximum output power for UE category 0	84
53151	Method of test	8/
53157	Test requirements	+0 ۶۶
5316	Transmitter maximum output nowar for UE catagory M1	05 20
5.3.1.0 5.2.1.2.1	Mothed of test	0J 07
5.5.1.0.1	weiliou of test	

6

53162	Test requirements	85
5.3.2	Transmitter Spectrum Emission Mask	86
5.3.2.1	Transmitter spectrum emission mask for Single Carrier	86
53211	Method of test	
5.3.2.1.2	Test requirements	86
5.3.2.2	Transmitter spectrum emission mask for intra-band contiguous Carrier Aggregation (DL CA and	
	UL CA)	87
5.3.2.2.1	Method of test	87
5.3.2.2.2	Test requirements	87
5.3.2.2A	Transmitter spectrum emission mask for inter-band Carrier Aggregation (DL CA and UL CA)	88
5.3.2.2A.1	Method of test	88
5.3.2.2A.2	Test requirements	89
5.3.2.3	Transmitter spectrum emission mask for UL-MIMO	89
5.3.2.3.1	Method of test	89
5.3.2.3.2	Test requirements	89
5.3.2.4	Transmitter spectrum emission mask for Multi-Cluster PUSCH within a component carrier	90
5.3.2.4.1	Method of test	90
5.3.2.4.2	Test requirements	90
5.3.2.5	Transmitter spectrum emission mask for category NB1	91
5.3.2.5.1	Method of test	91
5.3.2.5.2	Test requirements	91
5.3.2.6	Transmitter spectrum emission mask for UE category 0	92
5.3.2.6.1	Method of test	92
5.3.2.6.2	Test requirements	92
5.3.2.7	Transmitter spectrum emission mask for UE category M1	92
5.3.2.7.1	Method of test	92
5.3.2.7.2	Test requirements	93
5.3.3	Transmitter Spurious Emissions	93
5.3.3.1	Transmitter spurious emissions for Single Carrier	93
5.3.3.1.1	Method of test	93
5.3.3.1.2	Test requirements	94
5.3.3.2	Transmitter spurious emissions for intra-band contiguous Carrier Aggregation (DL CA and UL	
	CA)	94
5.3.3.2.1	Method of test	() (
		94
5.3.3.2.2	Test requirements	94 95
5.3.3.2.2 5.3.3.2A	Test requirements Transmitter spurious emissions for inter-band Carrier Aggregation (DL CA and UL CA)	94 95 95
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1	Test requirements Transmitter spurious emissions for inter-band Carrier Aggregation (DL CA and UL CA) Method of test	94 95 95 95
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2	Test requirements Transmitter spurious emissions for inter-band Carrier Aggregation (DL CA and UL CA) Method of test Test requirements	94 95 95 95 96
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3	Test requirements Transmitter spurious emissions for inter-band Carrier Aggregation (DL CA and UL CA) Method of test Test requirements Transmitter spurious emissions for UL-MIMO	94 95 95 95 96 96
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3.1 5.3.3.3.1	Test requirements Transmitter spurious emissions for inter-band Carrier Aggregation (DL CA and UL CA) Method of test Test requirements Transmitter spurious emissions for UL-MIMO Method of test	94 95 95 96 96 96
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3.1 5.3.3.3.2	Test requirements Transmitter spurious emissions for inter-band Carrier Aggregation (DL CA and UL CA) Method of test Test requirements Transmitter spurious emissions for UL-MIMO Method of test Test requirements Test requirements	94 95 95 96 96 96 97
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3.1 5.3.3.3.2 5.3.3.4 5.2.2.4 1	Test requirements Transmitter spurious emissions for inter-band Carrier Aggregation (DL CA and UL CA) Method of test Test requirements Transmitter spurious emissions for UL-MIMO Method of test Test requirements Test requirements Transmitter spurious emissions for Multi-Cluster PUSCH within a component carrier	94 95 95 96 96 96 97 97
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3.1 5.3.3.3.2 5.3.3.4 5.3.3.4 5.3.3.4.1 5.3.3.4.1	Test requirements Transmitter spurious emissions for inter-band Carrier Aggregation (DL CA and UL CA) Method of test Transmitter spurious emissions for UL-MIMO Method of test Test requirements Transmitter spurious emissions for Multi-Cluster PUSCH within a component carrier Method of test Transmitter spurious emissions for Multi-Cluster PUSCH within a component carrier	94 95 95 96 96 96 96 97 97
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3.1 5.3.3.3.2 5.3.3.4 5.3.3.4.1 5.3.3.4.2 5.3.3.4.2	Test requirements Transmitter spurious emissions for inter-band Carrier Aggregation (DL CA and UL CA) Method of test Test requirements Transmitter spurious emissions for UL-MIMO Method of test Test requirements Transmitter spurious emissions for Multi-Cluster PUSCH within a component carrier Method of test Test requirements Test requirements	94 95 95 96 96 96 97 97 97 97
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3.1 5.3.3.3.2 5.3.3.4 5.3.3.4.1 5.3.3.4.1 5.3.3.4.2 5.3.3.5 5.3.3.5	Test requirements Transmitter spurious emissions for inter-band Carrier Aggregation (DL CA and UL CA) Method of test Test requirements Transmitter spurious emissions for UL-MIMO Method of test Test requirements Transmitter spurious emissions for Multi-Cluster PUSCH within a component carrier Method of test Test requirements Transmitter spurious emissions for category NB1	94 95 95 96 96 96 97 97 97 97 97
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3.1 5.3.3.3.2 5.3.3.4 5.3.3.4 5.3.3.4.1 5.3.3.4.2 5.3.3.4 5.3.3.5 5.3.3.5.1 5.3.3.5.1	Test requirements Transmitter spurious emissions for inter-band Carrier Aggregation (DL CA and UL CA) Method of test Test requirements Transmitter spurious emissions for UL-MIMO Method of test Test requirements Transmitter spurious emissions for Multi-Cluster PUSCH within a component carrier Method of test Test requirements Transmitter spurious emissions for category NB1 Method of test Transmitter spurious emissions for category NB1 Method of test	94 95 95 96 96 96 96 97 97 97 97 98 98
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3 5.3.3.3.1 5.3.3.3.2 5.3.3.4 5.3.3.4.1 5.3.3.4.2 5.3.3.5 5.3.3.5.1 5.3.3.5.1 5.3.3.5.2 5.3.3.6	Test requirements Transmitter spurious emissions for inter-band Carrier Aggregation (DL CA and UL CA) Method of test Test requirements Transmitter spurious emissions for UL-MIMO Method of test Test requirements Transmitter spurious emissions for Multi-Cluster PUSCH within a component carrier Method of test Test requirements Transmitter spurious emissions for category NB1 Method of test Transmitter spurious emissions for category NB1 Method of test Test requirements Transmitter spurious emissions for LUE category 0	94 95 95 96 96 96 96 97 97 97 97 98 98 98
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3.1 5.3.3.3.2 5.3.3.4 5.3.3.4.1 5.3.3.4.2 5.3.3.5 5.3.3.5.1 5.3.3.5.1 5.3.3.5.2 5.3.3.6 5.3.3.6	Test requirements Transmitter spurious emissions for inter-band Carrier Aggregation (DL CA and UL CA) Method of test Test requirements Transmitter spurious emissions for UL-MIMO Method of test Test requirements Transmitter spurious emissions for Multi-Cluster PUSCH within a component carrier Method of test Test requirements Transmitter spurious emissions for category NB1 Method of test Test requirements Transmitter spurious emissions for UE category 0 Method of test Transmitter spurious emissions for UE category 0	94 95 95 96 96 96 97 97 97 97 98 98 98 98 98
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3.1 5.3.3.3.2 5.3.3.4 5.3.3.4 5.3.3.4.1 5.3.3.4.2 5.3.3.5 5.3.3.5.1 5.3.3.5.2 5.3.3.6 5.3.3.6.1 5.3.3.6.1 5.3.3.6.2	Test requirements	94 95 95 96 96 96 97 97 97 97 98 98 98 98 99 99
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3 5.3.3.3.1 5.3.3.2 5.3.3.4 5.3.3.4 5.3.3.4 5.3.3.5 5.3.3.5.1 5.3.3.5.2 5.3.3.6 5.3.3.6.1 5.3.3.6.2 5.3.3.7	Test requirements	94 95 95 96 96 96 97 97 97 98 98 98 98 99 99 99
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3 5.3.3.3.1 5.3.3.2 5.3.3.4 5.3.3.4 5.3.3.4.1 5.3.3.5 5.3.3.5.1 5.3.3.5.2 5.3.3.6 5.3.3.6.1 5.3.3.6.2 5.3.3.7 5.3.3.7 5.3.3.7	Test requirements	94 95 95 96 96 96 97 97 97 97 98 98 98 98 99 99 99 99
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3 5.3.3.3.1 5.3.3.2 5.3.3.4 5.3.3.4 5.3.3.4.1 5.3.3.5 5.3.3.5.1 5.3.3.5.2 5.3.3.6 5.3.3.6.1 5.3.3.6.2 5.3.3.7 5.3.3.7.1 5.3.3.7.1 5.3.3.7.2	Test requirements	94 95 95 96 96 96 97 97 97 97 98 98 98 99 99 99 99 99 99
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3 5.3.3.3.1 5.3.3.3.2 5.3.3.4 5.3.3.4.1 5.3.3.4.2 5.3.3.5 5.3.3.5.1 5.3.3.6 5.3.3.6.1 5.3.3.6.1 5.3.3.6.2 5.3.3.7 5.3.3.7.1 5.3.3.7.2 5.3.4	Test requirements	94 95 95 96 96 96 97 97 97 98 98 98 98 99 99 99 99 99 99 90
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3 5.3.3.3.1 5.3.3.3.2 5.3.3.4 5.3.3.4.1 5.3.3.4.2 5.3.3.5 5.3.3.5.1 5.3.3.5.2 5.3.3.6 5.3.3.6.1 5.3.3.6.2 5.3.3.7 5.3.3.7.1 5.3.3.7.2 5.3.4 5.3.4.1	Test requirements	94 95 95 96 96 96 97 97 97 97 98 99 99 99 99 99 99 90 90 90
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3 5.3.3.3.1 5.3.3.3.2 5.3.3.4 5.3.3.4 5.3.3.4.1 5.3.3.5 5.3.3.5.1 5.3.3.6.1 5.3.3.6.2 5.3.3.7 5.3.3.7.1 5.3.3.7.1 5.3.3.7.2 5.3.4 5.3.4.1 5.3.4.1 5.3.4.1	Test requirements	94 95 95 96 96 96 97 97 97 97 97 98 98 98 99 99 99 99 99 90 90 90 90 90 90
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3 5.3.3.3.1 5.3.3.3.2 5.3.3.4 5.3.3.4 5.3.3.4.1 5.3.3.5 5.3.3.5.1 5.3.3.6.2 5.3.3.7 5.3.3.7 5.3.3.7.1 5.3.3.7.2 5.3.4 5.3.4.1 5.3.4.1 5.3.4.1 5.3.4.12	Test requirements Transmitter spurious emissions for inter-band Carrier Aggregation (DL CA and UL CA) Method of test Test requirements Transmitter spurious emissions for UL-MIMO Method of test Test requirements Transmitter spurious emissions for Multi-Cluster PUSCH within a component carrier Method of test Test requirements Transmitter spurious emissions for category NB1 Method of test Test requirements Transmitter spurious emissions for UE category 0 Method of test Test requirements Transmitter spurious emissions for UE category M1 Method of test Transmitter funimum output Power Transmitter Minimum Output Power Transmitter minimum output power for Single Carrier Method of test Test requirements	944 95 95 96 96 96 97 97 97 97 98 98 98 99 99 99 99 99 99 90 .100 .10
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3.1 5.3.3.3.2 5.3.3.4 5.3.3.4.1 5.3.3.5 5.3.3.5.1 5.3.3.5.2 5.3.3.6 5.3.3.6.1 5.3.3.6.2 5.3.3.7 5.3.3.7.1 5.3.3.7.2 5.3.4 5.3.4.1 5.3.4.1 5.3.4.1.1 5.3.4.1.2 5.3.4.2	Test requirements Transmitter spurious emissions for inter-band Carrier Aggregation (DL CA and UL CA) Method of test	944 95 95 96 96 97 97 97 97 98 98 98 99 99 99 99 .100 .100 .101
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3.1 5.3.3.3.2 5.3.3.4 5.3.3.4.1 5.3.3.4.2 5.3.3.5.1 5.3.3.5.2 5.3.3.6 5.3.3.6.1 5.3.3.6.2 5.3.3.7 5.3.3.7.1 5.3.3.7.2 5.3.4.1 5.3.4.1 5.3.4.1.2 5.3.4.2	Test requirements	94 95 95 96 96 96 97 97 97 97 97 98 98 98 98 99 99 99 99 .100 .100 .101 .101
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3 5.3.3.3.1 5.3.3.2 5.3.3.4 5.3.3.4 5.3.3.4.1 5.3.3.5.2 5.3.3.6 5.3.3.6.1 5.3.3.6.1 5.3.3.7 5.3.3.7 5.3.3.7.1 5.3.3.7.2 5.3.4 5.3.4.1 5.3.4.1 5.3.4.12 5.3.4.2 5.3.3.5 5.3.4 5.3.4.2 5.3.5 5.5 5	Test requirements	94 95 95 96 96 96 96 97 97 97 97 98 98 99 98 99 99 99 99 90 .100 .100
5.3.3.2.2 5.3.3.2A 5.3.3.2A.1 5.3.3.2A.2 5.3.3.3 5.3.3.3 5.3.3.3.1 5.3.3.2 5.3.3.4 5.3.3.4.1 5.3.3.4.2 5.3.3.5 5.3.3.5.1 5.3.3.5.2 5.3.3.6 5.3.3.6.1 5.3.3.6.2 5.3.3.7 5.3.3.7.1 5.3.3.7.2 5.3.4.1 5.3.4.1.1 5.3.4.1.1 5.3.4.1.2 5.3.4.2 5.3.4.2.1 5.3.4.2.1 5.3.4.2.2	Test requirements	94 95 95 96 96 96 96 97 97 97 97 97 98 99 99 99 99 99 90 .100 .10

5.3.4.2A.1	Method of test	
5.3.4.2A.2	Test requirements	103
5.3.4.3	Transmitter minimum output power for UL-MIMO	103
5.3.4.3.1	Method of test	103
5.3.4.3.2	Test requirements	104
5.3.4.4	Transmitter minimum output power for category NB1	104
5.3.4.4.1	Method of test	104
5.3.4.4.2	Test requirements	104
5.3.4.5	Transmitter minimum output power for UE category 0	105
5.3.4.5.1	Method of test	105
5.3.4.5.2	Test requirements	
5.3.4.6	Transmitter minimum output power for UE category M1	105
5.3.4.6.1	Method of test	105
5.3.4.6.2	Provide A linear Change I School in (ACS)	106
5.5.5	Receiver Adjacent Channel Selectivity (ACS)	106
5.5.5.1	Receiver Adjacent Channel Selectivity (ACS) for Single Carrier	106
5.3.5.1.1	Test requirements	100
5352	Receiver Adjacent Channel Selectivity (ΔCS) for Carrier Aggregation in DL only hands	107
53521	Method of test	107
53522	Test requirements	107
5353	Receiver Adjacent Channel Selectivity (ACS) for category NB1	108
5.3.5.3.1	Method of test	
5.3.5.3.2	Test requirements	
5.3.5.4	Receiver Adjacent Channel Selectivity (ACS) for UE category 0	
5.3.5.4.1	Method of test	110
5.3.5.4.2	Test requirements	110
5.3.5.5	Receiver Adjacent Channel Selectivity (ACS) for UE category M1	110
5.3.5.5.1	Method of test	110
5.3.5.5.2	Test requirements	111
5.3.6	Receiver Blocking Characteristics	112
5.3.6.1	Receiver Blocking Characteristics for Single Carrier	112
5.3.6.1.1	Method of test	112
5.3.6.1.2	Test requirements	
5.3.6.2	Receiver Blocking Characteristics for Carrier Aggregation in DL-only bands	
5.3.6.2.1	Method of test	114
5.3.0.2.2	People Ploaking Characteristics for actagory NP1	110 116
53631	Method of test	110 116
53632	Test requirements	110
5364	Receiver Blocking Characteristics for UE category 0	117
5.3.6.4.1	Method of test	
5.3.6.4.2	Test requirements	
5.3.6.5	Receiver Blocking Characteristics for UE category M1	118
5.3.6.5.1	Method of test	118
5.3.6.5.2	Test requirements	120
5.3.7	Receiver Spurious Response	120
5.3.7.1	Receiver Spurious Response for Single Carrier	120
5.3.7.1.1	Method of test	120
5.3.7.1.2	Test requirements	120
5.3.7.2	Receiver Spurious Response for Carrier Aggregation in DL-only bands	
5.3.7.2.1	Method of test	120
5.3.7.2.2	l est requirements	121
3.3./.5	Nothed of text	121
J.J./.J.1 5 2 7 2 2	Ivietiiou oi test	121
5.5.1.5.2 5371	1 est requirements	121
5.5.7.4 5 3 7 <i>1</i> 1	Method of test	122
5.5.7.4.1 53749	Test requirements	122
5375	Receiver Spurious Response for UF category M1	122
5.3.7.5.1	Method of test	
5.3.7.5.2	Test requirements	

5.3.8	Receiver Intermodulation Characteristics	123
5.3.8.1	Receiver Intermodulation Characteristics for Single Carrier	123
5.3.8.1.1	Method of test	123
5.3.8.1.2	Test requirements	123
5.3.8.2	Receiver Intermodulation Characteristics for Carrier Aggregation in DL-only bands	124
5.3.8.2.1	Method of test	124
5.3.8.2.2	Test requirements	125
5.3.8.3	Receiver Intermodulation Characteristics for category NB1	125
5.3.8.3.1	Test requirements	125
5.3.8.3.2	Test requirements	125
5.3.8.4	Receiver Intermodulation Characteristics for UE category 0	126
5.3.8.4.1	Method of test	126
5.3.8.4.2	Test requirements	126
5.3.8.5	Receiver Intermodulation Characteristics for UE category M1	126
5.3.8.5.1	Method of test	126
5.3.8.5.2	Test requirements	
5.3.9	Receiver Spurious Emissions.	
5.3.9.1	Receiver Spurious Emissions for Single Carrier	
5.3.9.1.1	Method of test	
53912	Test requirements	128
5392	Receiver Spurious Emissions in DL-only bands	128
53921	Method of test	128
53922	Test requirements	120
5393	Receiver Spurious Emissions for LIE category ()	129
53931	Method of test	129
53932	Test requirements	129
5394	Receiver Spurious Emissions for LIE category M1	129
53941	Method of test	129
53942	Test requirements	130
5395	Receiver Spurious Emissions for LIE category NB1	130
53951	Method of test	130
53952	Test requirements	130
5 3 10	Transmitter Adjacent Channel Leakage Power Patio	130
5 3 10 1	Transmitter Adjacent channel leakage nower ratio for Single Carrier	130
5 3 10 1 1	Method of test	130
5 3 10 1 2	Test requirements	130
5 3 10 2	Transmitter adjacent channel leakage nower ratio for intra-hand contiguous Carrier Aggregation	151
5.5.10.2	(DL CA and UL CA)	132
5 3 10 2 1	Method of test	132
5 3 10 2 2	Test requirements	132
5 3 10 2 4	Transmitter adjacent channel leakage nower ratio for inter hand Carrier Aggregation (DL CA	155
J.J.10.2A	and UL CA)	133
5 3 10 2 1	Mathad of tast	122
5.3.10.2A.1	Test requirements	134
5.3.10.2A.2	Transmitter adjacent channel leakage power ratio for III MIMO	124
5 3 10 3 1	Mathad of tast	124
5 3 10 3 2	Test requirements	134
5 3 10 4	Transmitter adjacent channel leakage power ratio for Multi Cluster DUSCH within a component	155
5.5.10.4	maismitter aujacent enamer leakage power ratio for Multi-Cluster POSCII within a component	125
5 3 10 4 1	Caller	125
5 2 10 4 2	Test requirements	135
5.2.10.4.2	Transmitter adiagant abannal laglage neuror ratio for actagory ND1	130
5.5.10.5 5.2.10.5.1	Mathad of test	120
3.3.10.3.1	Ivicialuu UI iesi	127
5.5.10.5.2	Transmitter adiagant abannal laglage neuror ratio for UE actors of	127
5.5.10.6	I ransmuter adjacent channel leakage power ratio for UE category 0	13/
5.5.10.6.1	Internod of test.	137
5.5.10.6.2	rest requirements	138
5.5.10.7	I ransmitter adjacent channel leakage power ratio for UE category M1	138
5.5.10.7.1	Ivietnou of test	138
5.3.10.7.2	rest requirements	139
5.3.11	Receiver Reference Sensitivity Level	139
5.3.11.1	Receiver Reference Sensitivity Level for Single Carrier	139

5.3.11.1.1	Method of test	139
5.3.11.1.2	Test requirements	139
5.3.11.2 R	eceiver Reference Sensitivity Level for Carrier Aggregation in DL-only bands	140
5.3.11.2.1	Method of test	140
5.3.11.2.2	Test requirements	141
5.3.11.3 R	eceiver Reference Sensitivity Level for category NB1	141
5.3.11.3.1	Method of test	141
5.3.11.3.2	Test requirements	141
5.3.11.4 R	eceiver Reference Sensitivity Level for UE category 0	141
5.3.11.4.1	Method of test	141
5.3.11.4.2	Test requirements	142
5.3.11.5 R	eceiver Reference Sensitivity Level for UE category M1	142
5.3.11.5.1	Method of test	142
5.3.11.5.2	Test requirements	143
5.3.12 Rece	iver Total Radiated Sensitivity (TRS)	143
5.3.12.1 N	fethod of test	143
5.3.12.1.1	Initial conditions	143
5.3.12.1.2	Procedure	143
5.3.12.1.3	Procedure, reverberation chamber method	143
5.3.12.2 T	est requirements	143
5.3.13 Tota	Radiated Power (TRP)	143
5.3.13.1 N	fethod of test	143
5.3.13.1.1	Initial conditions	143
5.3.13.1.2	Procedure	143
5.3.13.1.3	Procedure, reverberation chamber method	144
5.3.13.2 T	est requirements	144
Annex A (inform	native). Relationship between the present document and the essential	
	requirements of Directive 2014/53/EU	
Anney B (norms	ntive). Environmental profile	147
B.I General		147
B.1.1 Introduc	tion	147
B.1.2 Tempera	iture	147
B.1.3 Voltage		147
B.1.4 Test env	ironment	148
Annex C (inform	native): Maximum Measurement Uncertainty	
Annex D (inform	native): Bibliography	
Annex E (inforn	native): Change history	151
History		152

Intellectual Property Rights

Essential patents

IPRs essential or potentially essential to normative deliverables may have been declared to ETSI. The declarations pertaining to these essential IPRs, if any, are publicly available for **ETSI members and non-members**, and can be found in ETSI SR 000 314: "Intellectual Property Rights (IPRs); Essential, or potentially Essential, IPRs notified to ETSI in respect of ETSI standards", which is available from the ETSI Secretariat. Latest updates are available on the ETSI Web server (https://ipr.etsi.org/).

Pursuant to the ETSI Directives including the ETSI IPR Policy, no investigation regarding the essentiality of IPRs, including IPR searches, has been carried out by ETSI. No guarantee can be given as to the existence of other IPRs not referenced in ETSI SR 000 314 (or the updates on the ETSI Web server) which are, or may be, or may become, essential to the present document.

Trademarks

The present document may include trademarks and/or tradenames which are asserted and/or registered by their owners. ETSI claims no ownership of these except for any which are indicated as being the property of ETSI, and conveys no right to use or reproduce any trademark and/or tradename. Mention of those trademarks in the present document does not constitute an endorsement by ETSI of products, services or organizations associated with those trademarks.

DECTTM, **PLUGTESTSTM**, **UMTSTM** and the ETSI logo are trademarks of ETSI registered for the benefit of its Members. **3GPPTM** and **LTETM** are trademarks of ETSI registered for the benefit of its Members and of the 3GPP Organizational Partners. **oneM2MTM** logo is a trademark of ETSI registered for the benefit of its Members and of the oneM2M Partners. **GSM**[®] and the GSM logo are trademarks registered and owned by the GSM Association.

Foreword

This draft Harmonised European Standard (EN) has been produced by ETSI Technical Committee Mobile Standards Group (MSG), and is now submitted for the combined Public Enquiry and Vote phase of the ETSI Standardisation Request deliverable Approval Procedure.

For non-EU countries the present document may be used for regulatory (Type Approval) purposes.

The present document has been prepared under the Commission's standardisation request C(2015) 5376 final [i.9] to provide one voluntary means of conforming to the essential requirements of Directive 2014/53/EU on the harmonisation of the laws of the Member States relating to the making available on the market of radio equipment and repealing Directive 1999/5/EC [i.2].

Once the present document is cited in the Official Journal of the European Union under that Directive, compliance with the normative clauses of the present document given in table A-1 confers, within the limits of the scope of the present document, a presumption of conformity with the corresponding essential requirements of that Directive and associated EFTA regulations.

The present document is part 13 of a multi-part deliverable. Full details of the entire series can be found in part 1 [i.12].

This version of the harmonized standard includes the following revisions:

- Revision of the maximum output power requirements outlined in clause 4.2.2.
- Inclusion of requirements for band 41.
- Inclusion of Additional spurious emissions limits for frequency range 470 to 694 MHz to protect Broadband Public Protection and Disaster Relief (BB-PPDR) and Digital Terrestrial Television (DTT) operations.
- Inclusion of specific requirements for bands 72, 87, and 88 to protect Broadband Public Protection and Disaster Relief (BB-PPDR) and Digital Terrestrial Television (DTT) operations.

Proposed national transposition dates	
Date of latest announcement of this EN (doa):	3 months after ETSI publication
Date of latest publication of new National Standard or endorsement of this EN (dop/e):	6 months after doa
Date of withdrawal of any conflicting National Standard (dow):	18 months after doa

12

Modal verbs terminology

In the present document "shall", "shall not", "should", "should not", "may", "need not", "will", "will not", "can" and "cannot" are to be interpreted as described in clause 3.2 of the <u>ETSI Drafting Rules</u> (Verbal forms for the expression of provisions).

"must" and "must not" are NOT allowed in ETSI deliverables except when used in direct citation.

Introduction

The present document is part of a set of standards developed by ETSI and is designed to fit in a modular structure to cover all radio and telecommunications terminal equipment within the scope of the Radio Equipment Directive [i.2]. The present document is produced following the guidance in ETSI EG 203 336 [i.3] as applicable.

1 Scope

The present document applies to the following radio equipment type:

• User Equipment for Evolved Universal Terrestrial Radio Access (E-UTRA).

This radio equipment type is capable of operating in all or any part of the frequency bands given in tables from 1-1 through 1-5.

E-UTRA Band	Direction of UE transmission	E-UTRA operating bands	Related EC/ECC decision
1	Transmit	1 920 MHz to 1 980 MHz	[i 21] and [i 22]
l	Receive	2 110 MHz to 2 170 MHz	[1.21] and [1.22]
2	Transmit	1 710 MHz to 1 785 MHz	[i 10] and [i 20]
3	Receive	1 805 MHz to 1 880 MHz	[i. 19] and [i.20]
7	Transmit	2 500 MHz to 2 570 MHz	[: 24] and [: 25]
/	Receive	2 620 MHz to 2 690 MHz	[1.24] and [1.25]
0	Transmit	880 MHz to 915 MHz	
8	Receive	925 MHz to 960 MHz	[1.19] and [1.20]
00	Transmit	832 MHz to 862 MHz	
20	Receive	791 MHz to 821 MHz	[1.6] and [1.7]
	Transmit	3 410 MHz to 3 490 MHz	
22	Receive	3 510 MHz to 3 590 MHz	[1.26] and [1.27]
28	Transmit	703 MHz to 748 MHz	
(see note 6)	Receive	758 MHz to 803 MHz	[1.14] and [1.15]
	Transmit	452.5 MHz to 457.5 MHz	
31	Receive	462.5 MHz to 467.5 MHz	[i.16]
32	Transmit	N/A	
(see note 1) (see note 2)	Receive	1 452 MHz to 1 496 MHz	[i.17] and [i.18]
33	Transmit and Receive	1 900 MHz to 1 920 MHz	[i.22]
34	Transmit and Receive	2 010 MHz to 2 025 MHz	[i.22]
38	Transmit and Receive	2 570 MHz to 2 620 MHz	[i.24] and [i.25]
40	Transmit and Receive	2 300 MHz to 2 400 MHz	[i.23]
41 (note 7)	Transmit and Receive	2 496 MHz to 2 690 MHz	[i.24] and [i.25]
42	Transmit and Receive	3 400 MHz to 3 600 MHz	[i.26] and [i.27]
43	Transmit and Receive	3 600 MHz to 3 800 MHz	[i.26] and [i.27]
46 (see note 3) (see note 4)	Transmit and Receive	5 150 MHz to 5 925 MHz	[i.29] and [i.30]
65	Transmit	1 920 MHz to 2 010 MHz	[i 21] [i 22] and [i 28]
(see note 5)	Receive	2 110 MHz to 2 200 MHz	[i.2 i]; [i.22] and [i.20]
67	Transmit	N/A	
07	Receive	738 MHz to 758 MHz	[i.14] and [i.15]
68	Transmit	698 MHz to 728 MHz	[i 14] and [i 15]
00	Receive	753 MHz to 783 MHz	[i. 14] and [i. 15]
69	Transmit	N/A	
(see note 1)	Receive	2 570 MHz to 2 620 MHz	[i.24] and [i.25]
72	Transmit	451 MHz to 456 MHz	[: 16]
	Receive	461 MHz to 466 MHz	[1. 10]
87	Transmit	410 MHz to 415 MHz	[: 40]
	Receive	420 MHz to 425 MHz	[1.16]
88	Transmit	412 MHz to 417 MHz	E 401
	Receive	422 MHz to 427 MHz	[1.16]

Table 1-1: E-UTRA UE operating bands

NOTE 1:	Restricted to E-UTRA operation when carrier aggregation is configured. The downlink operating band is paired with the uplink operating band (external) of the carrier aggregation configuration that is supporting the configured Pcell.
NOTE 2:	In Europe, according to [i.17] and [i.18], radio equipment in band 32 operates between 1 452 MHz and 1 492 MHz.
NOTE 3:	This band is an unlicensed band restricted to licensed-assisted operation using Frame Structure Type 3. In Europe according to [i.29] and [i.30], radio equipment in band 46 operates between 5 150 MHz and 5 725 MHz as in table 1-1A.
NOTE 4:	In this version of the present document, restricted to E-UTRA DL operation when carrier aggregation is configured.
NOTE 5:	A UE that complies with the E-UTRA Band 65 minimum requirements in the present document also complies with the E-UTRA Band 1 minimum requirements. This band includes two frequency ranges that are harmonised in Europe:
	 a) According to [i.21] and [i.22], radio equipment in band n65 operates between 2 110 MHz and 2 170 MHz for the transmitter (F_{DL_low} = 2 110 MHz and F_{DL_high} = 2 170 MHz), and between 1 920 MHz and 1 980 MHz for the receiver (F_{UL low} = 1 920 MHz and F_{UL high} = 1 980 MHz).
	b) Based on [i.29], radio equipment in band n65 operates between 2 170 MHz and 2 200 MHz for the transmitter (F _{DL_low} = 2 170 MHz and F _{DL_high} = 2 200 MHz) and between 1 980 MHz and 2 010 MHz for the receiver (F _{UL_low} = 1 980 MHz and F _{UL_high} = 2 010 MHz) as the Complementary Ground Component (CGC) of a Mobile-satellite service by reference to the present Harmonised Standard.
NOTE 6:	In Europe, according to [i.14], [i.15] and [i.16], radio equipment in band 28 operates between 703 MHz to 736 MHz for the transmitter ($F_{UL_low} = 703$ MHz and $F_{UL_high} = 736$ MHz) and between 758 MHz to 791 MHz for the receiver ($F_{DL_low} = 758$ MHz and $F_{DL_high} = 791$ MHz).
NOTE 7:	In Europe according to [i.24] and [i.25], radio equipment in band 41 operates between 2 500 MHz and 2 570 MHz ($F_{DL_low} = 2500$ MHz and $F_{DL_high} = 2570$ MHz).

NOTE 1: The relationship between the present document and essential requirements of article 3.2 of Directive 2014/53/EU [i.2] is given in annex A.

Table 1-1A: Sub-bands for band 46

	E-UTRA Band
	46a
	46b
	46c
NOTE:	The sub-bands 46a and 46b are restricted to
	indoor use only.

Table 1-2: E-UTRA UE Intra-band contiguous C	A operating bands
--	-------------------

E-UTRA CA Band
CA_1
CA_3
CA_7
CA_38
CA_40
CA_41
CA_42

E-UTRA CA Band
CA_1-3
CA_1-7
CA_1-8
CA_1-20
CA_1-41
CA_1-42
CA_1-46
CA_3-7
CA_3-8
CA_3-20
CA_3-28
CA_3-41
CA_3-42
CA_3-46
CA_7-20
CA_7-28
CA_7-46
CA_8-20
CA_8-40
CA_8-41
CA_20-32
CA_41-42
CA_41-46
CA_42-46
CA_20-67

 Table 1-3: E-UTRA UE Inter-band CA operating bands (two bands)

Table 1-4: E-UTRA UE Inter-band CA operating bands (three bands)

E-UTRA CA Band
CA_1-3-8
CA_1-3-20
CA_1-7-20
CA_3-7-20
CA_3-41-42

Table 1-5: Intra-band non-contiguous CA operating bands (with two sub-blocks)

E-UTRA CA Band
CA_3-3
CA_7-7
CA 41-41
CA_42-42

E-UTRA NB-IoT is designed to operate in the E-UTRA operating bands 1, 3, 8, 20, 28 and 65 defined in table 1-1. The present document covers requirements for E-UTRA FDD and E-UTRA TDD User Equipment from 3GPPTM Releases 8, 9, 10, 11, 12, and 13 defined in ETSI TS 136 101 [3]. This includes the requirements for E-UTRA UE operating bands and E-UTRA CA operating bands from 3GPPTM Release 13 defined in ETSI TS 136 101 [3].

NOTE 2: For Band 20:

- For user equipment designed to be mobile or nomadic, the requirements in the present document measured at the antenna port also show conformity to the corresponding requirement defined as Total Radiated Power (TRP), as described in Commission Decision 2010/267/EU [i.6] and ECC Decision (09)03 [i.7].
- For user equipment designed to be fixed or installed, the present document does not address the requirements described in Commission Decision 2010/267/EU [i.6] and ECC Decision (09)03 [i.7].

The present document contains requirements to demonstrate that radio equipment both effectively uses and supports the efficient use of radio spectrum in order to avoid harmful interference.

16

2 References

2.1 Normative references

References are either specific (identified by date of publication and/or edition number or version number) or nonspecific. For specific references, only the cited version applies. For non-specific references, the latest version of the referenced document (including any amendments) applies.

Referenced documents which are not found to be publicly available in the expected location might be found at https://docbox.etsi.org/Reference/.

NOTE: While any hyperlinks included in this clause were valid at the time of publication, ETSI cannot guarantee their long term validity.

The following referenced documents are necessary for the application of the present document.

- [1] <u>ETSI TS 136 521-1 (V17.7.0) (07-2023)</u>: "LTE; Evolved Universal Terrestrial Radio Access (E-UTRA); User Equipment (UE) conformance specification; Radio transmission and reception; Part 1: Conformance testing (3GPP TS 36.521-1 version 17.7.0 Release 17)".
- [2] <u>ETSI TS 136 508 (V17.6.0) (07-2023)</u>: "LTE; Evolved Universal Terrestrial Radio Access (E-UTRA) and Evolved Packet Core (EPC); Common test environments for User Equipment (UE) conformance testing (3GPP TS 36.508 version 17.6.0 Release 17)".
- [3] <u>ETSI TS 136 101 (V13.25.0) (07-2023)</u>: "LTE; Evolved Universal Terrestrial Radio Access (E-UTRA); User Equipment (UE) radio transmission and reception (3GPP TS 36.101 version 13.25.0 Release 13)".
- [4] IEC 60068-2-1:2007 ED 6.0: "Environmental testing Part 2-1: Tests Test A: Cold".
- [5] <u>IEC 60068-2-2:2007 ED 5.0</u>: "Environmental testing Part 2-2: Tests Test B: Dry heat".
- [6] ETSI TS 137 544 (V16.3.0) (05-2024): "Universal Mobile Telecommunications System (UMTS); LTE; Universal Terrestrial Radio Access (UTRA) and Evolved UTRA (E-UTRA); User Equipment (UE) Over The Air (OTA) performance; Conformance testing (3GPP TS 37.544 version 16.3.0 Release 16)".

2.2 Informative references

References are either specific (identified by date of publication and/or edition number or version number) or non-specific. For specific references, only the cited version applies. For non-specific references, the latest version of the referenced document (including any amendments) applies.

NOTE: While any hyperlinks included in this clause were valid at the time of publication, ETSI cannot guarantee their long term validity.

The following referenced documents are not necessary for the application of the present document but they assist the user with regard to a particular subject area.

- [i.1] Void.
- [i.2] <u>Directive 2014/53/EU</u> of the European Parliament and of the Council of 16 April 2014 on the harmonisation of the laws of the Member States relating to the making available on the market of radio equipment and repealing Directive 1999/5/EC.
- [i.3] ETSI EG 203 336 (V1.2.1) (05-2020): "Guide for the selection of technical parameters for the production of Harmonised Standards covering article 3.1(b) and article 3.2 of Directive 2014/53/EU".

- [i.4] Recommendation ITU-R SM.329-12 (2012): "Unwanted emissions in the spurious domain".
- [i.5] ETSI TR 100 028 (all parts) (V1.4.1): "Electromagnetic compatibility and Radio spectrum Matters (ERM); Uncertainties in the measurement of mobile radio equipment characteristics".
- [i.6] <u>Commission Decision 2010/267/EU of 6 May 2010</u> on harmonised technical conditions of use in the 790-862 MHz frequency band for terrestrial systems capable of providing electronic communications services in the European Union.
- [i.7] <u>ECC Decision (09)03 of 30 October 2009</u> on harmonised conditions for mobile/fixed communications networks (MFCN) operating in the band 790 862 MHz.
- [i.8] Void.
- [i.9] Commission Implementing Decision C(2015) 5376 final of 4.8.2015 on a standardisation request to the European Committee for Electrotechnical Standardisation and to the European Telecommunications Standards Institute as regards radio equipment in support of Directive 2014/53/EU of the European Parliament and of the Council.
- [i.10] ETSI TS 136 509 (V10.3.0) (09-2014): "LTE; Evolved Universal Terrestrial Radio Access (E-UTRA) and Evolved Packet Core (EPC); Special conformance testing functions for User Equipment (UE) (3GPP TS 36.509 version 10.3.0 Release 10)".
- [i.11] Void.
- [i.12] ETSI EN 301 908-1 (V13.1.1): "IMT cellular networks; Harmonised Standard for access to radio spectrum; Part 1: Introduction and common requirements".
- [i.13] ETSI TR 125 914 (V15.0.1) (09-2018): "Universal Mobile Telecommunications System (UMTS); Measurements of radio performances for UMTS terminals in speech mode (3GPP TR 25.914 version 15.0.1 Release 15)".
- [i.14] <u>Commission Implementing Decision (EU) 2016/687 of 28 April 2016</u> on the harmonisation of the 694-790 MHz frequency band for terrestrial systems capable of providing wireless broadband electronic communications services and for flexible national use in the Union.
- [i.15] <u>ECC Decision (15)01</u>: "Harmonised technical conditions for mobile/fixed communications networks (MFCN) in the band 694-790 MHz including a paired frequency arrangement (Frequency Division Duplex 2x30 MHz) and an optional unpaired frequency arrangement (Supplemental Downlink)", Approved 06 March 2015.
- [i.16] <u>ECC Decision (19)02</u>: "Land mobile systems in the frequency ranges 68-87.5 MHz, 146-174 MHz, 406.1-410 MHz, 410-430 MHz, 440-450 MHz and 450-470 MHz", approved 8 March 2019.
- [i.17] Commission Implementing Decision (EU) 2018/661 of 26 April 2018 amending Implementing Decision (EU) 2015/750 on the harmonisation of the 1452-1492 MHz frequency band for terrestrial systems capable of providing electronic communications services in the Union as regards its extension in the harmonised 1427-1452 MHz and 1492-1517 MHz frequency bands.
- [i.18]ECC Decision (13)03: "The harmonised use of the frequency band 1 452-1 492 MHz for
Mobile/Fixed Communications Networks Supplemental Downlink (MFCN SDL)".
- [i.19] <u>Commission Implementing Decision (EU) 2022/173 of 7 February 2022</u> on the harmonisation of the 900 MHz and 1800 MHz frequency bands for terrestrial systems capable of providing electronic communications services in the Union and repealing Decision 2009/766/EC.
- [i.20] <u>ECC Decision (06)13</u>: "Designation of the bands 880-915 MHz, 925-960 MHz, 1710-1785 MHz and 1805-1880 MHz for terrestrial UMTS, LTE, WiMAX and IoT cellular systems", Approved 01 December 2006, Amended 8 March 2019.
- [i.21]Commission Implementing Decision (EU) 2020/667 of 6 May 2020 amending Decision
2012/688/EU as regards an update of relevant technical conditions applicable to the frequency
bands 1 920-1 980 MHz and 2 110-2 170 MHz.

[i.22]	ECC Decision (06)01: The harmonised utilisation of the bands1920-1980 MHz and 2110-2170 MHz for mobile/fixed communications networks (MFCN) including terrestrial IMT systems, Approved 24 March 2006, Amended 8 March 2019.
[i.23]	ECC Decision (14)02: "Harmonised technical and regulatory conditions for the use of the band 2300-2400 MHz for Mobile/Fixed Communications Networks (MFCN)", Approved 27 June 2014.
[i.24]	<u>Commission Implementing Decision (EU) 2020/636 of 8 May 2020</u> amending Decision 2008/477/EC as regards an update of relevant technical conditions applicable to the 2 500-2 690 MHz frequency band.
[i.25]	ECC Decision (05)05: "Harmonised utilization of spectrum for Mobile/Fixed Communications Networks (MFCN) operating within the band 2500-2690 MHz", Approved 18 March 2005, Amended 05 July 2019.
[i.26]	Commission implementing Decision (EU) 2019/235 of 24 January 2019 on amending Decision 2008/411/EC as regards an update of relevant technical conditions applicable to the 3 400-3 800 MHz frequency band.
[i.27]	ECC Decision (11)06: "Harmonised frequency arrangements and least restrictive technical conditions (LRTC) for mobile/fixed communications networks (MFCN) operating in the band 3400-3800 MHz" Approved 09 December 2011, Amended 26 October 2018.
[i.28]	ECC Decision (06)09: "Designation of the bands 1980-2010 MHz and 2170-2200 MHz for use by systems in the Mobile-Satellite Service including those supplemented by a Complementary Ground Component (CGC)", Approved 01 December 2006, Amended 05 September 2007.
[i.29]	<u>Commission Implementing Decision (EU) 2022/179 of 8 February 2022</u> on the harmonised use of radio spectrum in the 5 GHz frequency band for the implementation of wireless access systems including radio local area networks and repealing Decision 2005/513/EC.
[i.30]	ECC Decision (04)08: "On the harmonised use of the 5 GHz frequency bands for Wireless Access Systems including Radio Local Area Networks (WAS/RLAN)". Approved 09 July 2004, latest amended 1 July 2022.

3 Definition of terms, symbols and abbreviations

3.1 Terms

For the purposes of the present document, the following terms apply:

aggregated Channel Bandwidth: RF bandwidth in which a UE transmits and receives multiple contiguously aggregated carriers

aggregated Transmission Bandwidth Configuration: number of resource block allocated within the aggregated channel bandwidth

carrier aggregation: aggregation of two or more component carriers in order to support wider transmission bandwidths

carrier aggregation band: set of one or more operating bands across which multiple carriers are aggregated with a specific set of technical requirements

18

carrier aggregation bandwidth class: class defined by the aggregated transmission bandwidth configuration and maximum number of component carriers supported by a UE

CA Bandwidth Class	Aggregated Transmission Bandwidth Configuration	Number of contiguous CC	Nominal Guard Band BW _{GB}
A	N _{RB,agg} ≤ 100	1	$a_1 BW_{Channel(1)} - 0,5\Delta f_1$ (note 2)
В	N _{RB,agg} ≤ 100	2	0,05 <i>max</i> (BW _{Channel(1)} ,BW _{Channel(2)}) - 0,5∆f ₁
С	100 < N _{RB,agg} ≤ 200	2	0,05 <i>max</i> (BW _{Channel(1)} , BW _{Channel(2)}) - 0,5∆f ₁
NOTE 1: BW _{Channel(j)}	, j = 1, 2, 3, is the channel bandwid	dth of an E-UTRA compo	onent carrier according to ETSI
TS 136 521	-1 [1], table 5.4.2-1 and $\Delta f_1 = \Delta f$ fo	r the downlink with Δf the	e subcarrier spacing while $\Delta f_1 = 0$ for
the uplink.			
NOTE 2: $a_1 = 0,16/1,$	4 for BW _{Channel(1)} = 1,4 MHz where	eas $a_1 = 0,05$ for all othe	er channel bandwidths.

Table 3.1-1: CA bandwidth classes and corresponding nominal guard bands

carrier aggregation configuration: combination of CA operating band(s) and CA bandwidth class(es) supported by a UE

channel bandwidth: RF bandwidth supporting a single E-UTRA RF carrier with the transmission bandwidth configured in the uplink or downlink of a cell

- NOTE 1: The channel bandwidth is measured in MHz and is used as a reference for transmitter and receiver RF requirements.
- NOTE 2: Channel Bandwidth and Transmission Bandwidth Configuration for one E UTRA carrier are described in figure 3.1-1.



Figure 3.1-1: Channel Bandwidth and Transmission Bandwidth Configuration for one E-UTRA carrier

channel bandwidth for carrier aggregation: RF bandwidth aggregated from more than one E-UTRA RF carrier with the transmission bandwidth configured in the uplink or downlink of different cells

NOTE 1: Aggregated channel bandwidth and aggregated channel bandwidth edges for more than one E-UTRA carrier are described in figure 3.1-2 as in ETSI TS 136 101 [3].



Figure 3.1-2: Aggregated channel bandwidth and aggregated channel bandwidth edges for more than one E-UTRA carrier

Figure 3.1-3 shows the relation between the category NB1 channel bandwidth (BW_{Channel}) and the category NB1 transmission bandwidth configuration (N_{tone}). The channel edges are defined as the lowest and highest frequencies of the carrier separated by the channel bandwidth, i.e. at $F_C \pm BW_{Channel} / 2$.

NOTE 2: Channel bandwidth and transmission bandwidth for category NB1 are described in figure 3.1-3 as in ETSI TS 136 101 [3].





channel edge: lowest and highest frequency of the carrier, separated by the channel bandwidth

contiguous carriers: set of two or more carriers configured in a spectrum block where there are no RF requirements based on co-existence for un-coordinated operation within the spectrum block

21

inter-band carrier aggregation: carrier aggregation of component carriers in different operating bands

Carriers aggregated in each band can be contiguous or non-contiguous. NOTE:

intra-band contiguous carrier aggregation: contiguous carriers aggregated in the same operating band

intra-band non-contiguous carrier aggregation: non-contiguous carriers aggregated in the same operating band

Maximum Output Power (MOP): mean power level per carrier of UE measured at the antenna connector in a specified reference condition

mean power: power measured in the operating system bandwidth of the carrier when applied to E-UTRA transmissions

The period of measurement is assumed to be at least one subframe (1 ms) unless otherwise stated. NOTE:

network signalled value: signalling value sent from the BS to the UE to indicate additional unwanted emission requirements to the UE

occupied bandwidth: width of a frequency band such that, below the lower and above the upper frequency limits, the mean powers emitted are each equal to a specified percentage $\beta/2$ of the total mean power of a given emission

operating band: frequency range (paired or unpaired) that is defined with a specific set of technical requirements, in which E-UTRA operates

NOTE: Operating bands for E-UTRA are designated with Arabic numerals, while the corresponding operating bands for UTRA are designated with Roman numerals.

output power: mean power of one carrier of the UE, delivered to a load with resistance equal to the nominal load impedance of the transmitter

reference bandwidth: bandwidth in which an emission level is specified

resource block: physical resource consisting of a number of symbols in the time domain and a number of consecutive subcarriers spanning 180 kHz in the frequency domain

sub-block: one contiguous allocated block of spectrum for transmission and reception by the same UE, in which there may be multiple instances of sub-blocks within an RF bandwidth

transmission bandwidth: bandwidth of an instantaneous transmission from a UE or BS, measured in Resource Block units

See figure 3.1-1. NOTE:

transmission bandwidth configuration: highest transmission bandwidth allowed for uplink or downlink in a given channel bandwidth, measured in Resource Block units

NOTE: See figure 3.1-1.

transmit diversity: transmit diversity is based on space-frequency block coding techniques complemented with frequency-shift time diversity when four transmit antennas are used

3.2 **Symbols**

For the purposes of the present document, the following symbols apply:

Δf_{OOB}	Δ Frequency of out-of-band emission
BW _{Channel}	Channel bandwidth
BW _{Channel_CA}	Aggregated channel bandwidth, expressed in MHz
BW _{GB}	Virtual guard band to facilitate transmitter (receiver) filtering above/below edge CCs

BW _{Interferer}	Channel Bandwidth of the interferer
E_{RS}	Transmitted energy per RE for reference symbols during the useful part of the symbol,
	i.e. excluding the cyclic prefix, (average power normalized to the subcarrier spacing) at the eNode B transmit antenna connector
\hat{E}_{s}	The received energy per RE during the useful part of the symbol, i.e. excluding the cyclic prefix,
	averaged across the allocated RB(s) (average power within the allocated RB(s)), divided by the number of RE within this allocation and normalized to the subcarrier spacing) at the UE antenna connector
BW _{UTRA}	Channel Bandwidth UTRA
F	Frequency
$F_{Interferer}$ (offset)	Frequency offset of the interferer
F _{Interferer}	Frequency of the interferer
F _{Ioffset}	Frequency offset of the interferer
F _C	The centre for success of the langest energies energies of the MILE
F _{CA_low}	The centre frequency of the <i>lishest carrier</i> , expressed in MHZ
^F CA_high	The centre frequency of the <i>downlink</i> operating hand
F DL_low	The highest frequency of the downlink operating band
DL_high	The lowest frequency of the uplink operating band
Llow	The highest frequency of the uplink operating band
F UL_high	The <i>lower edge</i> of aggregated channel bandwidth, expressed in MHz
F edge_low	The <i>higher edge</i> of aggregated channel bandwidth, expressed in MHz
F _{offset_NS_23}	Frequency offset in MHz needed if NS_23 is used
I_o	The power spectral density of the total input signal (power averaged over the useful part of the
	symbols within the transmission bandwidth configuration, divided by the total number of RE for this configuration and normalized to the subcarrier spacing) at the UE antenna connector, including the own-cell downlink signal or the power spectral density of the total input signal at the UE antenna connector (power averaged over the useful part of the symbols within a given bandwidth and normalized to the said bandwidth), including the own-cell downlink signal
I _{or}	The total transmitted power spectral density of the own-cell downlink signal (power averaged over
	the useful part of the symbols within the transmission bandwidth configuration, divided by the total number of RE for this configuration and normalized to the subcarrier spacing) at the eNode B transmit antenna connector
\hat{I}_{or}	The total received power spectral density of the own-cell downlink signal (power averaged over
-	the useful part of the symbols within the transmission bandwidth configuration, divided by the total number of RE for this configuration and normalized to the subcarrier spacing) at the UE antenna connector
I_{ot}	The received power spectral density of the total noise and interference for a certain RE (average
	power obtained within the RE and normalized to the subcarrier spacing) as measured at the UE
L _{CRB}	Transmission bandwidth which represents the length of a contiguous resource block allocation expressed in units of resources blocks
N_{oc}	The power spectral density of a white noise source (average power per RE normalized to the
N	subcarrier spacing), simulating interference from cells that are not defined in a test procedure, as measured at the UE antenna connector The power spectral density of a white poise source (average power per RE normalized to the
N a	subcarrier spacing), simulating interference in non-CRS symbols in ABS subframe from cells that are not defined in a test procedure, as measured at the UE antenna connector The power spectral density of a white noise source (average power per RE normalized to the
· '02	subcarrier spacing), simulating interference in CRS symbols in ABS subframe from all cells that are not defined in a test procedure, as measured at the UE antenna connector

N_{oc3}	The power spectral density of a white noise source (average power per RE normalized to the
	subcarrier spacing), simulating interference in non-ABS subframe from cells that are not defined
Name	In a test procedure, as measured at the UE antenna connector Offset used for calculating downlink EARECN
Nom w	Offset used for calculating unlink FARECN
Noffs-UL	Transmission bandwidth configuration expressed in units of resource blocks
N _R B	Aggregated Transmission Bandwidth Configuration The number of the aggregated RBs within the
RB_agg	fully allocated Aggregated Channel handwidth
N _{tona}	Transmission bandwidth configuration for category NB1, expressed in units of tones
N _{tone 3.75 kHz}	Transmission bandwidth configuration for category NB1 with 3,75 kHz sub-carrier spacing,
	expressed in units of tones
N _{tone 15 kHz}	Transmission bandwidth configuration for category NB1 with 15 kHz sub-carrier spacing,
	expressed in units of tones
N _{UL}	Uplink EARFCN
NS_x	Network signalled value "x"
Р	Number of cell-specific antenna ports
p	Antenna port number
PInterferer	Modulated mean power of the interferer
P _{UMAX}	Maximum UE Power with possible power reduction due to modulation type, network signalling
	values and location near the edge of the band
R_{av}	Minimum average throughput per RB

3.3 Abbreviations

For the purposes of the present document, the following abbreviations apply:

Access Channel
Adjacent Channel Leakage Ratio
Adjacent Channel Selectivity
Bit Error Rate
Broadband Public Protection and Disaster Relief
Base Station
BandWidth
Carrier Aggregation
Network Signalled value in Carrier Aggregation
CA for band X where X is the applicable E-UTRA operating band
CA for band X and Band Y where X and Y are the applicable E-UTRA operating band
CA for band X with bandwidth class Z where X is the applicable E-UTRA operating band and Z is
the bandwidth class
Component Carrier
Coverage Enhancement
Control Plane
Continuous Wave
Downlink Control Information
DownLink
Downlink Carrier Aggregation
Digital Terrestrial Television
Device Under Test
E-UTRA Absolute Radio Frequency Channel Number
European Free Trade Association
Effective Isotropic Sensitivity
Electromagnetic compatibility and Radio spectrum Matters
Equipment Under Test
Evolved UMTS Terrestrial Radio Access
Evolved UMTS Terrestrial Radio Access
Frequency Division Duplex
Gaussian Minimum-Shift Keying

24

GSM	Global System for Mobile
HARQ	Hybrid Acknowledge Request
HD-FDD	Half- Duplex FDD
IMT	International Mobile Telecommunications
LTE	Long Term Evolution
LTE-A	LTE-Advanced
MAC	Medium Access Control
MBW	Measurement BandWidth
MCC	Mobile Country Code
MOP	Maximum Output Power
MPDCCH	Machine-Type-Communications Physical Downlink Control Channel
MSG	Mobile Standards Group
NB	Narrowband IoT
NPDCCH	Narrowband Physical Downlink Control Channel
NPDSCH	Narrowband Physical Downlink Shared Channel
NPUSCH	Narrowband Physical Uplink Shared Channel
OCNG	OFDMA Channel Noise Generator
OOB	Out Of Band
OP	OFDMA Channel Noise Generator Pattern
OTA	Over The Air
PCC	Primary Component Carrier
PDCCH	Physical Downlink Control CHannel
PDSCH	Physical Downlink Shared Channel
PHICH	Physical Hybrid ARO Indicator CHannel
PSD	Power Spectral Density
PUSCH	Physical Uplink Shared Channel
OPSK	Quadrature Phase Shift Keying
RB	Resource Block
RE	Resource Element
REFSENS	REFerence SENSitivity power level
RF	Radio Frequency
RMC	Reference Measurement Channel
RNTI	Radio Network Temporary Identifier
RRC	Root Raised Cosine
Rx	Receiver
SCC	Secondary Component Carrier
SS	System Simulator
TDD	Time Division Duplex
TFES	Task Force for European Standards for IMT
TH	Temperature High
TH/VH	High extreme Temperature/High extreme Voltage
TH/VL	High extreme Temperature/Low extreme Voltage
TL	Temperature Low
TL/VH	Low extreme Temperature/High extreme Voltage
TL/VL	Low extreme Temperature/Low extreme Voltage
TPC	Transmitter Power Control
TRP	Total Radiated Power
TRS	Total Radiated Sensitivity
Tx	Transmitter
UE	User Equipment
UL	Uplink
UL-MIMO	Uplink Multiple Antenna transmission
UMTS	Universal Mobile Telecommunications System
UTRA	UMTS Terrestrial Radio Access
VH	Higher extreme Voltage
VL	Lower extreme Voltage

4 Technical requirements specifications

4.1 Environmental profile

The technical requirements of the present document apply under the environmental profile for operation of the equipment, which shall be in accordance with its intended use. The equipment shall comply with all the technical requirements of the present document at all times when operating within the boundary limits of the operational environmental profile defined by its intended use.

4.2 Conformance requirements

4.2.0 General

The requirements in the present document are based on the assumption that the operating band (see tables 1-1 through 1-5) is shared between systems of the IMT family (for bands 3 and 8 also GSM) or systems having compatible characteristics.

4.2.1 Introduction

To meet the essential requirement under article 3.2 of Directive 2014/53/EU [i.9] for IMT User Equipment (UE), a set of essential parameters in addition to those in ETSI EN 301 908-1 [i.12] have been identified. Table 4.2.1-1 provides a cross reference between these essential parameters and the corresponding technical requirements for equipment within the scope of the present document.

Essential parameter	(Corresponding technical requirements	Corresponding test suite
Transmitter spectrum mask	4.2.3	Transmitter Spectrum emissions mask	5.3.2
Transmitter unwanted emissions in the out-of-band domain	4.2.11	Transmitter adjacent channel leakage power ratio	5.3.10
Transmitter unwanted emissions in the spurious domain	4.2.4	Transmitter spurious emissions	5.3.3
Transmitter power limits	4.2.2	Transmitter maximum output power	5.3.1
Transmitter Power Control (TPC)	4.2.5	Transmitter minimum output power	5.3.4
Transmitter power accuracy	4.2.2	Transmitter maximum output power	5.3.1
Receiver unwanted emissions in the spurious domain	4.2.10	Receiver spurious emissions	5.3.9
Receiver blocking	407	Receiver Disclairs characteristics	5.0.0
Receiver desensitization	4.2.7	Receiver Blocking characteristics	5.3.0
Receiver spurious response rejection	4.2.8	Receiver spurious response	5.3.7
Receiver radio-frequency intermodulation	4.2.9	Receiver Intermodulation characteristics	5.3.8
Receiver adjacent signal selectivity	4.2.6	Receiver Adjacent Channel Selectivity (ACS)	5.3.5
Receiver sensitivity	4.2.12	Receiver Reference Sensitivity Level	5.3.11
Antonna	4.2.13	Receiver Total Radiated Sensitivity (TRS)	5.3.12
	4.2.14	Total Radiated Power (TRP)	5.3.13
Equipment operating under the control of a	ETSI E	N 301 908-1 [i.12], clause 4.2.4 Control and	
network	Monitor	ring functions	

Table 4.2.1-1: Cross references

Unless otherwise stated, the transmitter and receiver characteristics are specified at the antenna connector(s) of the UE. For UE(s) with an integral antenna only, a reference antenna(s) with a gain of 0 dBi should be assumed for each antenna port(s). A UE with integral antenna(s) may be taken into account by converting these power levels into field strength requirements, assuming a 0 dBi gain antenna. Over The Air (OTA) antenna characteristics are specified in terms of Receiver Total Radiated Sensitivity (TRS) and Total Radiated Power (TRP).

4.2.2 Transmitter Maximum Output Power

4.2.2.1 Transmitter maximum output power for Single Carrier

4.2.2.1.1 Definition

The following UE Power Classes define the maximum output power for any transmission bandwidth within the channel bandwidth. The period of measurement shall be at least one sub-frame (1 ms).

26

4.2.2.1.2 Limits

The UE maximum output power shall be within the range in table 4.2.2.1.2-1.

E-UTRA Band	Power Class 3						
	Lower Limit (dBm)	Upper Limit (dBm)					
1	20,3	25,7					
3	20,3 (see note)	25,7					
7	20,3 (see note)	25,7					
8	20,3 (see note)	25,7					
20	20,3 (see note)	25,0					
22	18,5	26,0					
28	19,8	25,0					
31	20,3	25,7					
33	20,3	25,7					
34	20,3	25,7					
38	20,3	25,7					
40	20,3	25,0					
41	20.3	25.7					
42	19,0	26,0					
43	19,0	26,0					
65	20,3	25,7					
68	20,3	25,7					
72	20,3	25,7					
87	20,3	25,7					
88	20,3	25,7					
NOTE: For transmission bar within F _{UL_low} and F _U maximum output pow	DTE: For transmission bandwidths (ETSI TS 136 521-1 [1], clause 5) confined within F _{UL_low} and F _{UL_low} + 4 MHz or F _{UL_high} – 4 MHz and F _{UL_high} , the maximum output power requirement applies by reducing the lower limit by						
Т,5 0В.							

Table 4.2.2.1.2-1:	UE	power	classes
--------------------	----	-------	---------

NOTE: These requirements do not take into account the maximum power reductions allowed to the UE subject to certain transmission conditions specified in ETSI TS 136 101 [3], clauses 6.2.3 and 6.2.4.

4.2.2.1.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.1 of the present document.

4.2.2.2 Transmitter output power for Carrier Aggregation (DL CA and UL CA)

4.2.2.2.1 Definition

The following UE Power Classes define the maximum output power for any transmission bandwidth within the aggregated channel bandwidth.

The maximum output power is measured as the sum of the maximum output power at each UE antenna connector. The period of measurement shall be at least one sub frame (1 ms).

For inter-band carrier aggregation with uplink assigned to two E-UTRA bands, UE maximum output power shall be measured over all component carriers from different bands. If each band has separate antenna connectors, maximum output power is measured as the sum of maximum output power at each UE antenna connector.

4.2.2.2.2 Limits

For intra-band contiguous carrier aggregation, the maximum output power shall be within the range in table 4.2.2.2.1.

	Power	Class 3						
E-UTRA CA Dano	Lower Limit (dBm)	Upper Limit (dBm)						
CA_1C	20,3	25,7						
CA_3C	20,3 (see note 1)	25,7						
CA_7C	20,3 (see note 1)	25,7						
CA_8B	20,3 (see note 1)	25,7						
CA_38C	20,3	25,7						
CA_40C	20,3	25,0						
CA_41C	20,3	25,7						
CA_42C	19,3	25,7						
NOTE 1: If all transmitted resource blocks (ETSI TS 136 521-1 [1], clause 5) over all								
componer	component carriers are confined within F_{III} low and F_{III} low + 4 MHz or/and F_{III} high							
- 4 MHz a	- 4 MHz and F _{UL high} , the maximum output power requirement applies by reducing							
the lower limit by $1,5$ dB.								
NOTE 2: For intra-t	NOTE 2: For intra-band contiguous carrier aggregation the maximum power requirement shall apply to the total transmitted power over all component carriers (per LIE)							

 Table 4.2.2.2.1: UE power class for intra-band contiguous CA

NOTE: These requirements do not take into account the maximum power reductions allowed to the UE in subject to certain transmission conditions specified in ETSI TS 136 101 [3], clauses 6.2.3A and 6.2.4A.

For inter-band carrier aggregation with uplink assigned to one E-UTRA band, the requirements in clause 4.2.2.1.2 apply.

For inter-band carrier aggregation with uplink assigned to two E-UTRA bands, the maximum output power is specified in table 4.2.2.2.2-2.

E-UTRA CA	Class 3 (dBm)						
Configuration	Lower Limit (dBm)	Upper Limit (dBm)					
CA_1A-3A	19,3 (see note 2)	25,7					
CA_1A-5A	19,3	25,7					
CA_1A-7A	19,3 (see note 2)	25,7					
CA_1A-8A	19,3 (see note 2)	25,7					
CA_1A-28A	19,3	25,7					
CA_1A-42A	19,3	25,7					
CA_3A-5A	19,3 (see note 2)	25,7					
CA_3A-7A	19,3 (see note 2)	25,7					
CA_3A-8A	19,3 (see note 2) 25,7						
CA_3A-20A	-20A 19,3 (see note 2) 25,7						
CA_7A-20A	19,3 (see note 2)	25,7					
CA_7A-28A	7A-28A 19,3 (see note 2) 25,7						
NOTE 1: Void.							
NOTE 2: ² refers to the transmission bandwidths (figure 3.1-1 confined within FUL_low and FUL_low + 4 MHz or FUL_high – 4 MHz and FUL_high, the maximum output power requirement applies by reducing the lower limit by 1.5 dB.							
NOTE 3: P _{powerClass} is the maximum UE power specified without taking into account the tolerance.							
NOTE 4: For inter- to the tot	For inter-band carrier aggregation the maximum power requirement should apply to the total transmitted power over all component carriers (per UE).						

Table 4.2.2.2.2-2: UE Power Class for uplink inter-band CA (two bands)

4.2.2.2.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.1 of the present document.

4.2.2.3 Transmitter output power for UL-MIMO

4.2.2.3.1 Definition

The following UE Power Classes define the maximum output power for UE with two transmit antenna connectors in closed-loop spatial multiplexing scheme. The UL-MIMO configurations are specified in table 4.2.2.3.1-1.

Table 4.2.2.3.1-1: UL-MIMO configuration in closed-loop spatial multiplexing scheme

Transmission mode	DCI format	Codebook Index
Mode 2	DCI format 4	Codebook index 0

The maximum output power is measured as the sum of the maximum output power at each UE antenna connector. The period of measurement shall be at least one sub frame (1 ms).

4.2.2.3.2 Limits

The UE maximum output power shall be within the range in table 4.2.2.3.2-1.

Table 4.2.2.3.2-1: UE power class for UL-MIMO in closed-loop spatial multiplexing scheme

E-UTRA Ban	d Pov	wer Class 3
	Lower Limit (dBm)	Upper Limit (dBm)
1	19,3	25,7
3	19,3 (see note)	25,7
7	19,3 (see note)	25,7
8	19,3 (see note)	25,7
20	19,3 (see note)	25,0
22	17,8 (see note)	25,7
28	19,3	25,0
33	19,3	25,7
34	19,3	25,7
38	19,3	25,7
40	19,3	25,0
41	19.3	25.7
42	18,0	26,0
43	18,0	26,0
65	19,3	25,7
68	19,3	25,7
72	19.3	25.7
NOTE: For tran F _{UL_low} output p	smission bandwidths (ETSI TS 136 s and F _{UL_low} + 4 MHz or F _{UL_high} – 4 ower requirement applies by reducir	521-1 [1], clause 5) confined within MHz and F _{UL_high} , the maximum a the lower limit by 1.5 dB.

NOTE: These requirements do not take into account the maximum power reductions allowed to the UE subject to certain transmission conditions specified in ETSI TS 136 101 [3], clauses 6.2.3 and 6.2.4.

4.2.2.3.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.1 of the present document.

4.2.2.4 Transmitter output power for category NB1

4.2.2.4.1 Definition

The following UE Power Classes define the maximum output power for any transmission bandwidth within the category NB1 channel bandwidth.

For 3,75 kHz sub-carrier spacing, the maximum output power is defined as mean power of measurement which period is at least one slot (2 ms) excluding the 2 304 Ts gap when UE is not transmitting. For 15 kHz sub-carrier spacing, the maximum output power is defined as mean power of measurement which period is at least one sub-frame (1 ms).

4.2.2.4.2 Limits

The maximum output power shall be within the range in table 4.2.2.4.2-1.

	Clas	ss 3	Class 5		
EUTRA band	Lower Limit (dBm)	Upper Limit (dBm)	Lower Limit (dBm)	Upper Limit (dBm)	
1	20,3	25,7	17,3	22,7	
3	20,3	25,7	17,3	22,7	
8	20,3	25,7	17,3	22,7	
20	20,3	25,0	17,3	22,7	
28	20,3	25,0	17,3	22,7	
65	20,3	25,7	17,3	22,7	
72	20,3	25,7	17,3	22,7	
87	20,3	25,7	17,3	22,7	
88	20,3	25,7	17,3	22,7	

Table 4.2.2.4.2-1: UE Power Class test requirement

4.2.2.4.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.1 of the present document.

4.2.2.5 Transmitter output power for UE category M1

4.2.2.5.1 Definition

The following UE Power Classes define the maximum output power for any transmission bandwidth within the channel bandwidth for non CA configuration and UL-MIMO unless otherwise stated. The period of measurement shall be at least one sub frame (1 ms).

4.2.2.5.2 Limits

The maximum output power shall be within the range in table 4.2.2.5.2-1.

	Cl	ass 3	Clas	s 5		
band	Lower Limit (dBm)	Upper Limit (dBm)	Lower Limit (dBm)	Upper Limit (dBm)		
1	20,3	25,7	17,3	22,7		
3	20,3 (see note)	25,7	17,3 (see note)	22,7		
7	20,3 (see note)	25,7	17,3 (see note)	22,7		
8	20,3 (see note)	25,7	17,3 (see note)	22,7		
20	20,3 (see note)	25,0	17,3 (see note)	22,7		
28	19,8	25,0	16,8	22,7		
31	20,3	25,7	17,3	22,7		
41	20,3	25,7	17,3	22,7		
72	20,3	25,7	17,3	22,7		
87	20,3	25,7	17,3	22,7		
88	20,3	25,7	17,3	22,7		
NOTE: For transmission bandwidths (ETSI TS 136 521-1 [1], clause 5) confined within F _{UL_low}						
and $F_{UL_{low}}$ + 4 MHz or $F_{UL_{high}}$ - 4 MHz and $F_{UL_{high}}$, the maximum output power requirement applies by reducing the lower limit by 1.5 dB						

Table 4.2.2.5.2-1: UE Power Class

30

4.2.2.5.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.1 of the present document.

4.2.3 Transmitter Spectrum Emission Mask

4.2.3.1 Transmitter spectrum emission mask for Single Carrier

4.2.3.1.1 Definition

The spectrum emission mask of the UE applies to frequencies (Δf_{OOB}) starting from the ± edge of the assigned E-UTRA channel bandwidth.

4.2.3.1.2 Limits

The power of any UE emission shall fulfil requirements in tables 4.2.3.1.2-1 to 4.2.3.1.2-3.

Δf _{OOB}	(MHz)	1,4 MHz	3,0 MHz	5 MHz	10 MHz	15 MHz	20 MHz	Measurement bandwidth
0 tc	b 1	-8,5	-11,5	-13,5	-16,5	-18,5	-19,5	30 kHz
1 to	2,5	-8,5	-8,5	-8,5	-8,5	-8,5	-8,5	1 MHz
2,5 to	0 2,8	-23,5	-8,5	-8,5	-8,5	-8,5	-8,5	1 MHz
2,8 t	to 5		-8,5	-8,5	-8,5	-8,5	-8,5	1 MHz
5 tc	6		-23,5	-11,5	-11,5	-11,5	-11,5	1 MHz
6 to	10			-23,5	-11,5	-11,5	-11,5	1 MHz
10 to	o 15				-23,5	-11,5	-11,5	1 MHz
15 to	o 20					-23,5	-11,5	1 MHz
20 to	o 25						-23,5	1 MHz
NOTE 1:	The first a	nd last mea	isurement p	position wit	h a 30 kHz	filter is at	∆f _{OOB} equ	als to 0,015 MHz and
	0.985 MHz.							
NOTE 2:	The first a	nd last mea	surement p	position wit	h a 1 MHz	filter for 1	MHz - 2,5 I	MHz offset range is at Δf_{OOB}
	equals to 1,5 MHz and 2,0 MHz. Similarly for other Δf_{OOB} ranges.							
NOTE 3:	3: The measurements shall be performed above the upper edge of the channel and below the lower edge							
	of the channel.							
NOTE 4:	4: For the 2,5 MHz - 2,8 MHz offset range with 1,4 MHz channel bandwidth, the measurement position is							
	at Δf_{OOB} e	quals to 3 I	MHz.					

		Spectrum emission limit (dBm)/Channel bandwidth						
Δf _{OOB} (MHz)	1,4 MHz	3,0 MHz	5 MHz	10 MHz	15 MHź	20 MHz	Measurement bandwidth	
0 to 1	-8,2	-11,2	-13,2	-16,2	-18,2	-19,2	30 kHz	
1 to 2,5	-8,2						1 MHz	
2,5 to 2,8	-23,2	-8,2	-8,2	-8,2	-8,2	-8,2	1 MHz	
2,8 to 5							1 MHz	
5 to 6		-23,2	-11,2	-11,2	-11,2	-11,2	1 MHz	
6 to 10			-23,2				1 MHz	
10 to 15				-23,2			1 MHz	
15 to 20					-23,2		1 MHz	
20 to 25						-23,2	1 MHz	
NOTE 1: The firs	t and last me	easurement	t position v	vith a 30 kH	Iz filter is a	t ∆f _{OOB} eq	uals to 0,015 MHz and	
0,985 N	/Hz.							
NOTE 2: At the b	oundary of s	spectrum er	mission lim	nit, the first	and last me	easuremer	nt position with a 1 MHz filter	
is the ir	side of +0,5	MHz and -	0,5 MHz, r	espectively	' .			
NOTE 3: The me	measurements shall be performed above the upper edge of the channel and below the lower							
edge of	edge of the channel.							
NOTE 4: For the	2,5 MHz to 2	2,8 MHz off	set range	with 1,4 MI	Iz channel	bandwidth	, the measurement position	
is at ∆f	_{DOB} equals to	o 3 MHz.						

Table 4.2.3.1.2-2: General E-UTRA spectrum emission mask, 3 GHz < E-UTRA bands ≤ 4,2 GHz

Fable 4.2.3.1.2-3: Additional	spectrum emission mask ((network signalled value "N	IS_01")
-------------------------------	--------------------------	-----------------------------	---------

E-UTRA band	Frequency range	Channel Spectrum emission Bandwidth limit (dBm)		Measurement Bandwidth			
863 MHz ≤ f ≤ 867 MHz		10 MHz (note 2)	-11,5	1 MHz			
20	867 MHz \leq f \leq 870 MHz 10 MHz (note 2)		-14,5	1 MHz			
NOTE 1: At the boundary of spectrum emission limit, the first and last measurement position with a 1 MHz filter is the							
inside of +0,5 MHz and -0,5 MHz, respectively.							
NOTE 2: The co	NOTE 2: The conformance shall be assessed at test frequency 857 MHz with 50 RB allocation.						

4.2.3.1.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.2 of the present document.

4.2.3.2 Transmitter spectrum emission mask for Carrier Aggregation (DL CA and UL CA)

4.2.3.2.1 Definition

For intra-band contiguous carrier aggregation, the spectrum emission mask of the UE applies to frequencies (Δf_{OOB}) starting from the ± edge of the aggregated channel bandwidth (table 3.1-1).

For intra-band contiguous carrier aggregation of the bandwidth class B and C, the power of any UE emission shall not exceed the levels specified in table 6.6.2.1A-0 and table 6.6.2.1A-1 in ETSI TS 136 101 [3] for the specified channel bandwidth.

4.2.3.2.2 Limits

For intra-band contiguous carrier aggregation of the bandwidth class B and C, the power of any UE emission shall not exceed the levels specified in tables 4.2.3.2.2-1 and 4.2.3.2.2-2 for the specified channel bandwidth.

Spectrum emission limit [dBm]/BW								
Δf _{OOB} (MHz)	25RB+100RB (24,95 MHz)	50RB+75RB (24,75 MHz)	50RB+100RB (29,9 MHz)	75RB+75RB (30 MHz)	75RB+100RB (34,85 MHz)	100RB+100RB (39,8 MHz)	Measurement bandwidth	
±0 - 1	-20,5	-20,5	-21	-21	-22	-22,5	30 kHz	
±1 - 5	-8,5	-8,5	-8,5	-8,5	-8,5	-8,5	1 MHz	
±5 - 24,75	-11,5	-11,5	-11,5	-11,5	-11,5	-11,5	1 MHz	
±24,75 - 24,95	-11,5	-23,5	-11,5	-11,5	-11,5	-11,5	1 MHz	
±24,95 - 29,75	-23,5	-23,5	-11,5	-11,5	-11,5	-11,5	1 MHz	
±29,75 - 29,9	-23,5		-11,5	-11,5	-11,5	-11,5	1 MHz	
±29,9 - 29,95	-23,5		-23,5	-11,5	-11,5	-11,5	1 MHz	
±29,95 - 30			-23,5	-11,5	-11,5	-11,5	1 MHz	
±30 - 34,85			-23,5	-23,5	-11,5	-11,5	1 MHz	
±34,85 - 34,9			-23,5	-23,5	-23,5	-11,5	1 MHz	
±34,9 - 35				-23,5	-23,5	-11,5	1 MHz	
±35 - 39,8					-23,5	-11,5	1 MHz	
±39,8 - 39,85					-23,5	-23,5	1 MHz	
±39,85 - 44,8						-23,5	1 MHz	

Table 4.2.3.2.2-1: General E-UTRA CA spectrum emission mask for Bandwidth Class C, E UTRA bands ≤ 3 GHz

NOTE 1: The first and last measurement position with a 30 kHz filter is at Δf_{OOB} equals to 0,015 MHz and 0,985 MHz.

NOTE 2: At the boundary of spectrum emission limit, the first and last measurement position with a 1 MHz filter is the inside of +0,5 MHz and -0,5 MHz, respectively.

NOTE 3: The measurements shall be performed above the upper edge of the aggregated channel bandwidth and below the lower edge of the aggregated channel bandwidth.

Spectrum emission limit [dBm]/BW _{Channel_CA}								
Δf _{OOB} (MHz)	25RB+100RB (24,95 MHz)	50RB+75RB (24,75 MHz)	50RB+100RB (29,9 MHz)	75RB+75RB (30 MHz)	75RB+100RB (34,85 MHz)	100RB+100RB (39,8 MHz)	Measurement bandwidth	
±0 - 1	-20,2	-20,2	-20,7	-20,7	-21,7	-22,2	30 kHz	
±1 - 5	-8,2	-8,2	-8,2	-8,2	-8,2	-8,2	1 MHz	
±5 - 24,75	-11,2	-11,2	-11,2	-11,2	-11,2	-11,2	1 MHz	
±24,75 - 24,95	-11,2	-23,2	-11,2	-11,2	-11,2	-11,2	1 MHz	
±24,95 - 29,75	-23,2	-23,2	-11,2	-11,2	-11,2	-11,2	1 MHz	
±29,75 - 29,9	-23,2		-11,2	-11,2	-11,2	-11,2	1 MHz	
±29,9 - 29,95	-23,2		-23,2	-11,2	-11,2	-11,2	1 MHz	
±29,95 - 30			-23,2	-11,2	-11,2	-11,2	1 MHz	
±30 - 34,85			-23,2	-23,2	-11,2	-11,2	1 MHz	

-23,2

-23,2

-23,2

-23,2

-23,2

-23,2

-11,2

-11,2

-11,2

-23,2

-23,2

1 MHz

1 MHz

1 MHz

1 MHz

1 MHz

Table 4.2.3.2.2-2: General E-UTRA CA spectrum emission mask for Bandwidth Class C, 3 GHz < E UTRA bands ≤ 4,2 GHz

33

44,8 NOTE 1: The first and last measurement position with a 30 kHz filter is at Δf_{OOB} equals to 0,015 MHz and 0,985 MHz. NOTE 2: At the boundary of spectrum emission limit, the first and last measurement position with a 1 MHz filter is the inside of +0,5 MHz and -0,5 MHz, respectively. NOTE 3: The measurements shall be performed above the upper edge of the aggregated channel bandwidth and

below the lower edge of the aggregated channel bandwidth.

-23,2

For inter-band carrier aggregation with one component carrier per operating band and the uplink active in two E-UTRA bands, the spectrum emission mask of the UE is defined per component carrier while both component carriers are active and the requirements are specified in clause 4.2.3.1.2. If for some frequency spectrum emission masks of component carriers overlap, then spectrum emission mask allowing higher power spectral density applies for that frequency. If for some frequency a component carrier spectrum emission mask overlaps with the channel bandwidth of another component carrier, then the emission mask does not apply for that frequency.

4.2.3.2.3 Conformance

±34,85 -

34,9

±34,9 -35

> ±35 -39,8

±39,8 -39,85

±39,85 -

The conformance tests for this requirement shall be as defined in clause 5.3.2 of the present document.

4.2.3.3 Transmitter spectrum emission mask for UL-MIMO

4.2.3.3.1 Definition

For UE supporting UL-MIMO, the requirements for out-of-band emissions resulting from the modulation process and non-linearity in the transmitters are specified at each transmit antenna connector.

The spectrum emission mask of the UE applies to frequencies (Δf_{OOB}) starting from the edge of the assigned E-UTRA channel bandwidth.

4.2.3.3.2 Limits

For UEs with two transmit antenna connectors in closed-loop spatial multiplexing scheme, the requirements in clause 4.2.3.1.2 apply to each transmit antenna connector. The requirements shall be met with the UL-MIMO configurations specified in table 4.2.3.1-1.

34

4.2.3.3.3 Conformance

Conformance tests described in clause 5.3.2 shall be carried out.

4.2.3.4 Transmitter spectrum emission mask for Multi-Cluster PUSCH within a component carrier

4.2.3.4.1 Definition

For UE supporting multi cluster PUSCH within a component carrier for the operating band.

The spectrum emission mask of the UE applies to frequencies (Δf_{OOB}) starting from the edge of the assigned E-UTRA channel bandwidth.

4.2.3.4.2 Limits

The power of any UE emission shall fulfil the requirements specified in tables 4.2.3.1.2-1 and 4.2.3.1.2-2.

4.2.3.4.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.2 of the present document.

4.2.3.5 Transmitter spectrum emission mask for category NB1

4.2.3.5.1 Definition

The spectrum emission mask of the UE applies to frequencies (Δf_{OOB}) starting from the ± edge of the assigned E-UTRA channel bandwidth.

4.2.3.5.2 Limits

The power of any UE emission shall fulfil requirements in table 4.2.3.5.2-1.

Table 4.2.3.5.2-1: Category NB1 UE spectrum emission mask

Δf _{OOB} (kHz)	Spectrum emission limit (dBm)	Measurement bandwidth
±0	24,5	30 kHz
±100	-3,5	30 kHz
±150	-6,5	30 kHz
±300	-27,5	30 kHz
±500 - 1 700	-33,5	30 kHz

4.2.3.5.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.2 of the present document.

4.2.4 Transmitter Spurious Emissions

4.2.4.1 Transmitter spurious emissions for Single Carrier

4.2.4.1.1 Definition

Spurious emissions are emissions which are caused by unwanted transmitter effects such as harmonics emission, parasitic emissions, intermodulation products and frequency conversion products, but exclude out-of-band emissions. The spurious emission limits are specified in terms of general requirements in line with Recommendation ITU-R SM.329-12 [i.4] and E-UTRA operating band requirement to address UE co-existence.

35

4.2.4.1.2 Limits

The spurious emission limits in table 4.2.4.1.2-2 apply for the frequency ranges that are more than Δf_{OOB} (MHz) from the edge of the channel bandwidth shown in table 4.2.4.1.2-1.

The measured average power of spurious emission for general requirements shall not exceed the described values in table 4.2.4.1.2-2.

The measured average power of spurious emission for E-UTRA operating band specific requirements to protected bands shall not exceed the described values in tables 4.2.4.1.2-3 to 4.2.4.1.2-6.

Table 4.2.4.1.2-1: Δf_{OOB} boundary between E-UTRA channel and spurious emission domain

Channel bandwidth	1,4 MHz	3,0 MHz	5 MHz	10 MHz	15 MHz	20 MHz
Δf _{OOB} (MHz)	2,8	6	10	15	20	25

NOTE 1: For measurement conditions at the edge of each frequency range, the lowest frequency of the measurement position in each frequency range should be set at the lowest boundary of the frequency range plus MBW/2. The highest frequency of the measurement position in each frequency range should be set at the highest boundary of the frequency range minus MBW/2. MBW denotes the measurement bandwidth defined for the protected band.

Frequency range	Maximum level	Measurement bandwidth	Comment
9 kHz ≤ f < 150 kHz	-36 dBm	1 kHz	
150 kHz ≤ f < 30 MHz	-36 dBm	10 kHz	
30 MHz ≤ f < 1 000 MHz	-36 dBm	100 kHz	
1 GHz ≤ f < 12,75 GHz	-30 dBm	1 MHz	
12,75 GHz ≤ f < 5 th	-30 dBm	1 MHz	See note
harmonic of the upper			
frequency edge of the UL			
operating band in GHz			
NOTE: Shall apply for Bar	nd 22, 42 and Band 43.		

Table 4.2.4.1.2-2: General spurious emissions limits

The additional requirements in table 4.2.4.1.2-3 apply for the frequency ranges that are more and less than Δf_{OOB} (MHz) from the edge of the channel bandwidth shown in table 4.2.4.1.2-1.

E.	Spurious emission						
UTRA Band	Protected band	Frequency range (MHz)			Maximum Level (dBm)	MBW (MHz)	Comment
1	E-UTRA Band 1, 3, 7, 8, 20, 22, 28, 31, 32, 38, 40, 41,42, 43, 65, 67, 68, 72, 87, 88	F _{DL_low}	-	$F_{DL_{high}}$	-50	1	
	E-UTRA Band 34	F _{DL low}	-	F _{DL high}	-50	1	Note 3
	Frequency range	1 895	-	1 915	-15,5	5	Notes 3, 8
	Frequency range	1 915	-	1 920	+1,6	5	Notes 3, 8, 42
3	E-UTRA Band 1, 7, 8, 20, 28, 31, 32, 33, 34, 38, 40, 41, 43, 65, 67, 68, 72, 87, 88	F _{DL_low}	-	$F_{DL_{high}}$	-50	1	
	E-UTRA Band 3	F _{DL low}	-	F _{DL high}	-50	1	Note 3
	E-UTRA Band 22, 42	F _{DL low}	-	F _{DL high}	-50	1	Note 2
7	E-UTRA Band 1, 3, 7, 8, 20, 22, 28, 31, 32, 33, 34, 40, 42, 43, 65, 67, 68, 72, 87, 88	F _{DL_low}	-	F _{DL_high}	-50	1	
	Frequency range	2 570	-	2 575	+1,6	5	Notes 3, 4
	Frequency range	2 575	-	2 595	-15,5	5	Notes 3, 4
	Frequency range	2 595		2 620	-40	1	Notes 3, 4
8	E-UTRA Band 1, 20, 28, 31, 32, 33, 34, 38, 40, 65, 67, 68, 72, 87, 88	F _{DL_low}	-	⊢ _{DL_high}	-50	1	
	E-UTRA Band 3, 41	F _{DL_low}	-	F _{DL_high}	-50	1	Note 2
	E-UTRA Band 7	F _{DL_low}	-	F _{DL_high}	-50	1	Note 2
	E-UTRA Band 8	F _{DL_low}	-	F _{DL_high}	-50	1	Note 3
	E-UTRA Band 22, 42, 43	F _{DL low}	-	F _{DL high}	-50	1	Note 2
20	E-UTRA Band 1, 3, 7, 8, 22, 31, 32, 33, 34, 40, 43, 65, 67, 68, 72, 87, 88	F _{DL_low}	-	F _{DL_high}	-50	1	
	E-UTRA Band 20	F _{DL low}	-	F _{DL high}	-50	1	Note 3
	E-UTRA Band 38, 42	F _{DL low}	-	F _{DL bigb}	-50	1	Note 2
	Frequency range	758		788	-50	1	
22	E-UTRA Band 1, 3, 7, 8, 20, 28, 31, 32, 33, 34, 38, 40, 43, 65, 67, 68, 72, 87, 88	F _{DL_low}	-	F_{DL_high}	-50	1	
	Frequency range	3 510	-	3 525	-40	1	Note 3
	Frequency range	3 525	-	3 590	-50	1	
28	E-UTRA Band 3, 7, 8, 20, 31, 34, 38, 41, 72, 87, 88	F _{DL_low}	-	F _{DL_high}	-50	1	
	E-UTRA Band 1, 22, 32, 42, 43, 65	F _{DL_low}	-	F _{DL_high}	-50	1	Note 2
	E-UTRA Band 1	F _{DL_low}	-	F _{DL_high}	-50	1	Notes 6, 19
	Frequency range	<u>/58</u> 772		//3	-32	1	Note 3
	Frequency range	113		803	-50	1	
	Frequency range	470	<u> </u> _	694	-42	8	Notes 3, 7
31	E-UTRA Band 1, 7, 8, 20, 22, 28, 31, 32, 33, 34, 38, 40, 42, 43, 65, 67, 68, 72, 87, 88	F _{DL_low}	-	F _{DL_high}	-50	1	
	E-UTRA Band 3	F _{DI IOW}	-	F _{DI biab}	-50	1	Note 2
33	E-UTRA Band 1, 7, 8, 20,				-50	1	Note 9
	22, 28, 31, 32, 34, 38, 40, 42, 43, 65, 67, 72, 87, 88			DE_IIIGII			

Table 4.2.4.1.2-3: Spurious emission band UE co-existence limits (network signalled value "NS_01")

36

-

 $\mathsf{F}_{\mathsf{DL}_\mathsf{low}}$

F_{DL_high}

-50

1

Note 3

E-UTRA Band 3
_	Spurious emission									
	Protected band	Frequency range			Maximum	MBW (MHz)	Comment			
Band		(MHz)		Level						
Bulla					(dBm)					
34	E-UTRA Band 1, 3, 7, 8, 20,	F _{DL_low}	-	F _{DL_high}	-50	1	Note 9			
	22, 28, 31, 32, 33, 38, 40,									
	41, 42, 43, 65, 67, 72, 87,									
20	88 FUTDA Band 1, 2, 9, 20				50	4				
38	E-UTRA Band T, 3, 8, 20,	F _{DL_low}	-	► _{DL_high}	-50	1				
	12 13 65 67 68 72 87									
	88									
	Frequency range	2 620	-	2 645	-15.5	5	Notes 3, 5			
	Frequency range	2 645	-	2 690	-40	1	Notes 3, 5			
40	E-UTRA Band 1, 3, 7, 8, 20,		-	F _{DL bigh}	-50	1				
_	22, 28, 31, 32, 33, 34, 38,	DL_IOW		DL_nign						
	41, 42, 43, 65, 67, 68, 72,									
	87, 88									
	Frequency range	1 884,5	-	1915,7	-41	0,3				
	Frequency range	1 475	-	1 518	-50	1	_			
	Frequency range	3 300	-	4 200	-50	1				
	Frequency range	4 400	-	5 000	-50	1	Note 2			
41	E-UTRA Band 1, 3, 8, 28,	FDL_low	-	FDL_high	-50	1				
	34, 40, 42, 65,	1 00 4 5		1 015 7	44	0.2	Note 20			
40	Frequency range	1 884,5		1915,7 E	-41	0,3	Note 30			
42	28 31 32 33 34 38 40	「DL_low	-	^F DL_high	-50	I				
	41 65 67 68 72 87 88									
	Frequency range	1884.5	-	1915.7	-41	0.3				
43	E-UTRA Band 1, 3, 7, 8, 20,	F _{DL low}	-	F _{DL bigh}	-50	1				
	28, 31, 32, 33, 34, 38, 40,	DL_IOW		DL_IIIgI1						
	65, 67, 68, 72, 87, 88									
65	E-UTRA Band 1, 7, 8, 20,	F _{DL low}	-	F _{DI high}	-50	1				
	22, 28, 31, 32, 38, 40, 42,									
	43, 65, 68, 72, 87, 88	_								
	E-UTRA Band 3	F _{DL_low}	-	F _{DL_high}	-50	1	Note 3			
	E-UTRA Band 5, 11, 18, 19,	F _{DL_low}	-	F _{DL_high}	-50	1				
	21, 26, 27, 41	_								
	E-UTRA Band 34	F _{DL_low}	-	F _{DL_high}	-50	1	Note 36			
	Frequency range	1 884,5	-	1 915,7	-41	0,3	Note 37			
	Frequency range	1 900	-	1 915	-15,5	5	Notes 3, 8			
	Frequency range	1 915	-	1 920	+1,6	5	Notes 3, 8			
68	E-UTRA Band 3, 7, 8, 20,	-	-	_	50					
	22, 28, 31, 38, 40, 42, 43,	F _{DL_low}		F _{DL_high}	-50	1				
		E			50		Nata 0			
	E-UTRA Band 1	FDL_low	-	FDL_high	-50	1	Note 2			
72	E-UIRA Band 1, 7, 20, 22,	FDL_low	-	FDL_high	-50	1				
	28, 31, 32, 33, 34, 38, 42,									
	43, 41, 32, 03, 08, 12, 81, 88									
	F-UTRA Band 3 8 40	EDI Jow		FDI high	-50	1	Note 2			
	Frequency range	470	-	<u>694</u>	-42	8	1002			

-	Spurious emission									
UTRA Band	Protected band	Frequ	enc (MH	y range z)	Maximum Level (dBm)	MBW (MHz)	Comment			
87	E-UTRA Band 1, 3, 7, 8, 22,	FDL low	-	FDL high	-50	1				
	28, 31, 32, 33, 34, 38, 40,			· <u>-</u> ··· 3 ··						
	42, 43, 47, 52, 65, 68, 72									
	E-UTRA Band, 20	FDL_low	-	FDL_high	-50	1	Note 2			
	E-UTRA Band 87, 88	FDL_low	-	FDL_high	-50	1	Note 3			
00	Frequency range	470	-	694	-42	8				
88	E-UTRA Band 1, 3, 7, 8, 20, 22, 28, 31, 32, 33, 34, 38	FDL_IOW	-	FDL_nign	-50	1				
	40, 42, 43, 47, 52, 65, 68, 72									
	E-UTRA Band 87	FDL low	-	FDL high	-50	1	Note 3			
	E-UTRA Band 88	FDL_low	-	FDL_high	-50	1	Note 3			
	Frequency range	470	-	694	-42	8				
NOTE 1:	F _{DL_low} and F _{DL_high} refer to	each frequency	rang	e of the protected	E-UTRA band	d.				
NOTE 2:	As exceptions, measuremen	nts with a level up	to to	the applicable requ	irements defi	ned in table 4.2.4.	1.2-2 are			
	permitted for each assigned	E-UTRA carrier	used	d in the measurem	ent due to 2 nd	^I , 3 rd or 4 th harmor	nic spurious			
	emissions. Due to spreading	g of the harmonic	em	ission, the exception	on shall be all	owed for the first 1	MHz			
	frequency range immediate	ly outside the har	mon	ic emission on bot	h sides of the	harmonic emissic	on. This			
	results in an overall exception	on interval centre	d at	the harmonic emis	sion of (2 MH	lz + N x L _{CRB} x 18	0 kHz),			
	where N is 2, 3, 4 for the 2 ⁿ	^d , 3 rd or 4 th harmo	onic	respectively. The	exception is a	llowed if the meas	surement			
	bandwidth (MBW) totally or	partially overlaps	the	overall exception i	nterval.					
NOTE 3:	These requirements shall al	so apply for the f	requ	ency ranges that a	re less than <i>L</i>	Δf _{OOB} (MHz) in				
	table 4.2.4.1.2-1 from the ed	dge of the channe	el ba	ndwidth.						
NOTE 4:	This requirement shall be a	oplicable for any	char	nel bandwidths wi	thin the range	e 2 500 MHz to 2 5	570 MHz with			
	the following restriction: for	carriers of 15 MH	z ba	ndwidth when cari	ier centre free	quency is within th	e range			
	2 300,3 MHZ 10 2 302,3 MH range 2 552 MHz to 2 560 M	/Hz the requirem	onti	is applicable only f	or an unlink tr	ansmission bandy	vidth less			
	than or equal to 54 RB	in iz the requirem	ent				vidin 1633			
NOTE 5:	This requirement shall be a	oplicable for any	char	nel bandwidths wi	thin the range	2 570 MHz to 2 6	515 MHz with			
	the following restriction: for	carriers of 15 MH	z ba	Indwidth when carr	ier centre fred	quency is within th	e range			
	2 605,5 MHz to 2 607,5 MH	z and for carriers	of 2	0 MHz bandwidth	when carrier o	centre frequency is	s within the			
	range 2 597 MHz to 2 605 M	/Hz the requirem	ent	is applicable only f	or an uplink tr	ansmission bandw	vidth less			
	than or equal to 54 RB.		:							
	For carriers with channel ba	ndwidth overlapp	ung i	the frequency rang	e 2 615 t IVIH. to 1 10 dBm	2 0 2 620 MHZ the)			
NOTE 6	As exceptions measurement	nts with a level ur	to 1	he applicable requ	10 + 19 ubiii.	6 dBm/MHz are ne	ermitted for			
	each assigned E-LITRA car	rier used in the m	-25	urement due to 3 rd	harmonic so		An exception			
	is allowed if there is at least	one individual RI	B wi	thin the transmission	on bandwidth	(see figure 5.4.2-	1 in ETSI			
	TS 136 521-1 [1]) for which	the 3 rd harmonic	tota	llv or partially over	aps the Meas	surement Bandwid	lth (MBW).			
NOTE 7:	This requirement shall be a	oplicable in the ca	ase (of a 10 MHz E-UTF	RA carrier con	fined within 703 M	/Hz and			
	733 MHz, otherwise the req	uirement of -25 d	Bm	with a measureme	nt bandwidth	of 8 MHz applies.				
NOTE 8:	This requirement shall be a	oplicable for any	char	nel bandwidths wi	thin the range	e 1 920 MHz to 1 9	80 MHz with			
	the following restriction: for	carriers of 15 MH	z ba	indwidth when carr	ier centre free	quency is within th	e range			
	1 927,5 MHZ to 1 929,5 MH	z and for carriers	OT 2	0 MHz bandwidth	when carrier of	centre frequency is	s within the			
	then or equal to 54 RB		ent	is applicable only in	or an uplink tr	ansmission bandy	vidin less			
NOTE 9	For non-synchronized TDD	operation to mee	t the	se requirements s	ome restrictio	ns will be needed	for either			
	the operating band or protect	cted band.								
NOTES 1	0 to 18: N/A.									
NOTE 19:	Applicable when the assign	ed E-UTRA carrie	er is	confined within 71	8 MHz and 74	8 MHz and when	the channel			
	bandwidth used is 5 or 10 M	1Hz.								
NOTES 2	0 to 29: N/A.									
NOTE 30:	This requirement applies wr	ien the E-UIRA (carri	er is contined withi	n 2 545 MHZ	to 2 575 MHz or 2	595 MHZ to			
	ע טיוס איס צווע באיט בייט ב anu the channel 1 to 35: N/∆	banuwiuth is 10	01 2							
NOTE 36	This requirement is applicat	ble for E-UTRA ch	nanr	el bandwidth alloc	ated within 1	920 MHz to 1 980	MHz.			
NOTE 37:	Applicable when the upper	edge of the chann	nel b	andwidth frequenc	y is greater th	nan 1 980 MHz.				
NOTES 3	8 to 41: N/A.	<u>.</u>			, . <u>.</u>					
NOTE 42:	For category NB1 and NB2	UE when carrier	cent	re frequency is 1 9	20,1 MHz, in	case of single-ton	e uplink			
	transmission the requirement	nt is applicable or	nly fo	or sub-carrier index	< > 2.					

38

NOTE 2: For measurement conditions at the edge of each frequency range, the lowest frequency of the measurement position in each frequency range should be set at the lowest boundary of the frequency range plus MBW/2. The highest frequency of the measurement position in each frequency range should be set at the highest boundary of the frequency range minus MBW/2. MBW denotes the measurement bandwidth defined for the protected band.

Table 4.2.4.1.2-4: Additional spurious emissions limits (network signalled value "NS_01")

E-UTRA band	Protected Frequency range	Maximum Level (dBm)	MBW (MHz)					
20	470 MHz ≤ f ≤ 790 MHz	-65	8 MHz					
NOTE: The co followir 786 MH	NOTE: The conformance shall be assessed using the measurement position placed at the following centre frequencies: 474 MHz, 586 MHz, 690 MHz, 754 MHz, 770 MHz and 786 MHz.							

Table 4.2.4.1.2-5: Additional spurious emissions limits (network signalled value "NS_22")

E-UTRA band	Protected Frequency range (MHz)	otectedChannel bandwidth/ Maximum Level (dBm)(MHz)5, 10, 15, 20 MHz						
42, 43	3 400 ≤ f ≤ 3 800	-23 (note 1)	5 MHz					
		-40 (note 2)	1 MHz					
NOTE 1: This required the lowe	uirement shall apply wit r and from the upper ec	hin an offset between 5 MHz and 2 dge of the channel bandwidth.	25 MHz from					
NOTE 2: This req E-UTRA up to 3 8	NOTE 2: This requirement shall apply from 3 400 MHz up to 25 MHz below the lower E-UTRA channel edge and from 25 MHz above the upper E-UTRA channel edge up to 3 800 MHz.							

Table 4.2.4.1.2-6: Additional spurious emissions limits (network signalled value "NS_23")

E-UTRA band		Protected Frequency range	Channel bandwidth/ Maximum Level (dBm)	MBW (MHz)
		(MHZ)	5, 10, 15, 20 MHZ	
42,	43	3 400 ≤ f ≤ 3 800	-23 (note 1)	5 MHz
			-40 (note 2)	1 MHz
NOTE 1:	This requir	rement shall apply with	nin an offset between 5 MHz + F _{offse}	et_NS_23 and
	25 + F _{offse}	t NS 23 MHz from the	lower and from the upper edge of th	e channel bandwidth.
NOTE 2:	This requir	rement shall apply fror	n 3 400 MHz up to 25 + F _{offset NS 23}	3 MHz below the lower
	E-UTRA c	hannel edge and from	25 MHz above the upper E-UTRA	channel edge up to
	3 800 MHz	<u>Z.</u>		
NOTE 3:	F _{offset_NS_2}	₂₃ is:		
	0 MHz for	5 MHz channel BW;		
	5 MHz for	10 MHz channel BW;		
	9 MHz for	15 MHz channel BW;	and	
	12 MHz fo	r 20 MHz channel BŴ		

Table 4.2.4.1.2-7: Additional spurious emissions limits (network signalled value "NS_36")

Frequency band (MHz)		Channel bandwidth / Spectrum emission limit (dBm) 5 MHz 10 MHz and 15 MHz	Measurement bandwidth		
47	0 ≤ f ≤ 694	-42	8 MHz		
NOTE: For a 5 MHz E-UTRA carrier confined within 698 MHz and 703 MHz, this requirement shall be met in normal conditions only. The requirement in extrem conditions is -30 dBm.					

4.2.4.1.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.3 of the present document.

4.2.4.2 Transmitter spurious emissions for Carrier Aggregation (DL CA and UL CA)

4.2.4.2.1 Definition

Spurious emissions are emissions which are caused by unwanted transmitter effects such as harmonics emission, parasitic emissions, intermodulation products and frequency conversion products, but exclude out-of-band emissions. The spurious emission limits are specified in terms of general requirements in line with Recommendation ITU-R SM.329-12 [i.4] and E-UTRA operating band requirement to address UE co-existence.

To improve measurement accuracy, sensitivity and efficiency, the resolution bandwidth may be smaller than the measurement bandwidth. When the resolution bandwidth is smaller than the measurement bandwidth, the result shall be integrated over the measurement bandwidth in order to obtain the equivalent noise bandwidth of the measurement bandwidth.

4.2.4.2.2 Limits

NOTE 1: For measurement conditions at the edge of each frequency range, the lowest frequency of the measurement position in each frequency range should be set at the lowest boundary of the frequency range plus MBW/2. The highest frequency of the measurement position in each frequency range should be set at the highest boundary of the frequency range minus MBW/2. MBW denotes the measurement bandwidth defined for the protected band.

For intra-band contiguous carrier aggregation, the spurious emission limits apply for the frequency ranges that are more than Δf_{OOB} (MHz) in table 4.2.4.2.2-1 from the ± edge of the aggregated channel bandwidth. For frequencies Δf_{OOB} greater than f_{OOB} as specified in table 4.2.4.2.2-1 the spurious requirements in table 4.2.4.2.2-2 are applicable.

For frequencies Δf_{OOB} greater than f_{OOB} as specified in table 4.2.4.2.2-1 the measured average power of spurious emission for general requirements, shall not exceed the described values in table 4.2.4.2.2-2.

-	_
CA Bandwidth Class	OOB boundary f _{OOB}
OA Banawiatin Olass	(MHz)
А	table 4.2.4.1.2-1
В	BW _{Channel_CA} + 5
С	BW _{Channel CA} + 5

spurious emission domain for intra-band contiguous CA

Table 4.2.4.2.2-1: Boundary between E-UTRA Δf_{OOB} and

Table 4.2.4.2.2-2: General sp	urious emissions	limits for intra-band	contiguous CA
Frequency Range	Maximum	Measurement	Comment

Frequency Range	Maximum	Measurement	Comment				
	Level	Bandwidth					
9 kHz ≤ f < 150 kHz	-36 dBm	1 kHz					
150 kHz ≤ f < 30 MHz	-36 dBm	10 kHz					
30 MHz ≤ f < 1 000 MHz	-36 dBm	100 kHz					
1 GHz ≤ f < 12,75 GHz	-30 dBm	1 MHz					
12,75 GHz \leq f < 5 th harmonic of	-30 dBm	1 MHz	See note				
the upper frequency edge of the							
UL operating band in GHz							
NOTE: Shall apply for Band 22, 42 and Band 43.							

For inter-band carrier aggregation with one component carrier per operating band and the uplink active in two E-UTRA bands, the spurious emission limit in table 4.2.4.2.2-2 applies for the frequency ranges that are more than f_{OOB} as defined in table 4.2.4.1.2-1 away from edges of the assigned channel bandwidth on a component carrier. If for some frequency a spurious emission limit of individual component carrier overlaps with the spectrum emission mask or channel bandwidth of another component carrier then it does not apply.

NOTE 2: For inter-band carrier aggregation with uplink assigned to two E-UTRA bands the limits in table 4.2.4.2.2-2 could be verified by measuring spurious emissions at the specific frequencies where second and third order intermodulation products generated by the two transmitted carriers can occur; in that case, the limits for remaining applicable frequencies in table 4.2.4.2.2-2 would be considered to be verified by the measurements verifying the one uplink inter-band CA spurious emission limit.

41

For combinations of intra-band and inter-band carrier aggregation with three uplink component carriers (up to two contiguously aggregated carriers per band), the spurious emission limit is defined as follows. For the E-UTRA band supporting one component carrier, the limits in table 4.2.4.2.2-2 apply for frequency ranges that are more than f_{OOB} (MHz) from the edges of assigned channel bandwidth as defined in table 4.2.4.1.2-1. For the E-UTRA band supporting two contiguous component carriers, the limits in table 4.2.4.2.2-2 apply for frequency ranges that are more than f_{OOB} (MHz) from the edges of assigned aggregated channel bandwidth as defined in table 4.2.4.2.2-1. For the E-UTRA band supporting two contiguous component carriers, the limits in table 4.2.4.2.2-2 apply for frequency ranges that are more than f_{OOB} (MHz) from the edges of assigned aggregated channel bandwidth as defined in table 4.2.4.2.2-1. If for some frequency a spurious emission limit of a single component carrier or two contiguous component carriers overlap with the spurious emission limit or channel bandwidth of another component carrier or two contiguously aggregated carriers then it does not apply.

For the specified intra-band contiguous CA configurations for co-existence with protected bands, the measured average power of spurious emission, shall not exceed the described value in table 4.2.4.2.2-3.

E-UTRA	Spurious emission								
CA	Protected band	Frequency	/ ran	ge (MHz)	Maximum	MBW	Comment		
Configura					Level	(MHz)			
tion				-	(dBm)				
CA_1C	E-UTRA Band 1, 3, 7, 8, 20, 22, 28, 32, 38, 40, 41, 42, 43	F _{DL_low}	-	F _{DL_high}	-50	1			
CA_3C	E-UTRA Band 1, 7, 8, 20, 28, 32, 33, 34, 38, 40, 41, 43	F _{DL_low}	-	F _{DL_high}	-50	1			
	E-UTRA Band 3	F _{DL_low}	-	F _{DL_high}	-50	1	Note 3		
	E-UTRA Band 22, 42	F _{DL_low}	-	F _{DL_high}	-50	1	Note 2		
CA_8B	E-UTRA Band 1, 20, 28, 32, 33, 34, 38, 40	F _{DL_low}	-	F _{DL_high}	-50	1			
	E-UTRA Band 3	F _{DL_low}	-	F _{DL_high}	-50	1	Note 2		
	E-UTRA Band 7	F _{DL_low}	-	F _{DL_high}	-50	1	Note 2		
	E-UTRA Band 8	F _{DL_low}	-	F _{DL_high}	-50	1	Note 3		
	E-UTRA Band 22, 41, 42, 43	F_{DL_low}	-	F _{DL_high}	-50	1	Note 2		
CA_7C	E-UTRA Band 1, 3, 7, 8, 20, 22, 28, 32, 33, 34, 40, 42, 43	F _{DL_low}	-	F _{DL_high}	-50	1			
CA_38C	E-UTRA Band 1, 3, 8, 20, 28, 33, 34, 40, 42, 43	F _{DL_low}	-	F _{DL_high}	-50	1			
CA_40C	E-UTRA Band 1, 3, 7, 8, 20, 22, 28, 32, 33, 34, 38, 41, 42, 43	F _{DL_low}	-	F _{DL_high}	-50	1			
CA_41C	E-UTRA Band 1, 3, 8, 28, 34, 40, 42, 65	F _{DL_low}	-	F _{DL_high}	-50	1			
CA_42C	E-UTRA Band 1, 3, 7, 8, 20, 28, 32 33, 34, 38, 40, 41	F _{DL_low}	-	F _{DL_high}	-50	1			
NOTE 1: I	$F_{DL_{low}}$ and $F_{DL_{high}}$ refer to each E-UTRA	frequency ba	and s	specified in t	able 5.2-1, in	ETSI TS 1	36 521-1 [1].		
NOTE 2:	As exceptions, measurements with a level	up to the app	licat	ole requirem	ents defined	in table 4.2	2.4.1.2-2 shall		
	be permitted for each assigned E-UTRA ca	arrier used in	the r	neasuremer	nt due to 2 nd ,	3rd or 4th h	narmonic		
	spurious emissions. An exception is allowe	ed if there is a	t lea	st one indivi	dual RE withi	in the trans	mission		
	pandwidth (see figure 5.4.2-1 in ETSI TS 1	36 521-1 [1])	for v	which the 2 ⁿ	^d , 3 rd or 4 th h	armonic, i.	e. the		
1	requency equal to two or three times the f	requency of t	hat F	RE, is within	the Measure	ment Band	width (MBW).		
NOTE 3:	The requirement shall also apply for the fre	equency rang	es th	at are less t	han ∆f _{OOB} (N	1Hz) in tab	les 4.2.4.1.2-1		
	and 4.2.4.2.2-1 from the edge of the aggre	gated channe	el ba	ndwidth.					

Table 4.2.4.2.2-3: Spurious emission band UE co-existence limits for intra-band contiguous CA (network signalled value "NS_01")

For inter-band carrier aggregation with the uplink assigned to two E-UTRA bands, the limits in table 4.2.4.2.2-3A apply on each component carrier with all component carriers are active.

NOTE 3: For inter-band carrier aggregation with uplink assigned to two E-UTRA bands the requirements in table 4.2.4.2.2-3A could be verified by measuring spurious emissions at the specific frequencies where second and third order intermodulation products generated by the two transmitted carriers can occur; in that case, the limits for remaining applicable frequencies in table 4.2.4.2.2-3A would be considered to be verified by the measurements verifying the one uplink inter-band CA UE to UE co-existence limits.

42

		»n					
E-UTRA CA Configuration	Protected band	Freque (I	ncy MHz	range :)	Maximum Level (dBm)	MBW (MHz)	Note
CA_1A-3A	E-UTRA Band 1, 7, 8, 20, 28, 32, 38, 40, 41, 43, 65, 67	F _{DL_low}	-	F _{DL_high}	-50	1	
	E-UTRA Band 3, 34	F _{DL_low}	-	F _{DL_high}	-50	1	Note 3
	E-UTRA Band 22, 42	F _{DL low}	-	F _{DL high}	-50	1	Note 2
	Frequency range	1 880		1 895	-40	1	Notes 3, 7
	Frequency range	1 895		1 915	-15,5	5	Notes 3, 7, 8
	Frequency range	1 915		1 920	+1,6	5	Notes 3, 7, 8
CA_1A-7A	E-UTRA Band 1, 7, 8, 20, 28, 32, 40, 42, 43, 65, 67	F _{DL_low}	-	F _{DL_high}	-50	1	
	E-UTRA Band 3, 34	F _{DL_low}	-	F _{DL_high}	-50	1	Note 3
	Frequency range	1 880		1 895	-40	1	Notes 3, 7
	Frequency range	1 895		1 915	-15,5	5	Notes 3, 7, 8
	Frequency range	1 915		1 920	+1,6	5	Notes 3, 7, 8
	Frequency range	2 570	-	2 575	+1,6	5	Notes 3, 8, 9
	Frequency range	2 575	-	2 595	-15,5	5	Notes 3, 8, 9
	Frequency range	2 595	-	2 620	-40	1	Notes 3, 9
CA_1A-8A	E-UTRA Band 1, 20, 28, 32, 38, 40, 65, 67	F _{DL_low}	-	F _{DL_high}	-50	1	
	E-UTRA Band 3	F _{DL_low}	-	F _{DL_high}	-50	1	Notes 2, 3
	E-UTRA Band 7, 22, 41, 42, 43	F _{DL_low}	-	F _{DL_high}	-50	1	Note 2
	E-UTRA Band 8, 34	F _{DL_low}	-	F _{DL_high}	-50	1	Note 3
	Frequency range	860	-	890	-40	1	Notes 3, 6
	Frequency range	1 880		1 895	-40	1	Notes 3, 7
	Frequency range	1 895		1 915	-15,5	5	Notes 3, 7, 8
	Frequency range	1 915		1 920	+1,6	5	Notes 3, 7, 8
CA_1A-28A	E-UTRA Band 7, 8, 20, 32, 38, 40, 41	F _{DL_low}	-	F _{DL_high}	-50	1	
	E-UTRA Band 22, 42, 43	F _{DL low}	-	F _{DL high}	-50	1	Note 2
	E-UTRA Band 3, 34	F _{DL_low}	-	F _{DL_high}	-50	1	Note 3
	E-UTRA Band 1	F _{DL low}	-	F _{DL high}	-50	1	Notes 4, 5
	Frequency range	470	-	694	-42	8	Notes 3, 11
	Frequency range	470	-	710	-26,2	6	Note 12
	Frequency range	758	-	773	-32	1	Note 3
	Frequency range	773	-	803	-50	1	
	Frequency range	662	-	694	-26,2	6	Note 3
	Frequency range	1 880		1 895	-40	1	Notes 3, 7
	Frequency range	1 895		1 915	-15,5	5	Notes 3, 7, 8
	Frequency range	1 915		1 920	+1,6	5	Notes 3, 7, 8
	Frequency range	1 839,9	-	1 879,9	-50	1	Note 3
CA_1A-42A	E-UTRA Band 1, 7, 8, 20, 22, 28, 32, 38, 40, 41, 65, 67	F _{DL_low}	-	F _{DL_high}	-50	1	
	E-UTRA Band 3, 34	F _{DL_low}	-	F _{DL_high}	-50	1	Note 3
	Frequency range	1 880		1 895	-40	1	Notes 3, 7
	Frequency range	1 895		1 915	-15,5	5	Notes 3, 7, 8
	Frequency range	1 915		1 920	+1,6	5	Notes 3, 7, 8
	Frequency range	1 839,9	-	1 879,9	-50	1	Note 3

Table 4.2.4.2.2-3A: Spurious emission band UE co-existence limits for uplink inter-band CA (two bands)

		Spurious emission						
E-UTRA C Configurat	CA tion	Protected band	Freque (N	ncy MHz	/ range z)	Maximum Level (dBm)	MBW (MHz)	Note
CA_3A-7	7A	E-UTRA Band 1, 7, 8, 20, 28, 32, 33, 34, 40, 43, 65, 67	F_{DL_low}	-	F _{DL_high}	-50	1	
		E-UTRA Band 3	F _{DL_low}	-	F _{DL_high}	-50	1	Note 3
		E-UTRA Band 22, 42	F _{DL low}	-	F _{DL high}	-50	1	Note 2
		Frequency range	2 570	-	2 575	+1,6	5	Notes 3, 8, 9
		Frequency range	2 575	-	2 595	-15,5	5	Notes 3, 8, 9
		Frequency range	2 595	-	2 620	-40	1	Notes 3, 9
CA_3A-8	3A	E-UTRA Band 1, 20, 28, 32, 33, 34, 38, 40, 65, 67	F _{DL_low}	-	F _{DL_high}	-50	1	
		E-UTRA Band 3, 8	F _{DL_low}	-	F _{DL_high}	-50	1	Notes 2, 3
		E-UTRA Band 7, 22, 41, 42, 43	F_{DL_low}	-	F _{DL_high}	-50	1	Note 2
		Frequency range	860	-	890	-40	1	Notes 3, 6,10
CA_3A-20	0A	E-UTRA Band 1, 7, 8, 32, 33, 34, 40, 43, 65, 67	F _{DL_low}	-	F _{DL_high}	-50	1	
		E-UTRA Band 3, 20	F _{DL_low}	-	F _{DL_high}	-50	1	Note 3
		E-UTRA Band 22, 38, 42	F _{DL_low}	-	F _{DL_high}	-50	1	Note 2
		Frequency range	758	-	788	-50	1	
CA_7A-20	0A	E-UTRA Band 1,3, 7, 8, 22, 28, 32, 33, 34, 40, 43, 65, 67	F _{DL_low}	-	F _{DL_high}	-50	1	
		E-UTRA Band 20	F _{DL_low}	-	F _{DL_high}	-50	1	Note 3
		E-UTRA Band 42	F _{DL_low}	-	F _{DL_high}	-50	1	Note 2
		Frequency range	2 570	-	2 575	+1,6	5	Notes 3, 8, 9
		Frequency range	2 575	-	2 595	-15,5	5	Notes 3, 8, 9
		Frequency range	2 595	-	2 620	-40	1	Notes 3, 9
CA_7A-28	8A	E-UTRA Band 3, 5, 7, 8, 20, 34, 40, 65, 67	F _{DL_low}	-	F _{DL_high}	-50	1	
		E-UTRA Band 1, 22, 42, 43	F _{DL_low}	-	F _{DL_high}	-50	1	Note 2
		E-UTRA Band 1	F _{DL_low}	-	F _{DL_high}	-50	1	Notes 4, 5
		Frequency range	758	-	773	-32	1	Note 3
		Frequency range	773	-	803	-50	1	
		Frequency range	2 570	-	2 5/5	+1,6	5	Notes 3, 8, 9
		Frequency range	2 5 7 5	-	2 595	-15,5	2 1	Notes 3, 6, 9
NOTE 1. E.		and E refer to each E-LITE	2 595 A frequen	rv h	2 020	ied in table 1-1	I	NOIES 3, 9
NOTE 2: A an ha er ha ce	NOTE 1. P _{DL_low} and P _{DL_high} refer to each E-OTRA frequency band specified in table 1-1. NOTE 2: As exceptions, measurements with a level up to the applicable requirements defined in table 4.2.4.1.2-1 are permitted for each assigned E-UTRA carrier used in the measurement due to 2 nd , 3 rd , 4 th [or 5 th] harmonic spurious emissions. In case the exceptions are allowed due to spreading of the harmonic emission the exception is also allowed for the first 1 MHz frequency range immediately outside the harmonic emission on both sides of the harmonic emission. This results in an overall exception interval centred at the harmonic emission of (2 MHz + N x L _{CRB} x 180 kHz), where N is 2, 3 or 4 for the 2 nd , 3 rd							
	or 4 ^{er} n artially	armonic respectively. The except	ion is allow iterval	ea	If the Meas	urement Band	wiath (IVIB)	vv) totally or
NOTE 3: TI	hese i	equirements also apply for the free	equency ra	nge	es that are I	ess than for	(MHz) in	
ta	able 4.	2.4.1.2-1 and table 4.2.4.2.2-1 fro	om the eda	e o	f the addred	bated channel	, bandwidth	
NOTE 4: A	Applica hanne	ble when the assigned E-UTRA c I bandwidth used is 5 or 10 MHz.	arrier is co	onfir	ned within 7	18 MHz and 7	48 MHz ar	nd when the
NOTE 5: A	s exce	ptions, measurements with a lev	el up to the	e ap	plicable red	quirement of -3	36 dBm/MF	Iz is permitted
fo	or each	assigned E-UTRA carrier used i	in the meas	sure	ement due 1	o 3 rd harmonio	spurious	emissions.
A	An exception is allowed if there is at least one individual RB within the transmission bandwidth (see							ridth (see
fiç NOTE 6: TI	gure 3 his rea	.1-1) for which the 3 rd harmonic t quirement is applicable only for th	otally or pa e following	irtia I ca	lly overlaps ses:	the Measurer	nent Band	width (MBW).
-	- for carriers of 5 MHz channel bandwidth when carrier centre frequency (F_c) is within the range							
	902	2,5 MHz $\leq F_c < 907,5$ MHz with a	n uplink tra	Insr	nission bar	dwidth less th	an or equa	II to 20 RB;
-	for	carriers of 5 MHz channel bandw	idth when	car	rier centre f	requency (F _c)	is within th	ne range
	90	7,5 MHz ≤ F_c ≤ 912,5 MHz withou	it any restr	ictic	on on uplink	transmission	bandwidth	;
-	for	carriers of 10 MHz channel band	width wher	n ca	rrier centre	frequency (F _c) is $F_c = 9^2$	10 MHz with
an uplink transmission bandwidth less than or equal to 32 RB with $RB_{start} > 3$.								

ETSI

			Spurious emissio	ous emission							
E-UTR	A CA	Protected band	Frequency range	Maximum	MBW	Note					
Configu	Iration		(MHz)	Level	(MHz)						
				(dBm)	. ,						
NOTE 7:	This rea	quirement is applicable for any ch	annel bandwidths within	the range 1 92	0 MHz to 1	980 MHz					
	with the	following restriction: for carriers	of 15 MHz bandwidth whe	en carrier cent	re frequenc	cy is within the					
	range 1	927,5 MHz to 1 929,5 MHz and 1	for carriers of 20 MHz bai	ndwidth when	carrier cent	tre frequency					
	is withir	h the range 1 930 MHz to 1 938 M	1Hz the requirement is ap	plicable only f	or an uplinl	k transmission					
	bandwi	andwidth less than or equal to 54 RB.									
NOTE 8:	For the	se adjacent bands, the emission I	imit could imply risk of ha	Irmful interfere	nce to UE(s) operating in					
	the prot	ected operating band.									
NOTE 9:	This rea	uirement is applicable for any ch	annel bandwidths within	the range 2 50	0 MHz to 2	2 570 MHz					
	with the	following restriction: for carriers	of 15 MHz bandwidth whe	en carrier cent	re frequenc	cy is within the					
	range 2	560,5 MHz to 2 562,5 MHz and 1	for carriers of 20 MHz bai	ndwidth when	carrier cent	tre frequency					
	is withir	n the range 2 552 MHz to 2 560 M	1Hz the requirement is ap	plicable only f	or an uplinl	k transmission					
	bandwi	dth less than or equal to 54 RB.									
NOTE 10	: This rea	quirement is applicable only when	Band 3 transmission fre	quency is less	than or eq	ual to					
	1 765 N	1Hz.									
NOTE 11	: This rea	quirement is applicable in the case	e of a 10 MHz E-UTRA c	arrier confined	within 703	MHz and					
	733 M⊦	Iz, otherwise the requirement of -:	25 dBm with a measurem	nent bandwidth	of 8 MHz	applies.					
NOTE 12	: This red	quirement is applicable for 5 MHz	and 10 MHz E-UTRA ch	annel bandwic	th allocate	d within					
	718 M⊦	z to 728 MHz. For carriers of 10	MHz bandwidth, this requ	irement applie	es for an up	olink					

transmission bandwidth less than or equal to 30 RB with RBstart > 1 and RBstart < 48.

Table 4.2.4.2.2-4: Additional Spurious emission band UE co-existence limits for intra-band contiguous CA (network signalled value "CA_NS_02")

E-UTRA CA Configuration	Protected band	Frequency range (MHz)		y range z)	Maximum Level (dBm)	MBW (MHz)	Comment	
CA_1C	E-UTRA Band 34	F _{DL_low}	-	F _{DL_high}	-50	1	See note	
	Frequency range	1 900		1 915	-15,5	5	See note	
	Frequency range	1 915	-	1 920	+1,6	5	See note	
NOTE: The requirement shall also apply for the frequency ranges that are less than Δf_{OOB} (MHz) in tables 4.2.4.1.2-1								
and 4.2.	4.2.2-1 from the edge	of the age	greg	ated channe	el bandwidth.			

Table 4.2.4.2.2-5: Additional Spurious emission band UE co-existence limits for intra-band contiguous CA (network signalled value "CA_NS_05")

E-UTRA CA		Protected band	Frequency range		Maximum	MBW	Comment	
Configuration			(MHz)		Level (dBm)	(MHz)		
CA_38C		Frequency range	2 620	ı	2 645	-15,5	5	Notes 1, 2
		Frequency range	2 645	-	2 690	-40	1	Notes 1, 2
NOTE 1:	This re	equirement shall be applicable	for carrier	s wi	th aggree	gated channel ba	andwidths a	are confined in
	2 570	MHz to 2 615 MHz. For assign	ed carriers	s wi	th bandw	idths overlapping	g the frequ	ency range 2 615 MHz
	to 2 62	20 MHz the requirements apply	v with the r	nax	imum ou	tput power config	gured to +'	19 dBm.
NOTE 2:	The re	quirement shall also apply for	the freque	ncy	ranges t	hat are less than	ι Δf _{OOB} (Μ	Hz) in tables 4.2.4.1.2-1
	and 4.	2.4.2.2-1 from the edge of the	aggregate	d cl	hannel ba	andwidth.		

Table 4.2.4.2.2-6: Additional Spurious emission band UE co-existence limits for intra-band contiguous CA (network signalled value "CA_NS_06")

E-UTRA CA Configuration	Protected band	Frequency range (MHz)			Maximum Level (dBm)	MBW (MHz)	Comment	
CA_7C	Frequency range	2 570	-	2 575	+1,6	5	See note	
	Frequency range	2 575	-	2 595	-15,5	5	See note	
	Frequency range	2 595	I	2 620	-40	1	See note	
NOTE: The requirement shall also apply for the frequency ranges that are less than Δf _{OOB} (MHz) in								
tables 4	.2.4.1.2-1 and 4.2.4.2	.2-1 from	the	edge of t	he aggregated cha	nnel bandwid	lth.	

Table 4.2.4.2.2-7: Additional Spurious emission band UE co-existence limits for intra-band contiguous CA (network signalled value "CA_NS_08")

45

E-UTRA CA Configuration	Protected band	Frequency band (MHz)	Channel bandwidth/Spectrum emission limit (dBm) 5, 10, 15, 20 MHz	MBW				
CA_42C	CA_42C 42, 43 $3 400 \le f \le 3 8$		-23 (note 1)	5 MHz				
			-40 (note 2)	1 MHz				
NOTE 1: This require	ement shall app	oly within an offset be	etween 5 MHz and 25 MHz from the lo	wer and				
from the up	per edge of the	e channel bandwidth.						
NOTE 2: This require	NOTE 2: This requirement shall apply from 3 400 MHz up to 25 MHz below the lower E-UTRA channel edge							
and from 2	5 MHz above th	ne upper E-UTRA cha	annel edge up to 3 800 MHz.	-				

4.2.4.2.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.1 of the present document.

4.2.4.3 Transmitter spurious emissions for UL-MIMO

4.2.4.3.1 Definition

For UE supporting UL-MIMO, the requirements for Spurious emissions which are caused by unwanted transmitter effects such as harmonics emission, parasitic emissions, intermodulation products and frequency conversion products are specified at each transmit antenna connector.

4.2.4.3.2 Limits

For UEs with two transmit antenna connectors in closed-loop spatial multiplexing scheme, the requirements in clause 4.2.4.1.2 apply to each transmit antenna connector. The requirements shall be met with the UL-MIMO configurations specified in table 4.2.2.3.1-1.

For single-antenna port scheme, the general requirements in clause 4.2.4.1.2 apply.

4.2.4.3.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.3 of the present document.

4.2.4.4 Transmitter spurious emissions for Multi-Cluster PUSCH within a component carrier

4.2.4.4.1 Definition

For UE supporting multi cluster PUSCH within a component carrier for the operating band.

Spurious emissions are emissions which are caused by unwanted transmitter effects such as harmonics emission, parasitic emissions, intermodulation products and frequency conversion products, but exclude out-of-band emissions.

To improve measurement accuracy, sensitivity and efficiency, the resolution bandwidth may be smaller than the measurement bandwidth. When the resolution bandwidth is smaller than the measurement bandwidth, the result shall be integrated over the measurement bandwidth in order to obtain the equivalent noise bandwidth of the measurement bandwidth.

4.2.4.4.2 Limits

The spurious emission limits in table 4.2.4.1.2-2 apply for the frequency ranges that are more than Δf_{OOB} (MHz) from the edge of the channel bandwidth shown in table 4.2.4.1.2-1.

The measured average power of spurious emission for general requirements shall not exceed the described values in table 4.2.4.1.2-2.

46

4.2.4.4.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.3 of the present document.

4.2.4.5 Transmitter spurious emissions for category NB1

4.2.4.5.1 Definition

Spurious emissions are emissions which are caused by unwanted transmitter effects such as harmonics emission, parasitic emissions, intermodulation products and frequency conversion products, but exclude out-of-band emissions. The spurious emission limits are specified in terms of general requirements in line with Recommendation ITU-R SM.329-12 [i.4] and E-UTRA operating band requirement to address UE co-existence.

To improve measurement accuracy, sensitivity and efficiency, the resolution bandwidth may be smaller than the measurement bandwidth. When the resolution bandwidth is smaller than the measurement bandwidth, the result should be integrated over the measurement bandwidth in order to obtain the equivalent noise bandwidth of the measurement bandwidth.

4.2.4.5.2 Limits

When UE is configured for category NB1 uplink transmissions the requirements in clause 4.2.4.1.2 apply with an exception that boundary between category NB1 out of band and spurious domain shall be $f_{OOB} = 1.7$ MHz.

4.2.4.5.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.3 of the present document.

4.2.5 Transmitter Minimum Output Power

4.2.5.1 Transmitter minimum output power for Single Carrier

4.2.5.1.1 Definition

The minimum output power of the UE is defined as the broadband transmit power of the UE, i.e. the power in the channel bandwidth for all transmit bandwidth configurations (resource blocks), when the power is set to a minimum value.

4.2.5.1.2 Limits

The minimum output power measured shall not exceed the values specified in table 4.2.5.1.2-1.

able 4.2.5.1.2-1:	Minimum	output	power
-------------------	---------	--------	-------

	Channe	Channel bandwidth/minimum output power/measurement bandwidth							
	1,4 MHz	3,0 MHz	5 MHz	10 MHz	15 MHz	20 MHz			
Minimum output power	For carrier frequency f ≤ 3,0 GHz: ≤ -39 dBm								
Minimum output power	For carrier frequency 3,0 GHz < f \leq 4,2 GHz: \leq -38,7 dBm								
Measurement bandwidth	13,5 MHz	18 MHz							

4.2.5.1.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.4 of the present document.

4.2.5.2 Transmitter minimum output power for Carrier Aggregation (DL CA and UL CA)

4.2.5.2.1 Definition

For inter-band carrier aggregation with uplink assigned to two E-UTRA bands and intra-band contiguous carrier aggregation, the minimum output power of the UE is defined as the transmit power of the UE per component carrier, i.e. the power in the channel bandwidth of each component carrier for all transmit bandwidth configurations (resource blocks), when the power on both component carriers is set to a minimum value.

4.2.5.2.2 Limits

For inter-band carrier aggregation with uplink assigned to two E-UTRA bands, the minimum output power is defined per carrier and the limit is specified in clause 4.2.5.1.2.

For intra-band contiguous carrier aggregation, the minimum output power is defined as the mean power in one sub-frame (1 ms). The minimum output power shall not exceed the values specified in table 4.2.5.2.2-1.

	CC Channel bandwidth/Minimum output power/Measurement bandwidth							
	1,4 MHz	3,0 MHz	5 MHz	10 MHz	15 MHz	20 MHz		
Minimum output power	For carrier frequency f ≤ 3,0 GHz: ≤ -39 dBm For carrier frequency 3,0 GHz < f ≤ 4,2 GHz: ≤ -38,7 dBm							
Measurement bandwidth	-	-	4,5 MHz	9,0 MHz	13,5 MHz	18 MHz		

Table 4.2.5.2.2-1: Minimum output power for intra-band contiguous CA UE

4.2.5.2.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.4 of the present document.

4.2.5.3 Transmitter minimum output power for UL-MIMO

4.2.5.3.1 Definition

For UE supporting UL-MIMO, the minimum output power is defined as the broadband transmit power of the UE, i.e. the sum of the power in the channel bandwidth for all transmit bandwidth configurations (resource blocks) at each transmit antenna connector, when the UE power is set to a minimum value.

For UE with two transmit antenna connectors in closed-loop spatial multiplexing scheme, the minimum output power is defined as the sum of the mean power at each UE antenna connector in one sub-frame (1 ms).

4.2.5.3.2 Limits

The minimum sum of mean output power of the UE at each antenna connector measured shall not exceed the values specified in table 4.2.5.3.2-1.

	Channel bandwidth/Minimum output power/Measurement bandwidth						
	1,4 MHz	3,0 MHz	5 MHz	10 MHz	15 MHz	20 MHz	
Minimum output power	For carrier frequency f \leq 3,0 GHz: \leq -39 dBm For carrier frequency 3.0 GHz < f \leq 4.2 GHz: \leq -38,7 dBm						
Measurement bandwidth	1,08 MHz	2,7 MHz	4,5 MHz	9,0 MHz	13,5 MHz	18 MHz	

Table 4.2.5.3.2-1: Minimum output power for UL-MIMO

4.2.5.3.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.4 of the present document.

4.2.5.4 Transmitter minimum output power for category NB1

4.2.5.4.1 Definition

For category NB1 UE the single-tone and multi-tone transmission minimum output power requirement for the channel bandwidth is -40 dBm.

For 3,75 kHz sub-carrier spacing the minimum output power is defined as mean power in one slot (2 ms) excluding the 2 304Ts gap when UE is not transmitting. For 15 kHz sub-carrier spacing the minimum output power is defined as mean power in one sub-frame (1 ms).

4.2.5.4.2 Limits

The minimum output power measured shall not exceed the value of -40 dBm for the channel bandwidth of category NB1 UE.

4.2.5.4.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.4 of the present document.

4.2.6 Receiver Adjacent Channel Selectivity (ACS)

4.2.6.1 Receiver Adjacent Channel Selectivity (ACS) for Single Carrier

4.2.6.1.1 Definition

Adjacent Channel Selectivity (ACS) is a measure of a receiver's ability to receive an E-UTRA signal at its assigned channel frequency in the presence of an adjacent channel signal at a given frequency offset from the centre frequency of the assigned channel. ACS is the ratio of the receive filter attenuation on the assigned channel frequency to the receive filter attenuation on the adjacent channel(s).

4.2.6.1.2 Limits

The throughput R_{av} shall be ≥ 95 % of the maximum throughput of the reference measurement channels as specified in ETSI TS 136 521-1 [1] under the conditions specified in table 4.2.6.1.2-2 and also under the conditions specified in table 4.2.6.1.2-3.

		Channel bandwidth						
Rx Parameter	Units	1,4 MHz	3 MHz	5 MHz	10 MHz	15 MHz	20 MHz	
ACS	dB	33,0	33,0	33,0	33,0	30	27	

Table 4.2.6.1.2-1: Adjacent channel selectivity

Dy Devenetor	Unite			Channel ba	andwidth		
RX Parameter	Units	1,4 MHz	3 MHz	5 MHz	10 MHz	15 MHz	20 MHz
Power in			- -			•	•
Transmission	dDm			DEESENS			
Bandwidth	UDITI			REFSENS	+ 14 UD		
Configuration							
_	dBm	REFSENS	REFSENS	REFSENS	REFSENS	REFSENS	REFSENS
PInterferer		+45,5 dB	+45,5 dB	+45,5 dB	+45,5 dB	+42,5 dB	+39,5 dB
BW _{Interferer}	MHz	1,4	3	5	5	5	5
F _{Interferer} (offset)	MHz	1,4025	3,0075	5,0025	7,5075	10,0125	12,5025
NOTE 1: The tran	smitter s	hall be set to 4	dB below P _{CN}	MAX OF PCMAX	L CA as define	d in clause 6.2	.5 in ETSI
TS 136	101 [3].						
NOTE 2: The inte	rferer sha	all consist of the	e Reference m	neasurement cha	annel specified	in clause A.3	2 of ETSI
TS 136	521-1 [1]	with set-up acc	cording to clau	se C.3.1 of ETS	I TS 136 521-	1 [1].	
NOTE 3: REFSEM	VS as def	ined in clause	7.3.3 in ETSI	TS 136 521-1 [1].		
NOTE 4: For DL of	ategory l	M1 UE, the refe	erence sensitiv	vity for category	M1 in ETSI TS	S 136 521-1 [1]	,
tables 7	.3EA-1 ar	nd 7.3EA-2 sho	uld be used a	s REFSENS for	the power in T	ransmission E	Bandwidth
Configu	ation and	PInterferer					
NOTE 5 For DL	ategory I	M1 LIF the nar	ameters for th	e annlicable cha	annel handwidt	h apply	

Table 4.2.6.1.2-2: Test parameters for Adjacent channel selectivity, Case 1

49

Table 4.2.6.1.2-3: Test parameters for Adjacent channel selectivity, Case 2

Dy Deremeter	Unite			Channel ba	andwidth					
KX Parameter	Units	1,4 MHz	3 MHz	5 MHz	10 MHz	15 MHz	20 MHz			
Power in Transmission Bandwidth Configuration	dBm	-56,5	-56,5	-56,5	-56,5	-53,5	-50,5			
PInterferer	dBm	-25								
BW _{Interferer}	MHz	1,4	3	5	5	5	5			
F _{Interferer} (offset)	MHz	1,4025	3,0075	5,0025	7,5075	10,0125	12,5025			
NOTE 1: The tra	NOTE 1: The transmitter shall be set to 24 dB below PCMAX OF PCMAX CA as defined in clause 6.2.5 in									
ETSI NOTE 2: The in TS 13	ETSI TS 136 101 [3]. VOTE 2: The interferer shall consist of the Reference measurement channel specified in clause A.3.2 of ETSI TS 136 521-1 [1] with set-up according to clause C.3.1 of ETSI TS 136 521-1 [1]									

4.2.6.1.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.5.1 of the present document.

4.2.6.2 Receiver Adjacent Channel Selectivity (ACS) for Carrier Aggregation in DL-only bands

4.2.6.2.1 Definition

Adjacent Channel Selectivity (ACS) is a measure of a receiver's ability to receive an E-UTRA signal at its assigned channel frequency in the presence of an adjacent channel signal at a given frequency offset from the centre frequency of the assigned channel. ACS is the ratio of the receive filter attenuation on the assigned channel frequency to the receive filter attenuation on the adjacent channel(s).

For inter-band carrier aggregation with uplink assigned to one E-UTRA band, the adjacent channel requirements are defined with the uplink active on the band other than the band whose downlink is being tested. The UE shall meet the requirements specified in clause 4.2.6.1.2 for each component carrier while both downlink carriers are active, or the requirement specified in table 4.2.6.2.2-1 with the test parameters specified in table 4.2.6.2.2-2 and table 4.2.6.2.2-3 for operating band 46. For E-UTRA CA configurations including an operating band without uplink band (as noted in table 1-1), the requirements for both downlinks shall be met with the uplink active in the band capable of UL operation.

Table 4.2.6.2.2-1: Ac	ljacent channel selectivity	/ for Band 46
-----------------------	-----------------------------	---------------

E-UTRA band	Rx Parameter	Units	Channel bandwidth						
			1,4 MHz	3 MHz	5 MHz	10 MHz	15 MHz	20 MHz	
46	ACS	dB	-	-	-	-	-	27	

Table 4.2.6.2.2-2: Test parameters for Adjacent channel selectivity for Band 46, Case 1

E-UTRA Band	Rx Parameter	Units	Channel bandwidth							
			1,4 MHz	3 MHz	5 MHz	10 MHz	15 MHz	20 MHz		
	Power in Transmission Bandwidth Configuration	dBm			REFSEN	S + 14 dB				
46	P _{Interferer}	dBm	-	-	-	-	-	REFSENS +39,5 dB		
	BW _{Interferer}	MHz	-	-	-	-	-	20		
	F _{Interferer} (offset)	MHz	-	-	-	-	-	20 + 0,0025 / -20 - 0,0025		
NOTE 1: The tran	smitter shall be s	et to 4 dE	3 below P _{CM}	AX L at the	minimum upl	ink configura	ation specifi	ed in ETSI		
TS 136 ⁻	101 [3] (table 7.3.	3-2 with	P _{CMAX L} as o	defined in c	lause 6.2.5).					
NOTE 2: The interferer shall consist of the Reference measurement channel specified in table A3.2-2c of ETSI TS 136 521-1 [1] with one sided dynamic OCNG Pattern OP.1 -F33 as described in clause A.5.4.1 of ETSI TS 136 521-1 [1] and set-up according to clause C.3.1 of ETSI TS 136 521-1 [1].										

Table 4.2.6.2.2-3: Test parameters for Adjacent channel selectivity for Band 46, Case 2

E-UT	RA band	Rx	Units	Channel bandwidth							
		Parameter		1,4 MHz	3 MHz	5 MHz	10 MHz	15 MHz	20 MHz		
	Power in Transmission Bandwidth Configuration	dBm	-	-	-	-	-	-50,5			
	46	PInterferer	dBm			-2	5				
		BW _{Interferer}	MHz	-	-	-	-	-	20		
	F _{Interferer} (offset)	MHz	-	-	-	-	-	20 + 0,0025 / -20 - 0,0025			
NOTE 1:	The transmitt	er shall be set to	24 dB b	elow P _{CMAX}	L at the min	imum uplink	configuratio	on specified	in ETSI		
	TS 136 101 [3] (table 7.3.3-2	with P _{CN}	_{IAX L} as defin	ed in clause	e 6.2.5).					
NOTE 2:	DTE 2: The interferer shall consist of the Reference measurement channel specified in clause A.3.2 of ETSI TS 136 521-1 [1] with one sided dynamic OCNG Pattern OP.1 F33 as described in clause A.5.1 of ETSI TS 136 521-1 [1] and set-up according to clause C.3.1 of ETSI TS 136 521-1 [1].										

4.2.6.2.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.5.2 of the present document.

4.2.6.3 Receiver Adjacent Channel Selectivity (ACS) for category NB1

4.2.6.3.1 Definition

Adjacent Channel Selectivity (ACS) is a measure of a receiver's ability to receive a signal at its assigned channel frequency in the presence of an adjacent channel signal at a given frequency offset from the centre frequency of the assigned channel. ACS is the ratio of the receive filter attenuation on the assigned channel frequency to the receive filter attenuation on the adjacent channel(s).

51

4.2.6.3.2 Limits

Category NB1 UE shall fulfil the minimum requirement specified in table 4.2.6.3.2-1 for all values of an adjacent channel interferer up to -25 dBm. However it is not possible to directly measure the ACS, instead, the lower and upper range of test parameters are chosen in table 4.2.6.3.2-1 where the throughput shall be \geq 95 % of the maximum throughput of the reference measurement channel as specified in clause A.3.2 of ETSI TS 136 521-1 [1].

ACS1 test Parameters										
Interferer	GSM (GMSK)	E-UTRA								
Category NB1 signal power (P _{wanted}) / dBm	REFSENS + 14 dB									
interferer signal power (P _{Interferer}) / dBm	REFSENS + 42 dB	REFSENS + 47 dB								
Interferer bandwidth	200 kHz	5 MHz								
Interferer offset from category NB1 channel edge	±200 kHz	±2,5 MHz								
ACS2 test Para	meters									
Interferer	GSM (GMSK)	E-UTRA								
Category NB1 signal power (P _{wanted}) / dBm	-53 dBm	-58 dBm								
interferer signal power (P _{Interferer}) / dBm	-25 dBm									
Interferer bandwidth	200 kHz	5 MHz								
Interferer offset from category NB1 channel edge	±200 kHz	±2,5 MHz								

Table 4.2.6.3.2-1: Test parameters for Adjacent channel selectivity, category NB1

4.2.6.3.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.5.3 of the present document.

4.2.7 Receiver Blocking Characteristics

4.2.7.1 Receiver Blocking Characteristics for Single Carrier

4.2.7.1.1 Definition

The blocking characteristic is a measure of the receiver's ability to receive a wanted signal at its assigned channel frequency in the presence of an unwanted interferer on frequencies other than those of the spurious response or the adjacent channels, without this unwanted input signal causing a degradation of the performance of the receiver beyond a specified limit. The blocking performance shall apply at all frequencies except those at which a spurious response occurs.

4.2.7.1.2 Limits

With parameters specified in tables 4.2.7.1.2-1 and 4.2.7.1.2-2, the throughput shall be \geq 95 % of the maximum throughput of the reference measurement channels as specified in clauses A.2.2, A.2.3 and A.3.2 (with one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in clauses A.5.1.1 and A.5.2.1) in ETSI TS 136 521-1 [1].

With parameters specified in tables 4.2.7.1.2-3 and 4.2.7.1.2-4, the throughput shall be \geq 95 % of the maximum throughput of the reference measurement channels as specified in clauses A.2.2, A.2.3 and A.3.2 (with one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in clauses A.5.1.1 and A.5.2.1) in ETSI TS 136 521-1 [1], except for the spurious response frequencies.

For table 4.2.7.1.2-4 in frequency ranges 1, 2 and 3, up to $\max(24, 6 \cdot \lceil N_{RB} / 6 \rceil)$ exceptions are allowed for spurious

52

response frequencies in each assigned frequency channel when measured using a 1 MHz step size, where N_{RB} is the number of resource blocks in the downlink transmission bandwidth configuration. For these exceptions the requirements of clause 4.2.8.1 Spurious response are applicable.

With parameters specified in table 4.2.7.1.2-5, the throughput shall be \geq 95 % of the maximum throughput of the reference measurement channels as specified in clauses A.2.2, A.2.3 and A.3.2 (with one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in clauses A.5.1.1 and A.5.2.1) in ETSI TS 136 521-1 [1].

By Parameter	Unito	Channel bandwidth									
RX Farameter	Units	1,4 MHz	3 MHz	5 MHz	10 MHz	15 MHz	20 MHz				
Power in			REFSENS + channel bandwidth specific value below								
Transmission Bandwidth Configuration	dBm	6	6	6	6	7	9				
BW _{Interferer}	MHz	1,4	3	5	5	5	5				
Floffset, case 1	MHz	2,1125	4,5075	7,5125	7,5025	7,5075	7,5125				
Floffset, case 2	MHz	3,5075	7,5075	12,5075	12,5125	12,5025	12,5075				
NOTE 1: The tra	ansmitter s	hall be set to 4	l dB below P _{CN}	MAX L at the mini	imum uplink co	onfiguration sp	ecified in				
ETSI T	S 136 101	[3] (table 7.3.	1-2 with P _{CMAX}	$_{\rm K~L}$ as defined in	clause 6.2.5).						
NOTE 2: The int	terferer sha	all consist of th	e Reference n	neasurement cha	annel specified	l in clause A.3	.2 of ETSI				
TS 136	6 521-1 [1]	with a set-up a	according to cl	ause C.3.1 of E	TSI TS 136 52	1-1 [1].					
NOTE 3: REFS	ENS as det	fined in clause	7.3.3 in ETSI	TS 136 521-1 [1].						
NOTE 4: For DL	category	M1 UE, the ref	erence sensiti	vity for category	M1 in ETSI TS	3 136 521-1 [1]],				
tables	7.3EA-1 a	nd 7.3EA-2 sh	ould be used a	is REFSENS for	the power in T	ransmission E	Bandwidth				
Config	uration.										
NOTE 5: For DL	category	M1 UE, the pa	rameters for th	ne applicable cha	annel bandwidt	h apply.					

Table 4.2.7.1.2-1: In-band blocking parameters

	Parameter	Units	Case 1	Case 2
	PInterferer	dBm	-56	-44
E-UTRA band	F _{Interferer}		= -BW/2 - F _{loffset, case 1}	≤ -BW/2 - F _{loffset, case 2}
	(Offset)		= +BW/2 + F _{loffset, case 1}	≥ +BW/2 + F _{loffset, case 2}
1, 3, 7, 8, 20, 22, 28, 31, 33, 34, 38, 40, 41, 42, 43, 65, 68, 72, 87, 88	F _{Interferer}	MHz	(note 2)	F _{DL_low} - 15 to F _{DL_high} + 15
NOTE 1: For certain b band, but wi NOTE 2: For each ca a) the carrie	bands, the unwanter ithin the first 15 MH rrier frequency the r er frequency - BW/2	d modulate z below or requiremen 2 - Floffset	ed interfering signal may not fa above the UE receive band. nt is valid for two frequencies: , case 1; and	II inside the UE receive
b) the carri NOTE 3: F _{Interferer} rar	er frequency + BW/2 nge values for unwa	2 + Floffse	et, case 1. Ilated interfering signal are inte	erferer centre frequencies.

Table 4.2.7.1.2-2: In-band blocking

By Parameter	Channel bandwidth							
KX Faraineter	I,4 MHz 3 MHz 5 MHz 10 MHz 15 MH						20 MHz	
Power in Transmission	dPm	REFSE	NS + chanr	nel bandw	idth speci	fic value b	elow	
Bandwidth Configuration	UDIII	6	6	6	6	7	9	
NOTE 1: The transmitter sh	all be set to	4 dB below F	⊃ _{CMAXL} att	he minimu	um uplink	configurat	tion	
specified in ETSI	TS 136 101 [3] (table 7.3.	1-2 with P _C	_{MAX L} as (defined in	clause 6.2	2.5).	
NOTE 2: Reference measu	rement chan	nel is clause	A.3.2 of ET	'SI TS 136	6 521-1 [1]].		
NOTE 3: REFSENS as defi	ned in clause	e 7.3.3 in ET	SI TS 136 5	521-1 [1].				
NOTE 4: For DL category M	11 UE, the re	ference sen	sitivity for ca	ategory M ²	1 in ETSI	TS 136 52	21-1 [1],	
tables 7.3EA-1 an	d 7.3EA-2 sh	ould be use	d as REFSE	ENS for th	e power ir	n Transmis	ssion	
Bandwidth Config	uration.							
NOTE 5: For DL category M	11 UE, the pa	arameters fo	r the applica	able chanr	nel bandw	idth apply.		

Table 4.2.7.1.2-3: Out-of-band blocking parameters

Table 4.2.7.1.2-4: Out-of-ba	and blocking
------------------------------	--------------

	Parameter	Unite	Frequency					
E-UTRA band	Faranielei	Units	Range 1	Range 2	Range 3			
	P _{Interferer} dBm -44		-44	-30	-15			
1, 3, 7, 8, 20, 22, 28,			F _{DL_low} - 15 to	F _{DL_low} - 60 to	F _{DL_low} - 85 to			
31, 33, 34, 38, 40,	F (CW/)		F _{DL_low} - 60	F _{DL_low} - 85	1 MHz			
(note 2), 65, 68, 72.	Interferer	IVIEZ	F _{DL_high} + 15 to	F _{DL_high} + 60 to	F _{DL_high} + 85 to			
87, 88			F _{DL_high} + 60	F _{DL_high} + 85	+12 750 MHz			
NOTE 1: Range 3 sha	all be tested only wi	th the high	nest channel bandw	<i>v</i> idth.				
NOTE 2: The power level of the interferer (P _{Interferer}) for Range 3 shall be modified to -20 dBm for F _{Interferer}								
> 2 800 MHz and F _{Interferer} < 4 400 MHz.								

Deremeter	Unito	Channel Bandwidth									
Parameter	Units	1,4 MHz	3 MHz	5 MHz	10 MHz	15 MHz	20 MHz				
Р	dBm	P _{REFSENS} + channel-bandwidth specific value below									
·w	abiii	22	18	16	13	14	16				
P _{uw} (CW)	dBm	-55	-55	-55	-55	-55	-55				
F _{uw} (offset for	MHz	0.9075	1.7025	2.7075	5.2125	7.7025	10.2075				
<i>∆f</i> = 15 kHz)		-,	.,	_,	-,	.,	,				
NOTE 1: The tran	smitter shall be	e set a 4 dB l	below P _{CMA}	AX L at the	minimum up	link configura	ation				
specified	d in ETSI TS 13	36 101 [3] (ta	ble 7.3.1-2	with P _{CMA}	_{XL} as define	ed in clause 6	6.2.5).				
NOTE 2: Referen	ce measureme	nt channel is	in clause A	A.3.2 of ET	SI TS 136 52	21-1 [1].					
NOTE 3: REFSEM	NS as defined i	n clause 7.3.	.3 in ETSI T	⁻ S 136 521	-1 [1].						
NOTE 4: For DL of	ategory M1 UE	E, the referer	nce sensitiv	ity for cate	gory M1 in E	TSI TS 136 9	521-1 [1],				
tables 7	.3EA-1 and 7.3	EA-2 should	be used as	PREFSENS	for P _w .						
NOTE 5: For DL of	category M1 UE	E, the parame	eters for the	e applicable	e channel ba	ndwidth app	ly.				

4.2.7.1.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.6.1 of the present document.

4.2.7.2 Receiver Blocking Characteristics for Carrier Aggregation in DL-only bands

4.2.7.2.1 Definition

The blocking characteristic is a measure of the receiver's ability to receive a wanted signal at its assigned channel frequency in the presence of an unwanted interferer on frequencies other than those of the spurious response or the adjacent channels, without this unwanted input signal causing a degradation of the performance of the receiver beyond a specified limit. The blocking performance shall apply at all frequencies except those at which a spurious response occurs.

4.2.7.2.2 Limits

With parameters specified in tables 4.2.7.1.2-1 and 4.2.7.2.2-1, or table 4.2.7.2.2-1a and table 4.2.7.2.2-1b for band combinations including operating band 46 without uplink operation, the throughput on SCC shall be \geq 95 % of the maximum throughput of the reference measurement channels as specified in clauses A.2.2, A.2.3 and A.3.2 (with one sided dynamic OCNG Pattern OP.1 FDD/TDD/FS3 for the DL-signal as described in clauses A.5.1.1, A.5.2.1 and A.5.4.1) in ETSI TS 136 521-1 [1].

With parameters specified in tables 4.2.7.1.2-3 and 4.2.7.2.2-2, or table 4.2.7.1.2-3 and table 4.2.7.2.2-2a for band combinations including operating band 46 without uplink operation, the throughput on SCC shall be \geq 95 % of the maximum throughput of the reference measurement channels as specified in clauses A.2.2, A.2.3 and A.3.2 (with one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in clauses A.5.1.1 and A.5.2.1) in ETSI TS 136 521-1 [1], except for the spurious response frequencies.

For table 4.2.7.2.2-2 or table 4.2.7.2.2-2a in frequency range 1, 2 and 3, up to $\max(24, 6 \cdot \lfloor N_{RB} / 6 \rfloor)$ exceptions are

allowed for spurious response frequencies in each assigned frequency channel when measured using a 1 MHz step size, where N_{RB} is the number of resource blocks in the downlink transmission bandwidth configuration. For these exceptions the requirements of clause 4.2.8.2 Spurious response are applicable.

With parameters specified in table 4.2.7.1.2-5, the throughput on SCC shall be \geq 95 % of the maximum throughput of the reference measurement channels as specified in clauses A.2.2, A.2.3 and A.3.2 (with one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in clauses A.5.1.1 and A.5.2.1) in ETSI TS 136 521-1 [1].

	Parameter	Units	Case 1	Case 2			
	PInterferer	dBm	-56	-44			
E-UTRA band	F		= -BW/2 - F _{loffset,case 1}	≤ -BW/2 - F _{loffset,case 2}			
	· Interferer	MHz	&	&			
	(onset)		= +BW/2 + F _{loffset,case 1}	≥ +BW/2 + F _{loffset,case 2}			
				F _{DL_low} - 15			
32	F _{Interferer}	MHz	(note 2)	to			
				F _{DL_high} + 15			
NOTE 1: For certain bands	s, the unwanted m	nodulated in	terfering signal may not fa	II inside the UE receive			
band, but within t	the first 15 MHz b	elow or abo	ve the UE receive band.				
NOTE 2: For each carrier	frequency the req	uirement is	valid for two frequencies:				
a) the carrier	a) the carrier frequency -BW/2 - F _{loffset, case 1;} and						
b) the carrier	b) the carrier frequency +BW/2 + $F_{loffset, case 1}$.						
NOTE 3: F _{Interferer} range v	alues for unwante	d modulate	d interfering signal are inte	erferer centre			
frequencies.							

Table 4.2.7.2.2-1a: In band	blocking parameters for	carrier aggregation with I	3and 46
-----------------------------	-------------------------	----------------------------	---------

E-UTRA	Rx parameter	Units	Channel bandwidth					
band			1,4 MHz	3 MHz	5 MHz	10 MHz	15 MHz	20 MHz
Power in REFSENS + c					channel band	dwidth specif	ic value belo	w
46	Transmission Bandwidth Configuration	dBm	-	-	-	-	-	9
(note 2)	BW _{Interferer}	MHz	-	-	-	-	-	20
	Floffset, case 1	MHz	-	-	-	-	-	30 + 0,0125
	F _{loffset, case 2}	MHz	-	-	-	-	-	50 + 0,0075
NOTE 1:	The transmitter sha	all be set	to 4 dB belov	v P _{CMAXL} at	the minimur	m uplink cont	figuration sp	ecified in
	ETSI TS 136 101 [3] (table 3	7.3.1-2 with F	P _{CMAX L} as d	efined in cla	use 6.2.5).		
NOTE 2:	 NOTE 2: The interferer consists of the Reference measurement channel specified in clause A.3.2 of ETSI TS 136 521-1 [1] with one sided dynamic OCNG Pattern OP.1 FDD/TDD/FS3 as described in clauses A.5.1.1, A.5.2.1 and A.5.4.1 of ETSI TS 136 521-1 [1] and set-up according to clause C.3.1 of ETSI TS 136 521-1 [1]. 							

E-UTRA band		Parameter	Unit	Case 1	Case 2		
		PInterferer	dBm	-50	-44		
				= -BW/2 - F _{loffset,case 1}	≤ -BW/2 - F _{loffset,case 2}		
		F _{Interferer} (onset)	MHz	& = +BW/2 + F _{loffset,case 1}	& ≥ +BW/2 + F _{loffset,case 2}		
					F _{DL_low} - 60		
46		F _{Interferer}	MHz	(note 2)	to		
					F _{DL_high} + 60		
NOTE 1:	For cer	tain bands, the unwa	anted modu	lated interfering signal ma	y not fall inside the UE		
	receive	band, but within the	e first 15 MH	lz or 60 MHz below or abo	ve the UE receive band.		
NOTE 2:	For eac	h carrier frequency	the requirer	nent is valid for two freque	encies:		
	a. the c	arrier frequency - B	W/2 - F _{loffset}	t, case 1; and			
	b. the carrier frequency + BW/2 + F _{loffset. case 1.}						
NOTE 3:	F _{Interfere}	F _{Interferer} range values for unwanted modulated interfering signal are interferer centre					
	frequen	cies.					

Table 4.2.7.2.2-1b: In-band blocking for carrier aggregation with Band 46

Table 1 2 7 2 2-2. Out-of-band blockin	r for inter-hand carrier	and a second second second	ona activa unlink
	g ior inter-band carrier	aggregation with	one active uplink

Parame	ter Units	s Range 1		Range 2	Range 3		
P _{wanted}	dBm	1	Table 4.2	2.7.1.2-3 for all component	carriers		
P _{interferer}	dBm	$-44 + \Delta R_{IB}$,c	-30 + ∆R _{IB,c}	-15 + ∆R _{IB,c}		
F _{interferer}	MHz	-60 < f - F _{DL_Low}	(₎₎ < -15	-85 < f - F _{DL_Low(j)} ≤ -60	1 ≤ f ≤ F _{DL_Low(1)} - 85		
(CW)		or		or	or		
		15 < f - F _{DL_High}	₍₎₎ < 60	60 ≤ f - F _{DL_High(<i>j</i>) < 85}	F _{DL_High(<i>j</i>)} + 85 ≤ f		
					≤ F _{DL_Low(<i>j</i>+1)} - 85		
					or		
					$F_{DL High(X)} + 85 \le f$		
					≤ 12 750		
NOTE 1:	1: F _{DL Low(i)} and F _{DL High(i)} denote the respective lower and upper frequency limits of the operating						
	band con	taining carrier $j, j = 1,$, X, with ca	arriers numbered in increas	ing order of carrier frequency		
	and X the	number of componen	t carriers in	the band combination (X =	2 or $X = 3$ for the present		
	document	t).					
NOTE 2:	For F _{DL L}	$_{ow(i+1)} - F_{DI - High(i)} < 14$	15 MHz and	$F_{\text{Interferer}}$ in F_{DI} High(i) < f <	$F_{DI I ow(i+1)}, F_{Interferer}$ can be		
	in both Ra	ange 1 and Range 2.	Then the lov	ver of the Platerform applies.			
NOTE 3	For F	15 MHz < f < F	+ 15	5 MHz the appropriate adia	cent channel selectivity and		
NOTE 0.	$DL_{Low(j)}$ is write $2 + 2 + DL_{High(j)}$ is write appropriate adjacent channel selectivity and						
	in-band blocking requirements in the respective clauses 4.2.6.2.3 and 4.2.7.2.3 shall be applied for						
	carrier <i>j</i> .						
NOTE 4:	∆R _{IB,c} acc	cording to table 7.3.3-1	IA of ETSI 7	IS 136 521-1 [1] shall apply	when serving cell <i>c</i> is		
	measured	ł.					

	A Deremeter	l lm it	Denge 1	Banga 2	Banga 2		
E-UIRA C	A Parameter	Unit	Range I	Range z	Range 3		
Configurat	ion						
	Pwanted	dBm	Table 4.2.7.1.2-3 for all component carriers				
	Pinterferer	dBm	-44 + ∆R _{IB,c}	-30 + ∆R _{IB,c}	-15 + ∆R _{IB,c}		
CA_1A-46	Α,			,	(note 5)		
CA_3A-46	A, F _{interferer}	MHz	-60 < f - F _{DL_Low(<i>j</i>)} < -15	-85 < f - F _{DL_Low(j)} ≤ -60	1 ≤ f ≤ F _{DL_Low(<i>j</i>)} - 85		
CA_7A-46	A, (CW)		with j ≤ K	or	or		
CA_42A-46	δA		or	60 ≤ f - F _{DL High()} < 85	F _{DL High(/)} + 85 ≤ f		
			15 < f - F _{DL_High(/)} < 60		≤ 12 750		
			with <i>j</i> ≤ K				
NOTE 1: F _I	_{DL Low()} and F _{DL Hig}	_{h(i)} , j = 1	,,K,N, denote the respec	tive lower and upper frequ	ency limits of the (non-		
0\	/erlapping) operating	gbands	of the CA configuration num	pered in increasing order o	f frequency, with N the		
ทเ	umber of bands in th	e band o	combination and K the numb	er of bands with F _{DL High} ≤	3 600 MHz		
(K	C = 1 and $N = 2$ in th	is versio	n of the present document).	<u> </u>			
NOTE 2: Fo	or F _{DL Low()} - 15 MH	lz≤f≤F	DI High() + 15 MHz the appro	priate adjacent channel se	electivity and in-band		
bl	ocking requirements	s shall be	e applied for carrier $j = 1$.				
NOTE 3: Fo	or F _{DL Low(N)} - 60 MI	Hz≤f≤l	F _{DL High(N)} + 60 MHz the app	ropriate adjacent channel	selectivity and in-band		
bl	ocking requirements	s shall be	e applied for carrier $N = 2$.				
NOTE 4: ΔΙ	E 4: ΔR_{IP} according to table 7.3.1-1A of ETSI TS 136 521-1 [1] applies when serving cell c is measured.						
NOTE 5: TH	TE 5: The power level (Putoferr) for Range 3 is modified to -20 dBm for Futoferr > 4 400 MHz except for band						
	mbinations with Ra	nd 42 for	which P. for Range 3	is modified to -20 dBm for	F. > 2 800 MHz		
	interferer of Marye 313 mouned to -20 dbm for 1 Interferer > 2 000 Milz.						

Table 4.2.7.2.2-2a: Out-of-band blocking for inter-band carrier aggregation with band 46

Table 4.2.7.2.2-3: Narrow-band blocking

Deremeter	l lmit	CA Bandwidth Class					
Parameter	Unit	В	C	D	E	F	
Pw in Transmission Bandwidth	dDm	REFSENS +	- CA Bandwidth	Class specif	ic value bel	ow	
Configuration, per CC	UDIII	16	16 ⁴	16	16	16	
Puw (CW)	dBm	-55	-55	-55	-55	-55	
F _{uw} (offset for ⊿f = 15 kHz)	MHz	- F _{offset} – 0.2 / + F _{offset} + 0.2	- F _{offset} - 0.2 / + F _{offset} + 0.2	- F _{offset} - 0.2 / + F _{offset} + 0.2	- F _{offset} - 0.2 / + F _{offset} + 0.2	- F _{offset} - 0.2 / + F _{offset} + 0.2	
F _{uw} (offset for ⊿f = 7,5 kHz)	MHz						
NOTE 1: The transmitter shall b TS 136 521-1 [1]. NOTE 2: Reference measureme	e set to 4 c ent channe	B below PCMAX_LOF PCMAX_L	as defined in cl	ause 6.2.5A ed dvnamic C	of ETSI CNG Patter	n OP.1	

FDD/TDD as described in clauses A.5.1.1/A.5.2.1 of ETSI TS 136 521-1 [1]. NOTE 3: The F_{uw} (offset) is the frequency separation of the centre frequency of the carrier closest to the interferer and the centre frequency of the interferer and shall be further adjusted to $[F_{interferer}/0.015 + 0.5]0.015 + 0.0075$ MHz to be offset from the sub-carrier raster.

NOTE 4: The requirement is applied for the band combinations whose component carriers' BW \geq 5 MHz.

For E-UTRA CA configurations with a component carrier assigned in Band 46, narrow-band blocking requirements do not apply in the presence of a narrow-band interferer in Band 46.

4.2.7.2.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.6.2 of the present document.

	-	1			1
E-UTRA CA	Parameter	Unit	Range 1	Range 2	Range 3
Configuration			-		_
	Pwanted	dBm	Table 4.2.7	7.1.2-3 for all component ca	arriers
CA_1A-46A,	Pinterferer	dBm	-44 + ΔR _{IR α}	$-30 + \Delta R_{IRc}$	$-15 + \Delta R_{IRc}$
CA_2A-46A,	interierer		10,0	10,0	(note 5)
CA_3A-46A,	F		60 - f E - 15	95 - f E - 60	
CA_4A-46A,	^r interferer		$-60 < 1 - F_{DL_{Low}(j)} < -15$	$-60 < 1 - F_{DL_{Low}(j)} \leq -60$	$1 \le 1 \le P_{DL_{Low}(j)} = 05$
CA_7A-46A,	(CW)		with <i>j</i> ≤ K	or	or
CA_41A-46A,			or	60 ≤ f - F _{DL High(/)} < 85	F _{DL Hiah(i)} + 85 ≤ f
CA_42A-46A			15 < f - F _{DL High(<i>i</i>)} < 60	_ 0 0,	_≤ 12 750
			with j ≤ K		
NOTE 1: F _{DL}	ow(i) and FDI Hig	h(i), j = 1	,,K,N, denote the respec	tive lower and upper freque	ency limits of the
(non-	overlapping) ope	erating b	ands of the CA configuration	numbered in increasing or	der of frequency, with
N the	number of band	ls in the	band combination and K the	number of bands with F	Link ≤ 3 600 MHz
(K -	1 and N = 2 in th		n of the present decument)	DL_	_High
			15 MHz the appre	priata adiacant channel co	loctivity and in hand
NOTE 2. FOIR	DL_Low(j) - 15 Will		DL_High() + 15 Wil 12 the appro	phate aujacent channel se	electivity and in-band
block	ing requirements	s shall be	e applied for carrier $j = 1$.		
NOTE 3: For F		Hz≤f≤	F _{DL_High(N)} + 60 MHz the app	ropriate adjacent channel	selectivity and in-band
blocking requirements shall be applied for carrier N = 2.					
NOTE 4: $\Delta R_{IIII} = \alpha a c c c c c c c c c c c c c c c c c c$					
NOTE 5: The	ower level (P _{inte}	rforor) for	Range 3 is modified to -20 c	IBm for $F_{Interferer} > 4400 \text{ M}$	1Hz except for band
	\ IIIe		0	Interielei	•

Table 4.2.7.2.3-1:	Out-of-band blockin	g for inter-band	carrier aggregation	with Band 46

4.2.7.3 Receiver Blocking Characteristics for category NB1

4.2.7.3.1 Definition

The blocking characteristic is a measure of the receiver's ability to receive a wanted signal at its assigned channel frequency in the presence of an unwanted interferer on frequencies other than those of the spurious response or the adjacent channels, without this unwanted input signal causing a degradation of the performance of the receiver beyond a specified limit. The blocking performance shall apply at all frequencies except those at which a spurious response occurs.

combinations with Band 42 for which $P_{Interferer}$ for Range 3 is modified to -20 dBm for $F_{Interferer} > 2 800$ MHz.

4.2.7.3.2 Limits

With parameters specified in table 4.2.7.3.2-1, the throughput shall be \geq 95 % of the maximum throughput of the reference measurement channels as specified in clauses A.2.2, A.2.3 and A.3.2 (with one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in clauses A.5.1.1 and A.5.2.1) in ETSI TS 136 521-1 [1].

With parameters specified in table 4.2.7.3.2-2, the throughput shall be \geq 95 % of the maximum throughput of the reference measurement channels as specified in clauses A.2.2, A.2.3 and A.3.2 (with one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in clauses A.5.1.1 and A.5.2.1) in ETSI TS 136 521-1 [1], except for the spurious response frequencies.

For table 4.2.7.3.2-2 in frequency range 1, 2 and 3, up to $\max(24, 6 \cdot \lceil N_{RB} / 6 \rceil)$ exceptions are allowed for spurious

response frequencies in each assigned frequency channel when measured using a 1 MHz step size, where N_{RB} is the number of resource blocks in the downlink transmission bandwidth configuration. For these exceptions the requirements of clause 4.2.8.3 Spurious response are applicable.

IBB1 test Parameters							
Category NB1 signal power (P _{wanted}) / dBm	REFSENS + 6 dB						
Interferer	E-UTRA						
Interferer signal power (P _{Interferer}) / dBm	- 56 dBm						
Interferer bandwidth	5 MHz						
Interferer offset from category NB1 channel edge	+7,5 MHz + 0,005 MHz and -7,5 MHz - 0,005 MHz						
IBB2 test Parameters							
Category NB1 signal power (P _{wanted}) / dBm	REFSENS + 6 dB						
Interferer	E-UTRA						
Interferer signal power (P _{Interferer}) / dBm	- 44 dBm						
Interferer bandwidth	5 MHz						
Interferer offset range from category NB1 channel edge	From +12,5 MHz to F _{DL_high} + 15 MHz and From -12,5 MHz to F _{DL_low} - 15 MHz						

Table 4.2.7.3.2-2: Out-of-band blocking parameters for category NB1 UE

Parameter	Unito	Frequency						
Parameter	Units	Range 1	Range 2	Range 3				
Pwanted	dBm		REFSENS + 6 dB					
P _{interferer} (CW)	dBm	-44	-30	-15				
F. range	MHz	F _{DL_low} - 15 to F _{DL_low} - 60	F _{DL_low} - 60 to F _{DL_low} - 85	F _{DL_low} - 85 to 1 MHz				
interferer range	MHz	F _{DL_high} + 15 to F _{DL_high} + 60	F _{DL_high} + 60 to F _{DL_high} + 85	F _{DL_high} + 85 to 12 750 MHz				
NOTE 1: For op the int by F _{DI} F _{DL_hi} NOTE 2: For op power which 1 805	 OTE 1: For operating bands which downlink band frequency range is between 729 MHz < 1 GHz the power level of the interferer (P_{Interferer}) for Range 3 shall be modified to: -18 dBm for the frequency range which is bounded by F_{DL_low} - 150 MHz of the lowest band that UE supports in frequency range 729 MHz < 1 GHz and F_{DL_high} + 150 MHz of the highest band that UE supports in frequency range 729 MHz < 1 GHz. OTE 2: For operating bands which downlink band frequency range is between 1 805 MHz < f < 2 200 MHz the power level of the interferer (P_{Interferer}) for Range 3 shall be modified to: -20 dBm for the frequency range which is bounded by F_{DL_high} + 150 MHz of the interferer (P_{Interferer}) for Range 3 shall be modified to: -20 dBm for the frequency range which is bounded by F_{DL_low} - 200 MHz of the lowest band that UE supports in frequency range 							
range	range 1 805 MHz < f < 2 200 MHz.							

4.2.7.3.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.6.3 of the present document.

4.2.8 Receiver Spurious Response

4.2.8.1 Receiver Spurious Response for Single Carrier

4.2.8.1.1 Definition

Spurious response is a measure of the receiver's ability to receive a wanted signal on its assigned channel frequency without exceeding a given degradation due to the presence of an unwanted CW interfering signal at any other frequency at which a response is obtained i.e. for which the out-of-band blocking limit as specified in table 4.2.7.1.2-4 is not met.

The throughput shall be \geq 95 % of the maximum throughput of the reference measurement channels as specified in clauses A.2.2, A.2.3 and A.3.2 (with one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in clauses A.5.1.1 and A.5.2.1) in ETSI TS 136 521-1 [1] with parameters specified in tables 4.2.8.1.2-1 and 4.2.8.1.2-2.

By Parameter	Units	Channel bandwidth						
KX Falalletel		1,4 MHz	3 MHz	5 MHz	10 MHz	15 MHz	20 MHz	
Power in Transmission	dPm	REFSENS + channel bandwidth specific value below						
Bandwidth Configuration	ubm	6	6	6	6	7	9	
NOTE 1: The transmitter sh	: The transmitter shall be set to 4 dB below P _{CMAX 1} at the minimum uplink configuration specified in							
ETSI TS 136 101	ETSI TS 136 101 [3] (table 7.3.1-2 with P _{CMAX L} as defined in clause 6.2.5).							
NOTE 2: Reference measu	Reference measurement channel is clause A.3.2 of ETSI TS 136 521-1 [1].							
NOTE 3: REFSENS as defi	REFSENS as defined in clause 7.3.3 in ETSI TS 136 521-1 [1].							

Table 4.2.0. I.2-2. Spurious Response	Table	4.2.8.1.2-2:	Spurious	Response
---------------------------------------	-------	--------------	-----------------	----------

Parameter	Units	Level
P _{Interferer} (CW)	dBm	-44
F _{Interferer}	MHz	Spurious response frequencies

4.2.8.1.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.7.1 of the present document.

4.2.8.2 Receiver Spurious Response for Carrier Aggregation in DL-only bands

4.2.8.2.1 Definition

Spurious response is a measure of the receiver's ability to receive a wanted signal on its assigned channel frequency without exceeding a given degradation due to the presence of an unwanted CW interfering signal at any other frequency at which a response is obtained i.e. for which the out-of-band blocking limit as specified in table 4.2.7.2.2-2 is not met.

4.2.8.2.2 Limits

The throughput shall be \geq 95 % of the maximum throughput on SCC of the reference measurement channels as specified in clauses A.2.2, A.2.3 and A.3.2 (with one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in clauses A.5.1.1 and A.5.2.1) in ETSI TS 136 521-1 [1] with parameters specified in tables 4.2.8.1.2-1 and 4.2.8.1.2-2.

4.2.8.2.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.7.2 of the present document.

4.2.8.3 Receiver Spurious Response for category NB1

4.2.8.3.1 Definition

Spurious response is a measure of the receiver's ability to receive a wanted signal on its assigned channel frequency without exceeding a given degradation due to the presence of an unwanted CW interfering signal at any other frequency at which a response is obtained i.e. for which the out-of-band blocking limit as specified in clause 4.2.7.3.2 is not met.

4.2.8.3.2 Limits

The throughput measurement derived in test procedure shall be \geq 95 % of the maximum throughput of the reference measurement channels as specified in clauses A.2.2, A.2.3 and A.3.2 (with one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in clauses A.5.1.1 and A.5.2.1) in ETSI TS 136 521-1 [1] with parameters specified in table 4.2.8.3.2-1.

Parameter	Unit	Level				
P _{signal}	dBm	REFSENS + 6				
P _{Interferer} (CW)	dBm	-44				
F _{Interferer}	MHz	Spurious response frequencies				
Number of spurious response		24 (in OOB range 1, 2, 3)				
frequencies		24 (III OOB lange 1, 2, 3)				
NOTE 1: Reference measureme	OTE 1: Reference measurement channel is specified in clause A.3.2 in ETSI TS 136 521-1 [1					
NOTE 2: The REFSENS power I	The REFSENS power level is specified in clause 7.3F.1.3 in ETSI TS 136 521-1 [1].					
OTE 3: OOB range 1, 2, 3 refers to table 4.2.7.3.2-2.						

Table 4.2.8.3.2-1: Spurious response parameters for category NB1

4.2.8.3.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.7.3 of the present document.

4.2.9 Receiver Intermodulation Characteristic

4.2.9.1 Receiver Intermodulation Characteristics for Single Carrier

4.2.9.1.1 Definition

Intermodulation response rejection is a measure of the capability of the receiver to receive- a wanted signal on its assigned channel frequency in the presence of two or more interfering signals which have a specific frequency relationship to the wanted signal.

4.2.9.1.2 Limits

The throughput shall be \geq 95 % of the maximum throughput of the reference measurement channels as specified in clauses A.2.2, A.2.3 and A.3.2 (with one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in clauses A.5.1.1 and A.5.2.1) in ETSI TS 136 521-1 [1] with parameters specified in table 4.2.9.1.2-1 for the specified wanted signal mean power in the presence of two interfering signals.

Dy Deveryor		Channel bandwidth							
RX Paramete	onits	1,4 MHz	3 1	/Hz	5 MHz	10 MHz	15 MHz	20 MHz	
Power in		REFSENS + channel bandwidth specific value below							
Transmission Bandwidth Configuration	dBm	12		8	6	6 6 7 9			
P _{Interferer 1} (CW)) dBm		-46						
P _{Interferer 2} (Modulated)	dBm	-46							
BW _{Interferer 2}		1,4		3			5		
F _{Interferer 1} (Offset) MHz		-BW/2 - 2,1 / +BW/2 + 2.1	-BW/	2 - 4,5 / 2 + 4.5	-BW/2 - 7,5 / +BW/2 + 7.5				
F _{Interferer 2} (Offs	et) MHz	- ,		1-	$2 \times F_{\text{Interfe}}$	erer 1	,-		
NOTE 1: The transmitter shall be set to 4 dB below PCMAX 1 at the minimum uplink configuration sp						n specified			
in ET	ISI TS 136 101	[3] (table 7.3.1-	2 with F		as defined i	in clause 6.2	2.5).		
NOTE 2: Refe NOTE 3: The i claus TS 1	 IOTE 2: Reference measurement channel is clause A.3.2 of ETSI TS 136 521-1 [1]. IOTE 3: The modulated interferer shall consist of the Reference measurement channel specified in clause A.3.2 of ETSI TS 136 521-1 [1] with set-up according to clause C.3.1 of ETSI TS 136 521-1 [1]. 								
TS 1	TS 136 521-1 [1] for channel bandwidth = 5 MHz.								
NOTE 4. REF NOTE 5: For I table Banc	 4: REFSENS as defined in clause 7.3.3 in ETSLIS 136 521-1 [1]. 5: For DL category M1 UE, the reference sensitivity for category M1 in ETSLTS 136 521-1 [1], tables 7.3EA-1 and 7.3EA-2 should be used as REFSENS for the power in Transmission Bandwidth Configuration. 								
NOTE 6: For D refer	5: For DL category M1 UE, the parameters for the applicable channel bandwidth apply, and BW refers to the corresponding channel bandwidth.							nd BW	

Table 4.2.9.1.2-1: Test parameters for Wide band intermodulation

4.2.9.1.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.8.1 of the present document.

4.2.9.2 Receiver Intermodulation Characteristics for Carrier Aggregation in DL-only bands

4.2.9.2.1 Definition

Intermodulation response rejection is a measure of the capability of the receiver to receive a wanted signal on its assigned channel frequency in the presence of two or more interfering signals which have a specific frequency relationship to the wanted signal.

4.2.9.2.2 Limits

The throughput shall be \geq 95 % of the maximum throughput on SCC of the reference measurement channels as specified in clauses A.2.2, A.2.3 and A.3.2 (with one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in clauses A.5.1.1 and A.5.2.1) in ETSI TS 136 521-1 [1] with parameters specified in table 4.2.9.1.2-1 or table 4.2.9.2.2-1 for band combinations including operating band 46 without uplink operation for the specified wanted signal mean power in the presence of two interfering signals.

E-UTR	A Rx Parameter	Units	Channel bandwidth						
band			1,4 MHz	3 M	Hz	5 MHz	10 MHz	15 MHz	20 MHz
	Power in		REFSENS + channel bandwidth specific value below						ow
	Transmission Bandwidth Configuration	dBm	-		-	-	-	-	9
	P _{Interferer 1} (CW)	dBm	-46						
46	P _{Interferer 2} (Modulated)	dBm	-46						
	BW Interferer 2		-	-		-	-	-	20
	F _{Interferer 1} (Offset)	MHz	-	-		-	-	-	-BW/2 - 30 / +BW/2 + 30
	F _{Interferer 2} (Offset)	MHz	2*F _{Interferer 1}						
NOTE 1:	The transmitter shall be	e set to 4 of	dB below P _{CMA}	_{XL} at th	ne mini	mum uplin	k configura	ition specifi	ed in ETSI
	TS 136 101 [3] (table 7	.3.1-2 with	n P _{CMAX I} as d	efined i	n claus	e 6.2.5).			
NOTE 2:	 Reference measurement channel is specified in clause A.3.2 of ETSI TS 136 521-1 [1] with one sided dynamic OCNG Pattern OP.1 FDD/TDD/FS3 as described in clauses A.5.1.1, A.5.2.1 and A.5.4.1 of ETSI TS 136 521-1 [1]. 								
NOTE 3:	The modulated interferer consists of the Reference measurement channel specified in clause A.3.2 of ETSI TS 136 521-1 [1] with one sided dynamic OCNG Pattern OP.1 FDD/TDD as described in clauses A.5.1.1 and A.5.2.1 of ETSI TS 136 521-1 [1].								

Table 4.2.9.2.2-1: Test Parameters for Wide band intermodulation for carrier aggregation with band 46

4.2.9.2.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.8.2 of the present document.

4.2.9.3 Receiver Intermodulation Characteristics for category NB1

4.2.9.3.1 Definition

Intermodulation response rejection is a measure of the capability of the receiver to receive- a wanted signal on its assigned channel frequency in the presence of two or more interfering signals which have a specific frequency relationship to the wanted signal.

4.2.9.3.2 Limits

The throughput shall be \geq 95 % of the maximum throughput of the reference measurement channels as specified in clauses A.2.2, A.2.3 and A.3.2 (with one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in clauses A.5.1.1 and A.5.2.1) in ETSI TS 136 521-1 [1] with parameters specified in table 4.2.9.3.2-1 for the specified wanted signal mean power in the presence of two interfering signals.

de band intermodulation for category NB1
de band intermodulation for category NB

Parameters for wideband intermodulation						
Category NB1 signal power	REFSENS + 12 dB					
CW interferer signal power	-46 dBm					
1,4 MHz E-UTRA interferer signal power	-46 dBm					
CW interferer offset	±2,2 MHz					
1,4 MHz E-UTRA interferer offset	±4,4 MHz					

4.2.9.3.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.8.3 of the present document.

4.2.10 Receiver Spurious Emissions

4.2.10.1 Receiver Spurious Emissions for Single Carrier

4.2.10.1.1 Definition

The spurious emissions power is the power of emissions generated or amplified in a receiver that appear at the UE antenna connector.

4.2.10.1.2 Limits

The measured spurious emissions derived in clause 5.3.9 shall not exceed the maximum level specified in table 4.2.10.1.2-1.

Table 4.2.10.1.2-1: General receiver spurious emission requirements

Frequency Band	Measurement bandwidth	Maximum level	Note				
30 MHz ≤ f < 1 GHz	100 kHz	-57 dBm					
1 GHz ≤ f ≤ 12,75 GHz	1 MHz	-47 dBm					
12,75 GHz \leq f \leq 5 th harmonic of	1 MHz	-47 dBm	Note 1				
the upper frequency edge of the							
DL operating band in GHz							
NOTE 1: Shall apply only for Band 22, 42 and Band 43.							
NOTE 2: Unused PDCCH resources are padded with resource element groups with power level given by							
PDCCH_RA/RB as defined in ETSI TS 136 101 [3], clause C.3.1.							

4.2.10.1.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.9 of the present document.

4.2.10.2 Receiver Spurious Emissions in DL-only bands

4.2.10.2.1 Definition

The spurious emissions power is the power of emissions generated or amplified in a receiver that appear at the UE antenna connector.

4.2.10.2.2 Limits

The measured spurious emissions in SCC derived in clause 5.3.9 shall not exceed the maximum level specified in table 4.2.10.2.2-1.

Frequency Band	Measurement bandwidth	Maximum level	Note				
30 MHz ≤ f < 1 GHz	100 kHz	-57 dBm					
1 GHz ≤ f ≤ 12,75 GHz	1 MHz	-47 dBm					
12,75 GHz ≤ f ≤ 26 GHz	1 MHz	-47 dBm	3				
NOTE 1: Unused PDCCH resources are padded with resource element groups with power level given by PDCCH_RA/RB as defined in ETSI TS 136 101 [3], clause C.3.1.							
NOTE 2: The requirements ap	E 2: The requirements apply when the UE is configured for carrier aggregation but is not transmitting.						
NOTE 3: Applies only to Band	46.						

4.2.10.2.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.9 of the present document.

4.2.11 Transmitter Adjacent Channel Leakage Power Ratio

4.2.11.1 Transmitter adjacent channel leakage power ratio for Single Carrier

4.2.11.1.1 Definition

Adjacent Channel Leakage power Ratio (ACLR) is the ratio of the filtered mean power centred on the assigned channel frequency to the filtered mean power centred on an adjacent channel frequency.

E-UTRA Adjacent Channel Leakage power Ratio (E-UTRA_{ACLR}) is the ratio of the filtered mean power centred on the assigned channel frequency to the filtered mean power centred on an adjacent channel frequency at nominal channel spacing. The assigned E-UTRA channel power and adjacent E-UTRA channel power are measured with rectangular filters with measurement bandwidths specified in table 6.6.2.3.3.1-1 in ETSI TS 136 521-1 [1].

UTRA Adjacent Channel Leakage power Ratio is specified for both the first UTRA adjacent channel (UTRA_{ACLR1}) and the 2nd UTRA adjacent channel (UTRA_{ACLR2}). The UTRA channel power is measured with a RRC bandwidth filter with roll-off factor $\alpha = 0.22$. The assigned E-UTRA channel power is measured with a rectangular filter with measurement bandwidth specified in table 4.2.11.1.2-2.

4.2.11.1.2 Limits

If the measured adjacent channel power is greater than -50 dBm then the measured E-UTRA_{ACLR} shall be higher than the limits in table 4.2.11.1.2-1.

	Channel bandwidth/E-UTRA _{ACLR1} /measurement bandwidth							
	1,4 MHz	1,4 MHz 3,0 MHz 5 MHz 10 MHz 15 MHz 20 MHz						
E-UTRA _{ACLR1}	29,2 dB	29,2 dB	29,2 dB	29,2 dB	29,2 dB	29,2 dB		
E-UTRA channel	1,08 MHz	2,7 MHz	4,5 MHz	9,0 MHz	13,5 MHz	18 MHz		
Measurement								
bandwidth								
UE channel	+1,4 MHz or	+3 MHz or	+5 MHz or	+10 MHz or	+15 MHz or	+20 MHz or		
	-1,4 MHz	-3 MHz	-5 MHz	-10 MHz	-15 MHz	-20 MHz		

Table 4.2.11.1.2-1: E-UTRA UE ACLR

If the measured UTRA channel power is greater than -50 dBm then the measured UTRA_{ACLR1}, UTRA_{ACLR2} shall be higher than the limits in table 4.2.11.1.2-2.

		Channel b	andwidth/UTRA _{ACL}	R1/2/measuremer	nt bandwidth				
	1,4 MHz	3,0 MHz	5 MHz	10 MHz	15 MHz	20 MHz			
UTRA _{ACLR1}	32,2 dB	32,2 dB	32,2 dB	32,2 dB	32,2 dB	32,2 dB			
Adjacent	0,7 +	1,5 + BW _{UTRA} /2	2,5 + BW _{UTRA} /2	5 + BW _{UTRA} /2	7,5 + BW _{UTRA} /2	10 + BW _{UTRA} /2			
channel centre	BW _{UTRA} /2	/	/	/	/	/			
frequency	/	-1,5 - BW _{UTRA} /2	-2,5 - BW _{UTRA} /2	-5 - BW _{UTRA} /2	-7,5 -	-10 - BW _{UTRA} /2			
offset (In MHZ)	-0,7 -	_	-	-	BW _{UTRA} /2				
	BW _{UTRA} /2								
UTRA _{ACLR2}	-	-	35,2 dB	35,2 dB	35,2 dB	35,2 dB			
Adjacent	-	-	2,5 + 3 ×	5 + 3 ×	7,5 + 3 ×	10 + 3 × BW _{UTRA} /2			
channel centre			BW _{UTRA} /2	BW _{UTRA} /2	BW _{UTRA} /2	/			
frequency			/	/	/	-10 - 3 × BW _{UTRA} /2			
offset (In MHZ)			-2,5 - 3 ×	-5 - 3 ×	-7,5 - 3 ×				
			BW _{UTRA} /2	BW _{UTRA} /2	BW _{UTRA} /2				
E-UTRA									
channel	1,08 MHz	2,7 MHz	4,5 MHz	9,0 MHz	13,5 MHz	18 MHz			
Measurement									
UTRA 5 MHz									
channel									
Measurement	3,84 MHz	3,84 MHz	3,84 MHz	3,84 MHz	3,84 MHz	3,84 MHz			
bandwidth									
(see note 1)									
UTRA 1,6 MHz									
channel	4 00 MUL	4 00 MU-	4 00 MU-	4.00 MUL	4 00 MUL	4.00 MUL			
measurement	1,28 MHZ	1,28 MHZ	1,28 MHZ	1,28 MHZ	1,28 MHZ	1,28 MHZ			
(see note 2)									
NOTE 1: Shall a	pply for F-UTRA	L A FDD co-existence	with UTRA FDD in	paired spectrum					
NOTE 2: Shall a	pply for E-UTRA	A TDD co-existence	e with UTRA TDD in	unpaired spectrur	n.				
NOTE 3: BWIITE	NOTE 3: BW_{IITRA} for UTRA FDD shall be 5 MHz and for UTRA TDD shall be 1,6 MHz.								

Table 4.2.11.1.2-2: UTRA UE ACLR

4.2.11.1.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.10 of the present document.

4.2.11.2 Transmitter adjacent channel leakage power ratio for Carrier Aggregation (DL CA and UL CA)

4.2.11.2.1 Definition

For intra-band contiguous carrier aggregation, the UTRA Adjacent Channel Leakage power Ratio (UTRA_{ACLR}) is the ratio of the filtered mean power centred on the assigned carrier aggregated channel bandwidth to the filtered mean power centred on an adjacent(s) UTRA channel frequency.

UTRA Adjacent Channel Leakage power Ratio is specified for both the first UTRA adjacent channel (UTRA_{ACLR1}) and the 2nd UTRA adjacent channel (UTRA_{ACLR2}). The UTRA channel power is measured with a RRC bandwidth filter with roll-off factor $\alpha = 0,22$. The assigned aggregated channel bandwidth power is measured with a rectangular filter with measurement bandwidth specified in table 4.2.11.2.2-1.

For intra-band non-contiguous carrier aggregation when all sub-blocks consist of one component carrier, the UTRA Adjacent Channel Leakage power Ratio (UTRA_{ACLR}) is the ratio of the sum of the filtered mean powers centred on the assigned sub-block frequencies to the filtered mean power centred on an adjacent(s) UTRA channel frequency. UTRA_{ACLR1/2} limits are applicable for all sub-blocks and are specified in table 4.2.11.2.2-2. UTRA_{ACLR1} is required to be met in the sub-block gap when the gap bandwidth Wgap is 5 MHz \leq Wgap < 15 MHz. Both UTRA_{ACLR1} and UTRA_{ACLR2} are required to be met in the sub-block gap when the gap bandwidth Wgap is 15 MHz \leq Wgap.

For inter-band carrier aggregation with one component carrier per operating band and the uplink active in two E-UTRA bands, the UTRA Adjacent Channel Leakage power Ratio (UTRA_{ACLR}) is the ratio of the filtered mean power centred on the assigned channel bandwidth on the component carrier to the filtered mean power centred on an adjacent channel frequency. The UTRA Adjacent Channel Leakage power Ratio is defined per carrier and the requirement is specified in clause 4.2.11.1.

66

For combinations of intra-band and inter-band carrier aggregation with three uplink component carriers (up to two contiguously aggregated carriers per band), the UTRA Adjacent Channel Leakage power Ratio (UTRA_{ACLR}) is defined as follows. For the E-UTRA band supporting one component carrier, the UTRA Adjacent Channel Leakage power Ratio (UTRA_{ACLR}) is the ratio of the filtered mean power centred on the assigned channel bandwidth of the component carrier to the filtered mean power centred on an adjacent(s) UTRA channel frequency and the requirements specified in clause 4.2.11.1 apply. For the E-UTRA band supporting two contiguous component carriers the UTRA Adjacent Channel Leakage power Ratio (UTRA_{ACLR}) is the ratio of the filtered mean power centred on the aggregated channel bandwidth to the filtered mean power centred on an adjacent(s) UTRA channel frequency and the requirements specified in clause 4.2.11.2 apply.

For intra-band contiguous carrier aggregation, the carrier aggregation E-UTRA Adjacent Channel Leakage power Ratio (CA E-UTRA_{ACLR}) is the ratio of the filtered mean power centred on the aggregated channel bandwidth to the filtered mean power centred on an adjacent aggregated channel bandwidth at nominal channel spacing. The assigned aggregated channel bandwidth power and adjacent aggregated channel bandwidth power are measured with rectangular filters with measurement bandwidth specified in table 4.2.11.2.3-1.

For intra-band non-contiguous carrier aggregation when all sub-blocks consist of one component carrier the E-UTRA Adjacent Channel Leakage power Ratio (E-UTRA_{ACLR}) is the ratio of the sum of the filtered mean powers centred on the assigned sub-block frequencies to the filtered mean power centred on an adjacent channel frequency at nominal channel spacing. In case the sub-block gap bandwidth Wgap is smaller than the sub-block bandwidth then for that sub-block no E-UTRA_{ACLR} limit is set for the gap. In the case of the sub-block gab bandwidth Wgap is smaller than either of the sub-block bandwidths then no E- UTRA_{ACLR} limit is set for the gap. The assigned E-UTRA sub-block power and adjacent E-UTRA channel power are measured with rectangular filters with measurement bandwidths specified in table 4.2.11.2.3-2. If the measured adjacent channel power is greater than -50 dBm then the E-UTRA_{ACLR} shall be higher than the value specified in table 4.2.11.2.3-2.

For inter-band carrier aggregation with one component carrier per operating band and the uplink active in two E-UTRA bands, E-UTRA Adjacent Channel Leakage power Ratio (E-UTRA_{ACLR}) is the ratio of the filtered mean power centred on the assigned channel bandwidth on a component carrier to the filtered mean power centred on an adjacent channel frequency. The E-UTRA Adjacent Channel Leakage power Ratio is defined per carrier and the limits are specified in clause 4.2.11.1.2.

For combinations of intra-band and inter-band carrier aggregation with three uplink component carriers (up to two contiguously aggregated carriers per band), the E-UTRA Adjacent Channel Leakage power Ratio (E-UTRA_{ACLR}) is defined as follows. For the E-UTRA band supporting one component carrier, the E-UTRA Adjacent Channel Leakage power Ratio (UTRA_{ACLR}) is the ratio of the filtered mean power centred on the assigned channel bandwidth of the component carrier to the filtered mean power centred on an adjacent channel frequency and the limit in clause 4.2.11.1.2 apply. For the E-UTRA band supporting two contiguous component carriers the E-UTRA Adjacent Channel Leakage power Ratio (E-UTRA_{ACLR}) is the ratio of the filtered mean power centred on the aggregated channel bandwidth to the filtered mean power centred on an adjacent(s) aggregated channel bandwidth at nominal channel spacing and the limits of CA E-UTRA_{ACLR} specified in clause 4.2.11.2.3 apply.

4.2.11.2.2 Limits for CA UTRA

If the measured adjacent UTRA channel power is greater than -50 dBm then the measured $UTRA_{ACLR1}$ and $UTRA_{ACLR2}$, shall be higher than the limits in table 4.2.11.2.2-1 for intra-band contiguous carrier aggregation.

	CA bandwidth class/UTRA _{ACLR1/2} /measurement bandwidth				
	CA bandwidth class C				
UTRA _{ACLR1}	32,2 dB				
Adjacent channel centre frequency offset (in MHz)	+ BW _{Channel_CA} / 2 + BW _{UTRA} /2 / - BW _{Channel_CA} / 2 - BW _{UTRA} /2				
UTRA _{ACLR2}	35,2 dB				
Adjacent channel centre frequency offset (in MHz)	+ BW _{Channel_CA} / 2 + 3 × BW _{UTRA} /2 / - BW _{Channel_CA} / 2 - 3 × BW _{UTRA} /2				
CA E-UTRA channel Measurement bandwidth	$BW_{Channel_{CA}} - 2 \times BW_{GB}$				
UTRA 5 MHz channel Measurement bandwidth (note 1)	3,84 MHz				
UTRA 1,6 MHz channel measurement bandwidth (note 2)	1,28 MHz				
NOTE 1: Shall apply for E-UTRA FDD co-existence with UTRA FDD in paired spectrum. NOTE 2: Shall apply for E-UTRA TDD co-existence with UTRA TDD in unpaired spectrum.					

Table 4.2.11.2.2-1: UTRA UE ACLR for CA

Table 4.2.11.2.2-2: Re	quirements for in	traband non-conti	guous CA	UTRAACLR1/2
------------------------	-------------------	-------------------	----------	-------------

	Channel bandwidth / UTRAACLR1/2 / measurement bandwidth							
	1,4	3,0	5	10	15	20		
	MHz	MHz	MHz	MHz	MHz	MHz		
UTRA _{ACLR1}	32,2 dB	32,2 dB	32,2 dB	32,2 dB	32,2 dB	32,2 dB		
Adjacent channel	0,7 +	1,5 +	2,5 +	5 + BW _{UTRA} /	7,5 +	10 + BW _{UTRA}		
centre frequency	BW _{UTRA} /2	BWutra/2	BW _{UTRA} /2	2	BW _{UTRA} /2	/ 2		
offset (in MHz)	/	/	/	/	/	/		
	-0,7 -	-1,5 -	-2,5 -	-5 - BW _{UTRA} /	-7,5 -BWutra	-10 -BWUTRA		
	BW _{UTRA} /2	BW _{UTRA} /2	BW _{UTRA} /2	2	/ 2	/ 2		
UTRA _{ACLR2}	-	-	35,2 dB	35,2 dB	35,2 dB	35,2 dB		
Adjacent channel	-	-	2,5 + 3 ×	5 + 3 ×	7,5 + 3 ×	10 + 3 ×		
centre frequency			BW _{UTRA} /2	BW _{UTRA} / 2	BW _{UTRA} /2	BW _{UTRA} /2		
offset (in MHz)			/	/	/	/		
			-2,5 - 3 ×	-5 -3 ×	-7,5 -3 ×	-10 -3 ×		
			BW _{UTRA} /2	BW _{UTRA} / 2	BW _{UTRA} /2	BW _{UTRA} /2		
E-UTRA channel								
Measurement	1,08 MHz	2,7 MHz	4,5 MHz	9,0 MHz	13,5 MHz	18 MHz		
bandwidth								
UTRA 5MHz								
channel	3.84 MHz	3 84 MHz	3.84 MHz	3.84 MHz	3 84 MHz	3.84 MHz		
Measurement	5,04 101 12	5,04 10112	5,04 10112	5,04 101 12	5,04 10112	5,04 101 12		
bandwidth ¹								
UTRA 1,6MHz								
channel	1 28 MHz	1 28 MHz	1 28 MHz	1 28 MHz	1 28 MHz	1 28 MHz		
measurement	1,20 10112	1,20 10112	1,20 1011 12	1,20 1011 12	1,20 1011 12	1,20 1011 12		
bandwidth ²								
NOTE 1: Applicable	e for E-UTRA F	DD co-existend	e with UTRA F	DD in paired s	pectrum.			
NOTE 2: Applicable	e for E-UTRA T	DD co-existend	e with UTRA T	DD in unpaired	spectrum.			
NOTE 3: BWUTRA for UTRA FDD is 5 MHz and for UTRA TDD is 1,6 MHz.								

4.2.11.2.3 Limits for CA EUTRA

If the measured adjacent E-UTRA channel power is greater than -50 dBm then the measured CA E-UTRA_{ACLR}, shall be higher than the limits in table 4.2.11.2.3-1.

67

	CA bandwidth class/CA E-UTRA _{ACLR} /Measurement bandwidth
	CA bandwidth class B and C
CA E-UTRA _{ACLR}	29,2 dB
CA E-UTRA channel Measurement bandwidth	$BW_{Channel_{CA}} - 2 \times BW_{GB}$
Adjacent channel centre frequency offset (in MHz)	+ BW _{Channel_CA} / - BW _{Channel_CA}

Table 4.2.11.2.3-1: CA E-UTRA ACLR

Table 4.2.11.2.3-2: Intraband non-contiguous CA E-UTRA UE ACLR

Channel bandwidth / E-UTRA _{ACLR1} / measurement bandwidth									
	1,4 3,0 5 10 15 20								
	MHZ	MHZ	MHZ	MHZ	MHZ	MHZ			
E-UTRAACLR1	29,2 dB	29,2 dB	29,2 dB	29,2 dB	29,2 dB	29,2 dB			
E-UTRA channel	1,08 MHz	2,7 MHz	4,5 MHz	9,0 MHz	13,5 MHz	18 MHz			
Measurement									
bandwidth									
UE channel	+1,4 MHz or	+3 MHz or	+5 MHz or	+10 MHz or	+15 MHz or	+20 MHz or			
	-1,4 MHz	-3 MHz	-5 MHz	-10 MHz	-15 MHz	-20 MHz			

4.2.11.2.4 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.10 of the present document.

4.2.11.3 Transmitter adjacent channel leakage power ratio for UL-MIMO

4.2.11.3.1 Definition

E-UTRA ACLR (E-UTRA_{ACLR}) is the ratio of the filtered mean power centred on the assigned channel frequency to the filtered mean power centred on an adjacent channel frequency at nominal channel spacing. The assigned E-UTRA channel power and adjacent E-UTRA channel power are measured with rectangular filters with measurement bandwidth specified in table 4.2.11.3.2-1.

For UEs with two transmit antenna connectors in closed-loop spatial multiplexing scheme, the requirements in table 4.2.11.3.2-1 apply to each transmit antenna connector with the UL-MIMO configurations specified in table 4.2.2.3.1-1.

UTRA ACLR (UTRA_{ACLR}) is the ratio of the filtered mean power centred on the assigned E-UTRA channel frequency to the filtered mean power centred on an adjacent UTRA channel frequency.

UTRA ACLR is specified for both the first UTRA adjacent channel (UTRA_{ACLR1}) and the 2^{nd} UTRA adjacent channel (UTRA_{ACLR2}). The UTRA channel power is measured with a RRC bandwidth filter with roll-off factor

 α = 0,22. The assigned E-UTRA channel power is measured with a rectangular filter with measurement bandwidth specified in table 4.2.11.3.2-2.

For UEs with two transmit antenna connectors in closed-loop spatial multiplexing scheme, the requirements in table 4.2.11.3.2-2 apply to each transmit antenna connector with the UL-MIMO configurations specified in table 4.2.2.3.1-1.

4.2.11.3.2 Limits

If the measured E-UTRA adjacent channel power is greater than -50 dBm then the E-UTRA_{ACLR} shall be higher than the value specified in table 4.2.11.3.2-1.

	Channel bandwidth/E-UTRA _{ACLR1} /measurement bandwidth							
	1,4 MHz	3,0 MHz	5 MHz	10 MHz	15 MHz	20 MHz		
E-UTRA _{ACLR1}	29,2 dB	29,2 dB	29,2 dB	29,2 dB	29,2 dB	29,2 dB		
E-UTRA channel Measurement bandwidth	1,08 MHz	2,7 MHz	4,5 MHz	9,0 MHz	13,5 MHz	18 MHz		
UE channel	+1,4 MHz or -1,4 MHz	+3 MHz or -3 MHz	+5 MHz or -5 MHz	+10 MHz or -10 MHz	+15 MHz or -15 MHz	+20 MHz or -20 MHz		

If the measured UTRA channel power is greater than -50 dBm then the $UTRA_{ACLR1}$ and $UTRA_{ACLR2}$ shall be higher than the value specified in table 4.2.11.3.2-2.

	Channel bandwidth/E-UTRA _{ACLR1/2} /measurement bandwidth								
	1,4 MHz	3,0 MHz	5 MHz	10 MHz	15 MHz	20 MHz			
E-UTRA _{ACLR1}	32,2 dB	32,2 dB	32,2 dB	32,2 dB	32,2 dB	32,2 dB			
Adjacent channel	0,7 +	1,5 + BW _{UTRA} /2	2,5 +	5 + BW _{UTRA} /2	7,5 +	10 + BW _{UTRA} /2			
centre frequency	BW _{UTRA} /2	/	BW _{UTRA} /2	/	BW _{UTRA} /2	/			
offset (in MHz)	/	-1,5 - BW _{LITRA} /2	/	-5 - BW _{LITRA} /2	/	-10 - BW _{LITRA} /2			
	-0,7 -	Onter	-2,5 -	Onix	-7,5 -	Onw			
	BW _{UTRA} /2		BW _{UTRA} /2		BW _{UTRA} /2				
UTRA _{ACLR2}	-	-	35,2 dB	35,2 dB	35,2 dB	35,2 dB			
Adjacent channel	-	-	2,5 + 3 ×	5+3×	7,5 + 3 ×	10 + 3 ×			
centre frequency			BW _{UTRA} /2	BW _{UTRA} /2	BW _{UTRA} /2	BW _{UTRA} /2			
offset (in MHz)			/	/	/	/			
			-2,5 - 3 ×	-5 - 3 ×	-7,5 - 3 ×	-10 - 3 ×			
			BW _{UTRA} /2	BW _{UTRA} /2	BW _{UTRA} /2	BW _{UTRA} /2			
E-UTRA channel	1,08 MHz	2,7 MHz	4,5 MHz	9,0 MHz	13,5 MHz	18 MHz			
Measurement									
bandwidth									
UTRA 5 MHz	3,84 MHz	3,84 MHz	3,84 MHz	3,84 MHz	3,84 MHz	3,84 MHz			
channel									
Measurement									
	1 29 MU-	1 29 MUz	1 29 MU-	1 29 MU-7	1 20 MU-	1 29 MU-7			
channel	1,20 1011 12	1,20 1011 12	1,20 1011 12	1,20 1011 12	1,20 1011 12	1,20 1011 12			
measurement									
bandwidth (note 2)									
NOTE 1: Shall apply f	or E-UTRA FDI	C co-existence with	UTRA FDD in pa	aired spectrum.		1			
NOTE 2: Shall apply for E-UTRA TDD co-existence with UTRA TDD in unpaired spectrum.									

Table 4.2.11.3.2-2: General requirements for UTRA_{ACLR1/2}

4.2.11.3.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.10 of the present document.

4.2.11.4 Transmitter adjacent channel leakage power ratio for Multi-Cluster PUSCH within a component carrier

4.2.11.4.1 Definition

For UE supporting multi cluster PUSCH within a component carrier for the operating band.

Adjacent Channel Leakage power Ratio (ACLR) is the ratio of the filtered mean power centred on the assigned channel frequency to the filtered mean power centred on an adjacent channel frequency.

4.2.11.4.2 Limits

If the measured adjacent channel power is greater than -50 dBm then the measured E-UTRA_{ACLR} shall be higher than the limits in table 4.2.11.1.2-1.

If the measured UTRA channel power is greater than -50 dBm then the measured UTRA_{ACLR1}, UTRA_{ACLR2} shall be higher than the limits in table 4.2.11.1.2-1.

4.2.11.4.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.10 of the present document.

4.2.11.5 Transmitter adjacent channel leakage power ratio for category NB1

4.2.11.5.1 Definition

Adjacent Channel Leakage power Ratio (ACLR) is the ratio of the filtered mean power centred on the assigned channel frequency to the filtered mean power centred on an adjacent channel frequency.

4.2.11.5.2 Limits

The assigned category NB1 channel power and adjacent channel power are measured with filters and measurement bandwidths specified in table 4.2.11.5.2-1.

If the measured adjacent channel power is greater than -50 dBm then the measured GSM_{ACLR} and $\text{UTRA}_{\text{ACLR}}$ shall be higher than the limits in table 4.2.11.5.2-1. GSM_{ACLR} requirement is intended for protection of GSM system. UTRA_{ACLR} requirement is intended for protection of UTRA and E-UTRA systems.

	GSM _{ACLR}		
ACLR	19,2 dB	36,2 dB	
Adjacent channel centre frequency offset	±200 kHz	±2,5 MHz	
from category NB1 Channel edge			
Adjacent channel measurement bandwidth	180 kHz	3,84 MHz	
Measurement filter	Rectangular	RRC-filter $\alpha = 0,22$	
Category NB1 channel measurement bandwidth	180 kHz	180 kHz	
Category NB1 channel Measurement filter	Rectangular	Rectangular	

Table 4.2.11.5.2-1: Category NB1 UE ACLR Test requirements

4.2.11.5.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.10 of the present document.

4.2.12 Receiver Reference Sensitivity Level

4.2.12.0 General

Unless otherwise stated, the receiver characteristics are specified at the antenna connector(s) of the UE. For UE(s) with an integral antenna only, a reference antenna(s) with a gain of 0 dBi should be assumed for each antenna port(s).

4.2.12.1 Receiver Reference Sensitivity Level for Single Carrier

4.2.12.1.1 Definition

Reference sensitivity measures the UE's ability to receive data with a given average throughput for a specified reference measurement channel, under conditions of low signal level, ideal propagation and no added noise.

A UE unable to meet the throughput requirement under these conditions will decrease the effective coverage area of an e-NodeB.

4.2.12.1.2 Limits

The throughput shall be \geq 95 % of the maximum throughput of the reference measurement channels as specified in ETSI TS 136 521-1 [1], clauses A.2.2, A.2.3 and A.3.2 (with one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in ETSI TS 136 521-1 [1], clauses A.5.1.1 and A.5.2.1) with parameters specified in table 4.2.12.1.2-1 and table 7.3.3-2 in ETSI TS 136 521-1 [1].

Channel bandwidth									
E-UTRA	1,4 MHz	3 MHz	5 MHz	10 MHz	15 MHz	20 MHz	Duplex Mode		
Band	(dBm)	(dBm)	(dBm)	(dBm)	(dBm)	(dBm)			
1	-	-	-99,3	-96,3	-94,5	-93,3	FDD		
3	-101,0	-98,0	-96,3	-93,3	-91,5	-90,3	FDD		
7	-	-	-97,3	-94,3	-92,5	-91,3	FDD		
8	-101,5	-98,5	-96,3	-93,3	-	-	FDD		
20	-	-	-96,3	-93,3	-90,5	-89,3	FDD		
22	-	-	-96,0	-93,0	-91,2	-90,0	FDD		
28	-	-99,5	-97,8	-94,8	-93,0	-90,3	FDD		
31	-98,3	-95,0	-92,8	-	-	-	FDD		
33	-	-	-99,3	-96,3	-94,5	-93,3	TDD		
34	-	-	-99,3	-96,3	-94,5	-	TDD		
38	-	-	-99,3	-96,3	-94,5	-93,3	TDD		
40	-	-	-99,3	-96,3	-94,5	-93,3	TDD		
41			-97.3	-94.3	-92.5	-91.3	TDD		
42	98,0		-98,0	-95,0	-93,2	-92,0	TDD		
43	-	-	-98,0	-95,0	-93,2	-92,0	TDD		
65	-103,5	-100,5	-98,8	-95,8	95,8 -94,0 -92,8		FDD		
68			-97,8	-94,8	-93,0		FDD		
72	-98,3	-95,0	-92,8				FDD		
87	-98,3	-95,0	-92,8				FDD		
88	-98,3	-95,0	-92,8				FDD		
NOTE 1:	NOTE 1: The transmitter shall be set to maximum output power level (ETSI TS 136 521-1 [1],								
	table 7.3.5-2).								
NOTE 2:	NOTE 2: The reference measurement channel is specified in ETSI TS 136 521-1 [1],								
clause A.3.2 with one sided dynamic OCNG Pattern OP.1 FDD/TDD as described in									
	ETSI TS 136 521-1 [1], clauses A.5.1.1 and A.5.2.1.								
NOTE 3: The signal power is specified per port.									

The reference receive sensitivity (REFSENS) requirement specified in table 4.2.12.1.2-1 shall be met for an uplink transmission bandwidth less than or equal to that specified in ETSI TS 136 521-1 [1], table 7.3.5-2.

4.2.12.1.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.11 of the present document.

4.2.12.2 Receiver Reference Sensitivity Level for Carrier Aggregation in DL-only bands

4.2.12.2.1 Definition

Reference sensitivity measures the UE's ability to receive data with a given average throughput for a specified reference measurement channel, under conditions of low signal level, ideal propagation and no added noise.

A UE unable to meet the throughput requirement under these conditions will decrease the effective coverage area of an e-NodeB.

4.2.12.2.2 Limits

The throughput shall be \geq 95 % of the maximum throughput of the reference measurement channels as specified in ETSI TS 136 521-1 [1], clauses A.2.2, A.2.3 and A.3.2 (with one sided dynamic OCNG Pattern OP.1 FDD/TDD/FS3 for the DL-signal as described in ETSI TS 136 521-1 [1], clauses A.5.1.1, A.5.2.1 and A.5.4.1) with parameters specified in table 4.2.12.2.2-1 and table 7.3A.3.5-2 of ETSI TS 136 521-1 [1].

Channel bandwidth										
EUTRA CA		E-UTRA	1,4 MHz	3 MHz	5 MHz	10 MHz	15 MHz	20 MHz	Duplex	
Configuration		Band	(dBm)	(dBm)	(dBm)	(dBm)	(dBm)	(dBm)	Mode	
CA_1A-46A		1	-	-	-99,3	-96,3	-94,5	-93,3	FDD	
		46	-	-	-	-	-	-88,5	TDD	
CA_3A-46A		3	-	-	-96,3	-93,3	-91,5	-90,3	FDD	
		46	-	-	-	-	-	-88,5	TDD	
CA_7A-46A		7	-	-	-97,3	-94,3	-92,5	-91,3	FDD	
		46	-	-	-	-	-	-88,5	TDD	
CA_20A-32A		20	-	-	-96,3	-93,3	-	-	FDD	
		32	-	-	-99,3	-96,3	-94,5	-93,3		
CA_42A-46A		42	-	-	-98	-95	-93,2	-92	TDD	
		46	-	-	-	-	-	-81,5		
NOTE 1: The transmitter shall be set to maximum output power level.										
NOTE 2: Reference measurement channel is in ETSI TS 136 521-1 [1], clause A.3.2 with one sided dynamic										
(OCNG Pattern OP.1 FDD/TDD/FS3 as described in ETSI TS 136 521-1 [1] in clauses A.5.1.1, A.5.2.1									
	and A.5.4.1.									
NOTE 3:	3: The signal power is specified per port.									
NOTE 4:	4: Void.									
NOTE 5:	5: The requirement for B46 does not apply when there is at least one individual RE within the B46									
	downlink transmission bandwidth which falls into the reference sensitivity exclusion region as specified									
i	in table 4	1.2.12.2.2-2.						0	•	
NOTE 6:	: Void.									

Table 4.2.12.2.2-1: Reference sensitivity QPSK PREFSENS for inter-band Carrier Aggregation in DL-only band

Table 4.2.12.2.2-2 specifies the Band 46 reference measurement exclusion region for different licensed component carrier s and channel bandwidth. The exclusion region is defined according to the licensed component carrier channel bandwidth. The UL configurations to be adopted for the test are specified in table 7.3.1-2 in ETSI TS 136 101 [3]. The exclusion region in table 4.2.12.2.2-2 is specified for the case of 10 MHz and 20 MHz channel bandwidth in Band 46.
Li	Licensed Component Carriers / E-UTRA Band / Harmonic order / Channel BW in UL						
Licensed Component		Harmonic order	5 MHz	10 MHz	15 MHz	20 MHz	
Ca	rriers			_		-	
	1	3	±15	±23	±35	±45	
3		3	±15	±23	±35	±45	
7 ¹		2	±15	±25	±38	±50	
NOTE 1: E	Even though UL harmonic does not fall directly into Band 46 the exclusion region still applies.						
NOTE 2: T	The centre of the exclusion region is obtained by multiplying the UL channel centre frequency by the harmonic order.						

Table 4.2.12.2.2-2: Band 46 Reference sensitivity measurement exclusion region in MHz

73

4.2.12.2.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.11 of the present document.

4.2.12.3 Receiver Reference Sensitivity Level for category NB1

4.2.12.3.1 Definition

To verify the UE's ability to receive data with a given average throughput for a specified reference measurement channel, under conditions of low signal level, ideal propagation and no added noise.

A UE unable to meet the throughput requirement under these conditions will decrease the effective coverage area of an e-NodeB.

4.2.12.3.2 Limits

The throughput shall be \geq 95 % of the maximum throughput of the reference measurement channels as specified in ETSI TS 136 521-1 [1], clause A.3.2 2 (with one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in ETSI TS 136 521-1 [1], clauses A.5.1.1 and A.5.2.1) with parameters specified in table 4.2.12.3.2-1.

Table 4.2.12.3.2-1: Reference sensitivity for category NB1

Operating band	REFSENS [dBm]
1, 3, 8, 20, 28, 65, 72, 87, 88	-107,5

4.2.12.3.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.11 of the present document.

4.2.12.4 Receiver Reference Sensitivity Level for UE category 0

4.2.12.4.1 Definition

Reference sensitivity measures the UE's ability to receive data with a given average throughput for a specified reference measurement channel, under conditions of low signal level, ideal propagation and no added noise.

A UE unable to meet the throughput requirement under these conditions will decrease the effective coverage area of an e-NodeB.

4.2.12.4.2 Limits

The throughput shall be \geq 95 % of the maximum throughput of the reference measurement channels as specified in ETSI TS 136 521-1 [1], clauses A.2.2, A.2.3 and A.3.2 (with one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in ETSI TS 136 521-1 [1], clauses A.5.1.1 and A.5.2.1) with parameters specified in table 4.2.12.4.2-1 and table 4.2.12.4.2-2.

Channel bandwidth								
E-UTRA Band		1,4 MHz (dBm)	3 MHz (dBm)	5 MHz (dBm)	10 MHz (dBm)	15 MHz (dBm)	20 MHz (dBm)	Duplex Mode
3		-98,5	-95,5	-93,8	-90,8	-89	-87,8	FDD
8 -99		-99	-96	-93,8	-90,8			FDD
20 -93,8		-90,8	-87,5	-86,3	FDD			
41			-94.8	-91.8	-90,0	-88.8	TDD	
NOTE 1: The transmitter shall be set to P _{UMAX} as defined in ETSI TS 136 521-1 [1], clause 6.2.5.								
NOTE 2: Reference measurement channel is specified in ETSI TS 136 521-1 [1], clause A.3.2 with one sided dynamic OCNG Pattern OP.1 FDD/TDD as described in ETSI TS 136 521-1 [1], clauses A.5.1.1 and A.5.2.1.								

Table 4.2.12.4.2-1: Reference sensitivity for FDD and TDD UE category 0 QPSK PREFSENS

74

Table 4.2.12.4.2-2: Reference sensitivity for HD-FDD U	E category 0 QPSK P _{REFSENS}
--	--

Channel bandwidth							
E-UTRA Band	1,4 MHz (dBm)	3 MHz (dBm)	5 MHz (dBm)	10 MHz (dBm)	15 MHz (dBm)	20 MHz (dBm)	Duplex Mode
3	-99,3	-96,3	-94,6	-91,6	-89,8	-88,6	HD-FDD
8	-99,8	-96,8	-94,6	-91,6			HD-FDD
20			-94,6	-91,6	-88,8	-87,6	HD-FDD
NOTE 1: The transmitter shall be set to P _{UMAX} as defined in ETSI TS 136 521-1 [1], clause 6.2.5.							
NOTE 2: Reference measurement channel is specified in ETSI TS 136 521-1 [1], clause A.3.2 with one sided dynamic OCNG Pattern OP.1 FDD/TDD as described in ETSI TS 136 521-1 [1], clauses A.5.1.1 and A.5.2.1.							

The reference receive sensitivity (REFSENS) requirement specified in table 4.2.12.4.2-1 and table 4.2.12.4.2-2 shall be met for an uplink transmission bandwidth less than or equal to that specified in ETSI TS 136 101 [3], table 7.3.1E-2.

4.2.12.4.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.11.3 of the present document.

4.2.12.5 Receiver Reference Sensitivity Level for UE category M1

4.2.12.5.1 Definition

Reference sensitivity measures the UE's ability to receive data with a given average throughput for a specified reference measurement channel, under conditions of low signal level, ideal propagation and no added noise.

A UE unable to meet the throughput requirement under these conditions will decrease the effective coverage area of an e-NodeB.

4.2.12.5.2 Limits

The throughput shall be \geq 95 % of the maximum throughput of the reference measurement channels as specified in ETSI TS 136 521-1 [1], clauses A.2.2, A.2.3 and A.3.2 with parameters specified in table 4.2.12.5.2-1 and table 4.2.12.5.2-2.

E-UTRA Band		REFSENS (dBm) Duplex Mode				
	1	-101,5	FDD			
:	3	-98,5	FDD			
7	7	-95,5	FDD			
8	3	-99,0	FDD			
2	0	-99,0	FDD			
2	8	-100,0	FDD			
3	1	-95,8	FDD			
4	1	-101,0	TDD			
7	2	-95,8	FDD			
8	7	-95,8	FDD			
8	8	-95,8	FDD			
NOTE 1:	The transr	nitter shall be set to P _{UMAX} as defined in ETSI T	S 136 521-1 [1], clause 6.2.5EA.			
NOTE 2: NOTE 3:	 Reference measurement channel is specified in ETSI TS 136 521-1 [1], clause A.3.2 with one sided dynamic OCNG Pattern OP.1 FDD/TDD as described in ETSI TS 136 521-1 [1], clauses A.5.1.1 and A.5.2.1. Void. 					
NOTE 4:						
NOTE 5: NOTE 6:	For cat M1 the same RF bandwidth applies for all applicable channel bandwidths as specified in ETSI TS 136 521-1 [1], table 5.4.2.1-1. The reference receive sensitivity shall be met for an uplink transmission bandwidth less than or equal to 6 RB except for band 31. For band 31; in the case of 3 MHz channel bandwidth 5 RB applies and the UL resource blocks shall be located at RB _{start} 9. In case of 5 MHz					
	channel ba	channel bandwidth 5 RB applies and the UL resource blocks shall be located at RB _{start} 10.				
NOTE 7:	The UL resource blocks shall be located as close as possible to the downlink operating band but confined within the transmission bandwidth configuration for the channel bandwidth.					
NOTE 8:	These REFSENS levels are applicable only to 6 RB. OCNG shall be filled to the entire channel bandwidth with the same PSD of REFSENS level in 6 RBs.					

Table 4.2.12.5.2-1: Reference sensitivity for FDD and TDD UE category M1 QPSK PREFSEN

75

E-UTRA Band		REFSENS (dBm)	Duplex Mode			
1		-102,3	HD-FDD			
3	3	-99,3	HD-FDD			
7	7	-100,3	HD-FDD			
8	3	-99,8	HD-FDD			
2	0	-99,8	HD-FDD			
2	8	-100,8	HD-FDD			
3	1	-96,6	HD-FDD			
7	2	-96,6	HD-FDD			
8	7	-96,6	HD-FDD			
8	8	-96,6	HD-FDD			
NOTE 1:	The transr	nitter shall be set to P _{UMAX} as defined in ETSI T	S 136 521-1 [1], clause 6.2.5EA.			
NOTE 2:	NOTE 2: Reference measurement channel is specified in ETSI TS 136 521-1 [1], clause A.3.2 with one sided dynamic OCNG Pattern OP.1 FDD/TDD as described in ETSI TS 136 521-1 [1], clauses A.5.1.1 and A.5.2.1.					
NOTE 3:	Void.					
NOTE 4:	Void.					
NOTE 5:	5: For cat M1 the same RF bandwidth applies for all applicable channel bandwidths as specified in ETSI TS 136 521-1 [1], table 5.4.2-1.					
NOTE 6:	These REFSENS levels are applicable only to 6 RB. OCNG shall be filled to the entire channel bandwidth with the same PSD of REFSENS level in 6 RBs.					

The reference receive sensitivity (REFSENS) requirement specified in table 4.2.12.5.2-1 and table 4.2.12.5.2-2 shall be met for an uplink transmission bandwidth less than or equal to that specified in ETSI TS 136 101 [3], table 7.3.1E-5.

4.2.12.5.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.11.4 of the present document.

4.2.13 Receiver Total Radiated Sensitivity (TRS)

4.2.13.0 Applicability

The present requirement applies to handheld phones/DUTs that are wider than or equal to 56 mm and narrower than or equal to 72 mm.

'Receiver Total Radiated Sensitivity (TRS)' requirements for devices not fitting this width criteria (e.g. devices narrower than 56 mm and wider than 72 mm) are not covered by the current version of the present document.

4.2.13.1 Definition

The Total Radiated Sensitivity is defined as:

$$TRS = \frac{4\pi}{\oint \left[\frac{1}{EIS_{\theta}(\Omega; f)} + \frac{1}{EIS_{\varphi}(\Omega; f)}\right]} d\Omega$$

Where the Effective Isotropic Sensitivity (EIS) is defined as the power available at the antenna output such as the sensitivity threshold is achieved for each polarization. Ω is the solid angle describing the direction, f is frequency. θ and φ are the orthogonal polarizations.

$$TRS \approx \frac{2NM}{\pi \sum_{n=0}^{N-1} \sum_{m=0}^{M-1} \left[\frac{1}{EIS_{\theta}(\theta_n, \varphi_m; f)} + \frac{1}{EIS_{\varphi}(\theta_n, \varphi_m; f)} \right] \sin(\theta_n)}$$

In these formulas, N and M are the number of sampling intervals for theta and phi. θ_n and φ_m are the measurement angles. The sampling intervals are discussed further in clause 4.4 of ETSI TS 137 544 [6].

The TRS can also be calculated from measurements in a Rayleigh fading 3 dimensional isotropic environment within average uniform elevation and azimuth distribution. The calculation of the TRS is in this case based on searching for the lowest power received by the UE/MS for a discrete number of field combinations in the chamber that gives a BER that is better than the specified target BER level. By calibrating the average power transfer function, an absolute value of the TRS can be obtained. The following expression can be used to find the TRS:

$$TRS \approx 2N \frac{\left(\sum_{n=1}^{N} \left(C_n \left(1-R_n\right) P_{thres,n}\right)\right)^{-1}}{\sum_{n=1}^{N} P_{ref,n}}$$

where $P_{ref,n}$ is the reference power transfer function for fixed measurement antenna n, R_n is the reflection coefficient for fixed measurement antenna n and C_n is the path loss in the cables connecting the measurement receiver to fixed measurement antenna n. These parameters are calculated from the calibration measurement and are further discussed in clause B.2 of ETSI TS 137 544 [6]. P_{meas} is calculated by using the following equation:

$$P_{thres,n} = \frac{\sum_{m=1}^{M} \frac{1}{\left|S_{21,n,m}^{thres}\right|^2}}{M}$$

where $S_{21,n,m}^{thres}$ is the m:th value of the transfer function for fixed measurement antenna n, which gives the BER threshold. *M* is the total number of values of the BER threshold power measured for each fixed measurement antenna.

4.2.13.2 Limits

The average measured Total Radiated Sensitivity (TRS) of low, mid and high channels for handheld UE shall be lower than the average TRS requirement specified in table 4.2.13.2-1. The averaging shall be done in linear scale for the TRS results of both right and left side of the phantom head. Average TRS requirement is shown in the column "Average" on the requirement tables.

$$TRS_{average} = 10\log\left[6 \left/ \left(\frac{1}{10^{P_{left_low}/10}} + \frac{1}{10^{P_{left_mid}/10}} + \frac{1}{10^{P_{left_high}/10}} + \frac{1}{10^{P_{right_low}/10}} + \frac{1}{10^{P_{right_mid}/10}} + \frac{1}{10^$$

Table 4.2.13.2-1: TRS minimum requirements for E-UTRA FDD and TDD bands in the speech position Beside Head Hand Left and Beside Head Hand Right (BHHL/BHHR) for the primary mechanical mode

Operating band	Unit	<refî<sub>or></refî<sub>
		Average
1	dBm/10 MHz	-86
3	dBm/10 MHz	-86
7	dBm/10 MHz	-85,7
8	dBm/10 MHz	-82,5
20	dBm/10 MHz	-82,5
28	dBm/10 MHz	-82,5
38	dBm/20 MHz	-82,5
40	dBm/20 MHz	-82,5
NOTE: Not applicable for carrier aggregation.		

NOTE: The TRS minimum requirements are applicable for devices wider than or equal to 56 mm and narrower than or equal to 72 mm as defined in ETSI TR 125 914 [i.13].

4.2.13.3 Conformance

The conformance tests for this requirement shall be as defined in clause 5.3.12 of the present document.

4.2.14 Total Radiated Power (TRP)

4.2.14.0 Applicability

The present requirement applies to handheld phones/DUTs that are wider than or equal to 56 mm and narrower than or equal to 72 mm.

'Total Radiated Power (TRP)' requirements for devices not fitting this width criteria (e.g. devices narrower than 56 mm and wider than 72 mm) are not covered by the current version of the present document.

4.2.14.1 Definition

The Total Radiated Power (TRP) is a measure of how much power the DUT actually radiates. The TRP is defined as the integral of the power transmitted in different directions over the entire radiation sphere:

$$TRP = \frac{1}{4\pi} \oint \left(EIRP_{\theta}(\Omega; f) + EIRP_{\phi}(\Omega; f) \right) d\Omega$$

Where Ω is the solid angle describing the direction, f is frequency. θ and φ are the orthogonal polarizations.

 $EIRP_{\theta}$ and $EIRP_{\phi}$ are the actually transmitted power-levels in corresponding polarizations.

Thus:

$$TRP \approx \frac{\pi}{2NM} \sum_{n=0}^{N-1} \sum_{m=0}^{M-1} \left[EIRP_{\theta}(\theta_n, \varphi_m; f) + EIRP_{\varphi}(\theta_n, \varphi_m; f) \right] \sin(\theta_n)$$

In these formulas N and M are the number of sampling intervals for theta and phi. θ_n and φ_n are the measurement angles. The sampling intervals are discussed further in clause 4.4 of ETSI TS 137 544 [6].

78

The TRP can also be calculated from Rayleigh faded samples of the total power transmitted from the UE. The measurement of transmitter performance in an isotropic Rayleigh fading environment is based on sampling the radiated power of the UE for a discrete number of field combinations in the chamber. The average value of these statistically distributed samples is proportional to the TRP and by calibrating the average power transfer function, an absolute value of the TRP can be obtained.

Thus:

$$TRP \approx \frac{\sum_{n=1}^{N} \left(\frac{P_n}{C_n (1 - R_n)} \right)}{\sum_{n=1}^{N} P_{ref,n}}$$

where $P_{ref,n}$ is the reference power transfer function for fixed measurement antenna n, R_n is the reflection coefficient for fixed measurement antenna n and C_n is the path loss in the cables connecting the measurement receiver to fixed measurement antenna n. These parameters are calculated from the calibration measurement and are further discussed in clause B.2 of ETSI TS 137 544 [6]. P_n is the average power measured by fixed measurement antenna n and can be calculated using the following expression:

$$P_{n} = \frac{\sum_{m=1}^{M} \left| S_{21,n,m} \right|^{2}}{M}$$

where $S_{21n,m}$ is sample number m of the complex transfer function measured with fixed measurement antenna n and M is the total number of samples measured for each fixed measurement antenna.

- NOTE 1: All averaging are performed using linear power values (e.g. measurements in Watts).
- NOTE 2: The requirements and this test apply to all types of UTRA for the FDD UE for Release 7 and later releases.

4.2.14.2 Limits

The average TRP of low, mid and high channels in beside head position shall be higher than minimum performance requirements for roaming bands shown in table 4.2.14.2-1. The averaging shall be done in linear scale for the TRP results of both right and left side of the phantom head.

$$TRP_{average} = 10\log\left[\frac{10^{P_{left_low}/10} + 10^{P_{left_mid}/10} + 10^{P_{left_high}/10} + 10^{P_{right_low}/10} + 10^{P_{right_mid}/10} + 10^{P_{right_high}/10}}{6}\right]$$

Table 4.2.14.2-1: TRP minimum performance requirement for E-UTRA FDD and TDD bands in the speech position Beside Head Hand Left and Beside Head Hand Right (BHHL/BHHR) for primary mechanical mode

79

Operating band	Unit	Power Class 3		
		Power (dBm)		
		Average		
1	dBm/10 MHz	10,9		
3	dBm/10 MHz	10,9		
7	dBm/10 MHz	10,9		
8	dBm/10 MHz	7,6		
20	dBm/10 MHz	7,6		
28	dBm/10 MHz	7,6		
38	dBm/20 MHz	10,9		
40	dBm/20 MHz	10,9		
NOTE: Not applicable for carrier aggregation.				

NOTE: The TRP minimum requirements are applicable for devices wider than or equal to 56 mm and narrower than or equal to 72 mm as defined in ETSI TR 125 914 [i.13].

5 Testing for compliance with technical requirements

5.1 Environmental conditions for testing

Tests defined in the present document shall be carried out at representative points within the boundary limits of the operational environmental profile defined by its intended use.

Where technical performance varies subject to environmental conditions, tests shall be carried out under a sufficient variety of environmental conditions (within the boundary limits of the operational environmental profile defined by its intended use) to give confidence of compliance for the affected technical requirements.

Normally it should be sufficient for all tests to be conducted using normal test conditions except where otherwise stated.

Test environments are specified in annex B.

For guidance on the use of other conditions to be used in order to show compliance reference can be made to ETSI TS 136 521-1 [1], clause F.1.1.

For each operating frequency band of the UE, the tests in the present document are performed with appropriate frequencies defined in ETSI TS 136 508 [2].

5.2 Void

5.3 Essential radio test suites

5.3.0 General

This clause describes the test suites that shall be used for E-UTRA FDD and TDD.

This clause also describes the test suites that shall be used for E-UTRA half duplex FDD and full duplex FDD for UE category 0 and UE category M1.

5.3.1 Transmitter Maximum Output Power

5.3.1.1 Transmitter maximum output power for Single Carrier

5.3.1.1.1 Method of test

5.3.1.1.1.1 Initial conditions

Test environment: normal, TL/VL, TL/VH, TH/VL, TH/VH; as specified in annex B.

Frequencies to be tested: low range, mid-range, high range; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidths to be tested: lowest, 5 MHz and highest channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

80

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.2.2.4.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.1.1.1.2 Procedure

- 1) Sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to table 6.2.2.1.4.1-1 of ETSI TS 136 521-1 [1]. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.
- 2) Send continuously uplink power control "up" commands in every uplink scheduling information to the UE; allow at least 200 ms for the UE to reach P_{UMAX} level.
- 3) Measure the mean power of the UE in the channel bandwidth of the radio access mode. The period of measurement shall be at least the continuous duration of one sub-frame (1 ms). For TDD slots with transient periods are not under test.
- 4) Repeat for applicable test frequencies, channel bandwidths, operating bands and environmental conditions.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.2.2.

5.3.1.1.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.2.1.2 in order to show compliance.

5.3.1.2 Transmitter maximum output power for intra-band contiguous Carrier Aggregation (DL CA and UL CA)

81

- 5.3.1.2.1 Method of test
- 5.3.1.2.1.1 Initial conditions

Test environment: normal, TL/VL, TL/VH, TH/VL, TH/VH; as specified in annex B.

Frequencies to be tested: low range, high range; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Test CC Combination setting (N_{RB_agg}): lowest N_{RB_agg} , highest N_{RB_agg} , as specified in ETSI TS 136 521-1 [1], in clause 5.4.2A.1 for the CA Configuration.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals for PCC are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.2.2A.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.1.2.1.2 Procedure

- 1) Configure SCC according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 for all downlink physical channels, except PHICH.
- 2) The SS shall configure SCC as per ETSI TS 136 508 [2], clause 5.2A.4.
- 3) SS activates SCC by sending the activation MAC-CE. Wait for at least 2 seconds.
- 4) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to table 6.2.2A.1.4.1-1 of ETSI TS 136 521-1 [1] on both PCC and SCC. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.
- 5) Send continuously uplink power control "up" commands in every uplink scheduling information to the UE; allow at least 200 ms for the UE to reach P_{UMAX} level.
- 6) Measure the mean transmitted power over all component carriers in the CA configuration of the radio access mode. The period of measurement shall be at least the continuous duration of one sub-frame (1 ms). For TDD slots with transient periods are not under test.
- 7) Repeat for applicable test frequencies, channel bandwidths, operating bands and environmental conditions.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.2.2A.1.

5.3.1.2.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.2.2.2 in order to show compliance.

5.3.1.2A Transmitter maximum output power for inter-band Carrier Aggregation (DL CA and UL CA)

82

- 5.3.1.2A.1 Method of test
- 5.3.1.2A.1.1 Initial conditions

Test environment: normal, TL/VL, TL/VH, TH/VL, TH/VH; as specified in annex B.

Frequencies to be tested: low range for PCC and SCC, high range for PCC and SCC; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Test CC Combination setting (N_{RB_agg}): lowest N_{RB_agg} , highest N_{RB_agg} , as specified in ETSI TS 136 521-1 [1], in clause 5.4.2A.1 for the CA Configuration.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals for PCC are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], table 6.2.2A.2.4.1-1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.1.2A.1.2 Test procedure

- 1) Configure SCC according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 for all downlink physical channels, except PHICH.
- 2) The SS shall configure SCC as per ETSI TS 136 508 [2], clause 5.2A.4.
- 3) SS activates SCC by sending the activation MAC-CE. Wait for at least 2 seconds.
- 4) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to table 6.2.2A.2.4.1-1 of ETSI TS 136 521-1 [1] on both PCC and SCC. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.
- 5) Send continuously uplink power control "up" commands in every uplink scheduling information to the UE; allow at least 200 ms for the UE to reach P_{UMAX} level.
- 6) Measure the mean transmitted power over all component carriers in the CA configuration of the radio access mode. The period of measurement shall be at least the continuous duration of one sub-frame (1 ms). For TDD slots with transient periods are not under test.
- 7) Repeat for applicable test frequencies, channel bandwidths, operating bands and environmental conditions.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.2.2A.2.

5.3.1.2A.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.2.2.2 in order to show compliance.

5.3.1.3 Transmitter maximum output power for UL-MIMO

5.3.1.3.1 Method of test

5.3.1.3.1.1 Initial conditions

Test environment: normal, TL/VL, TL/VH, TH/VL, TH/VH; as specified in annex B.

Frequencies to be tested: low range, mid-range, high range; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidths to be tested: lowest, 5 MHz and highest channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.2.2B.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.1.3.1.2 Procedure

- SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to table 6.2.2B.4.1-1 of ETSI TS 136 521-1 [1]. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.
- 2) Send continuously uplink power control "up" commands in every uplink scheduling information to the UE; allow at least 200 ms for the UE to reach P_{UMAX} level.
- 3) Measure the mean power of the UE in the channel bandwidth of the radio access mode. The period of measurement shall be at least the continuous duration of one sub-frame (1 ms). For TDD slots with transient periods are not under test.
- 4) Repeat for applicable test frequencies, channel bandwidths, operating bands and environmental conditions.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.2.2B.

5.3.1.3.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.2.3.2 in order to show compliance.

5.3.1.4 Transmitter maximum output power for category NB1

- 5.3.1.4.1 Method of Test
- 5.3.1.4.1.1 Initial conditions

Test environment: normal, TL/VL, TL/VH, TH/VL, TH/VH; as specified in annex B.

Frequencies to be tested: frequency ranges as defined in ETSI TS 136 521-1 [1], clause K.1.2 as specified in ETSI TS 136 508 [2], clause 8.1.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors using only the main UE Tx/Rx antenna.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 8.1.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.4.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.2.2F.4.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 2A-NB with CP CIoT Optimization according to ETSI TS 136 508 [2], clause 8.1.5.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.1.4.1.2 Procedure

- SS sends uplink scheduling information for each UL HARQ process via NPDCCH DCI format N0 for C_RNTI to schedule the UL RMC according to table 6.2.2F.4.1-1 of ETSI TS 136 521-1 [1]. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC (UE should be already transmitting P_{UMAX} after initial conditions setting).
- 2) Measure the mean power of the UE in the channel bandwidth of the radio access mode. The period of measurement shall be at least the continuous duration of one sub-frame (1 ms) for subcarrier spacing of 15 kHz or one slot (2 ms) excluding the 2 304 Ts gap when UE is not transmitting for sub-carrier spacing of 3,75 kHz. Half-duplex guard subframes are not under test.
- NOTE: For configuration IDs applicable to UE depending on UE capability in Test Configuration Table with different UL sub-carrier spacing, the SS releases the connection through State 3A-NB and finally ensures that the UE is in State 2A-NB with CP CIoT Optimization according to ETSI TS 136 508 [2], clause 8.1.5 using the appropriate UL subcarrier spacing in Random Access Response message.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.2.2F.

5.3.1.4.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.2.4.2 in order to show compliance.

5.3.1.5 Transmitter maximum output power for UE category 0

- 5.3.1.5.1 Method of test
- 5.3.1.5.1.1 Initial conditions

Same initial conditions as in clause 5.3.1.1.1 with the following exceptions:

- Instead of clause 6.2.2.4.1 in ETSI TS 136 521-1 [1] → use clause 6.2.2E.4.1 in ETSI TS 136 521-1 [1].
- Connect the SS to the UE antenna connectors as shown in ETSI TS 136 508 [2] annex A, figure A.3 using only main UE Tx/Rx antenna.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.1.5.1.2 Procedure

Same test procedure as in clause 5.3.1.1.1.2 with the following exception for HD-FDD:

• In step 3), slots with transient periods are not under test. Half-duplex guard subframes are not under test.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.2.2E.

5.3.1.5.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.2.1.2 in order to show compliance.

5.3.1.6 Transmitter maximum output power for UE category M1

5.3.1.6.1 Method of test

5.3.1.6.1.1 Initial conditions

Test environment: normal, TL/VL, TL/VH, TH/VL, TH/VH; as specified in annex B.

Frequencies to be tested: low range, mid-range, high range; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidths to be tested: highest channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors as shown in ETSI TS 136 508 [2] annex A, figure A.3 using only main UE Tx/Rx antenna.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.2.2EA.4.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF-CE according to ETSI TS 136 508 [2], clause 5.2A.2AA.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.1.6.1.2 Procedure

- 1) SS sends uplink scheduling information for each UL HARQ process via MPDCCH DCI format 6-0A for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 6.2.2EA.4.1-1. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.
- 2) Send continuously uplink power control "up" commands in every uplink scheduling information to the UE; allow at least 200 ms for the UE to reach P_{UMAX} level.
- 3) Measure the mean power of the UE in the channel bandwidth of the radio access mode. The period of measurement shall be at least the continuous duration of one sub-frame (1 ms). For TDD slots with transient periods are not under test. For HD-FDD slots with transient periods and Half-duplex guard subframe are not under test.
- 4) Repeat for applicable test frequencies, channel bandwidths, operating bands and environmental conditions.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.2.2EA.

5.3.1.6.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.2.5.2 in order to show compliance.

85

5.3.2 Transmitter Spectrum Emission Mask

5.3.2.1 Transmitter spectrum emission mask for Single Carrier

- 5.3.2.1.1 Method of test
- 5.3.2.1.1.1 Initial conditions

Test environment: normal, as specified in annex B.

Frequencies to be tested: low range, mid-range and high range; as specified in ETSI TS 136 508 [2], clause 4.3.1.

86

Channel bandwidths to be tested: lowest, 5 MHz, 10 MHz and highest channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.6.2.1.4.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.2.1.1.2 Procedure

- 1) SS sends uplink scheduling information via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 6.6.2.1.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 2) Send continuously uplink power control "up" commands in the uplink scheduling information to the UE until the UE transmits at P_{UMAX} level.
- 3) Measure the power of the transmitted signal with a measurement filter of bandwidths according to tables 4.2.3.1.2-1 or 4.2.3.1.2-2 or 4.2.3.1.2-3, as applicable. The centre frequency of the filter shall be stepped in continuous steps according to the same table. The measured power shall be recorded for each step. The measurement period shall capture the active TSs.
- 4) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.6.2.1.

5.3.2.1.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.3.1.2 in order to show compliance.

5.3.2.2 Transmitter spectrum emission mask for intra-band contiguous Carrier Aggregation (DL CA and UL CA)

87

- 5.3.2.2.1 Method of test
- 5.3.2.2.1.1 Initial conditions

Test environment: normal; as specified in annex B.

Frequencies to be tested: low range, high range; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Test CC Combination setting (N_{RB_agg}): lowest N_{RB_agg} , highest N_{RB_agg} , as specified in ETSI TS 136 521-1 [1], in clause 5.4.2A.1 for the CA Configuration.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals for PCC are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.6.2.1A.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.2.2.1.2 Procedure

- 1) Configure SCC according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 for all downlink physical channels, except PHICH.
- 2) The SS shall configure SCC as per ETSI TS 136 508 [2], clause 5.2A.4.
- 3) SS activates SCC by sending the activation MAC-CE. Wait for at least 2 seconds.
- 4) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to table 6.6.2.1A.1.4.1-1 of ETSI TS 136 521-1 [1] on both PCC and SCC. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.
- 5) Send continuously uplink power control "up" commands in every uplink scheduling information to the UE; allow at least 200 ms for the UE to reach P_{UMAX} level.
- 6) Measure the power of the transmitted signal with a measurement filter of bandwidths according to tables 4.2.3.2.2-1 or 4.2.3.2.2-2, as applicable. The centre frequency of the filter shall be stepped in continuous steps according to the same table. The measured power shall be recorded for each step. The measurement period shall capture the active TSs.
- 7) Repeat for applicable test frequencies, channel bandwidths, operating bands and environmental conditions.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.6.2.1A.1.

5.3.2.2.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.3.2.2 in order to show compliance.

5.3.2.2A Transmitter spectrum emission mask for inter-band Carrier Aggregation (DL CA and UL CA)

88

- 5.3.2.2A.1 Method of test
- 5.3.2.2A.1.1 Initial conditions

Test environment: normal; as specified in annex B.

Frequencies to be tested: low range for PCC and SCC, high range for PCC and SCC; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Test CC Combination setting (N_{RB_agg}): lowest N_{RB_agg} , highest N_{RB_agg} , as specified in ETSI TS 136 521-1 [1], in clause 5.4.2A.1 for the CA Configuration.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals for PCC are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.6.2.1A.2.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.2.2A.1.2 Procedure

- 1) Configure SCC according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 for all downlink physical channels, except PHICH.
- 2) The SS shall configure SCC as per ETSI TS 136 508 [2], clause 5.2A.4.
- 3) SS activates SCC by sending the activation MAC-CE. Wait for at least 2 seconds.
- 4) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to table 6.6.2.1A.2.4.1-1 of ETSI TS 136 521-1 [1] on both PCC and SCC. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.
- 5) Send continuously uplink power control "up" commands in every uplink scheduling information to the UE; allow at least 200 ms for the UE to reach P_{UMAX} level.
- 6) Measure the power of the transmitted signal for PCC with a measurement filter of bandwidths according to tables 4.2.3.1.2-1 or 4.2.3.1.2-2, as applicable. The centre frequency of the filter shall be stepped in continuous steps according to the same table. The measured power shall be recorded for each step. The measurement period shall capture the active TSs.
- 7) Measure the power of the transmitted signal for SCC with a measurement filter of bandwidths according to tables 4.2.3.1.2-1 or 4.2.3.1.2-2, as applicable. The centre frequency of the filter shall be stepped in continuous steps according to the same table. The measured power shall be recorded for each step. The measurement period shall capture the active TSs.
- 8) Repeat for applicable test frequencies, channel bandwidths, operating bands and environmental conditions.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.6.2.1A.2.

5.3.2.2A.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.3.1.2 in order to show compliance.

5.3.2.3 Transmitter spectrum emission mask for UL-MIMO

5.3.2.3.1 Method of test

5.3.2.3.1.1 Initial conditions

Test environment: normal, as specified in annex B.

Frequencies to be tested: low range, mid-range and high range; as specified in ETSI TS 136 508 [2], clause 4.3.1.

89

Channel bandwidths to be tested: lowest, 5 MHz, 10 MHz and highest channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.6.2.1B.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.2.3.1.2 Procedure

- 1) SS sends uplink scheduling information via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 6.6.2.1B.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 2) Send continuously uplink power control "up" commands in the uplink scheduling information to the UE until the UE transmits at P_{UMAX} level.
- 3) Measure the power of the transmitted signal with a measurement filter of bandwidths according to tables 4.2.3.1.2-1 or 4.2.3.1.2-2, as applicable. The centre frequency of the filter shall be stepped in continuous steps according to the same table. The measured power shall be recorded for each step. The measurement period shall capture the active TSs.
- 4) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.6.2.1B.

5.3.2.3.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.3.3.2 in order to show compliance.

5.3.2.4 Transmitter spectrum emission mask for Multi-Cluster PUSCH within a component carrier

90

- 5.3.2.4.1 Method of test
- 5.3.2.4.1.1 Initial conditions

Test environment: normal, as specified in annex B.

Frequencies to be tested: low range, mid-range and high range; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidths to be tested: lowest, 5 MHz, 10 MHz and highest channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.6.2.1_1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.2.4.1.2 Procedure

- 1) SS sends uplink scheduling information via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 6.6.2.1_1.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 2) Send continuously uplink power control "up" commands in the uplink scheduling information to the UE until the UE transmits at P_{UMAX} level.
- 3) Measure the power of the transmitted signal with a measurement filter of bandwidths according to tables 4.2.3.1.2-1 or 4.2.3.1.2-2, as applicable. The centre frequency of the filter shall be stepped in continuous steps according to the same table. The measured power shall be recorded for each step. The measurement period shall capture the active TSs.
- 4) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.6.2.1.1.

5.3.2.4.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.3.4.2 in order to show compliance.

5.3.2.5 Transmitter spectrum emission mask for category NB1

5.3.2.5.1 Method of test

5.3.2.5.1.1 Initial conditions

Test environment: normal, as specified in annex B.

Frequencies to be tested: frequency ranges defined in ETSI TS 136 521-1 [1], clause K.1.1; as specified in ETSI TS 136 508 [2], clause 8.1.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors using only main UE Tx/Rx antenna.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 8.1.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.4.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.6.2.1F.4.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 2A-NB with CP CIoT Optimization according to ETSI TS 136 508 [2], clause 8.1.5.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.2.5.1.2 Procedure

- SS sends uplink scheduling information via NPDCCH DCI format N0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 6.6.2.1F.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC (UE should be already transmitting P_{UMAX} after Initial Conditions setting).
- 2) Measure the mean power of the UE in the channel bandwidth of the radio access mode according to the test configuration, which shall meet the requirements described in ETSI TS 136 521-1 [1], table 6.2.3F.5-1. The measurement duration is at least one sub-frame (1 ms) for 15 kHz channel spacing, and at least a 2 ms slot (excluding the 2 304Ts gap when UE is not transmitting) respectively for the 3,75 kHz channel spacing.
- 3) Measure the power of the transmitted signal with a measurement filter of bandwidths according to table 4.2.3.5.2-1, as applicable. The centre frequency of the filter shall be stepped in continuous steps according to the same table. The measured power shall be recorded for each step. The measurement period shall capture the active TSs.
- NOTE: For configuration IDs applicable to UE depending on UE capability in Test Configuration Table with different UL sub-carrier spacing, the SS releases the connection through State 3A-NB and finally ensures that the UE is in State 2A-NB with CP CIoT Optimization according to ETSI TS 136 508 [2], clause 8.1.5 using the appropriate UL subcarrier spacing in Random Access Response message.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.6.2.1.

5.3.2.5.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.3.5.2 in order to show compliance.

5.3.2.6 Transmitter spectrum emission mask for UE category 0

5.3.2.6.1 Method of test

5.3.2.6.1.1 Initial conditions

Same initial conditions as in clause 5.3.2.1.1.1 with following exceptions:

- Instead of clause 6.6.2.1.4.1 in ETSI TS 136 521-1 [1] → use clause 6.6.2.1E.4.1 in ETSI TS 136 521-1 [1].
- Connect SS to the UE antenna connectors as shown in ETSI TS 136 508 [2], annex A, figure A.3 using only main UE Tx/Rx antenna.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.2.6.1.2 Procedure

Same test procedure as in clause 5.3.2.1.1.2 with following exception:

• Instead of table 6.6.2.1.4.1-1 in ETSI TS 136 521-1 [1] → use table 6.6.2.1E.4.1-1 in ETSI TS 136 521-1 [1].

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.6.2.1E.

5.3.2.6.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.3.1.2 in order to show compliance.

5.3.2.7 Transmitter spectrum emission mask for UE category M1

5.3.2.7.1 Method of test

5.3.2.7.1.1 Initial conditions

Test environment: normal, as specified in annex B.

Frequencies to be tested: low range, mid-range and high range; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidths to be tested: lowest, 5 MHz, 10 MHz and 15 MHz channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors as shown in ETSI TS 136 508 [2], annex A, figure A.3 using only main UE Tx/Rx antenna.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.6.2.1EA.4.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF-CE according to ETSI TS 136 508 [2], clause 5.2A.2AA.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.2.7.1.2 Procedure

1) SS sends uplink scheduling information via PDCCH DCI format 6-0A for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 6.6.2.1EA.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.

93

- 2) Send continuously uplink power control "up" commands in the uplink scheduling information to the UE until the UE transmits at P_{UMAX} level.
- 3) Measure the power of the transmitted signal with a measurement filter of bandwidths according to tables 4.2.3.1.2-1 and 4.2.3.1.2-3, as applicable. The centre frequency of the filter shall be stepped in continuous steps according to the same table. The measured power shall be recorded for each step. The measurement period shall capture the active TSs.
- 4) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.6.2.1EA.

5.3.2.7.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.3.1.2 in order to show compliance.

5.3.3 Transmitter Spurious Emissions

5.3.3.1 Transmitter spurious emissions for Single Carrier

5.3.3.1.1 Method of test

5.3.3.1.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

Frequencies to be tested: low range, mid-range, high range; see ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidth to be tested: lowest, 5 MHz and highest channel bandwidth as defined in ETSI TS 136 508 [2].

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.6.3.1.4.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.3.1.1.2 Procedure

- SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 6.6.3.1.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- Send continuously Up power control commands in the uplink scheduling information to the UE until the UE transmits at P_{UMAX} level.

3) For each applicable requirement in tables from 4.2.4.1.2-2 to 4.2.4.1.2-6; measure the power of the transmitted signal with a measurement filter of bandwidths. The centre frequency of the filter shall be stepped in contiguous steps according to the tables. The measured power shall be verified for each step. The measurement period shall capture the active time slots.

94

4) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clauses 6.6.3.1, 6.6.3.2 and 6.6.3.3.

5.3.3.1.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.4.1.2 in order to show compliance.

5.3.3.2 Transmitter spurious emissions for intra-band contiguous Carrier Aggregation (DL CA and UL CA)

- 5.3.3.2.1 Method of test
- 5.3.3.2.1.1 Initial conditions

Test environment: normal; as specified in annex B.

Frequencies to be tested: low range, high range; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Test CC Combination setting (N_{RB_agg}): lowest N_{RB_agg} , highest N_{RB_agg} , as specified in ETSI TS 136 521-1 [1], clause 5.4.2A.1 for the CA Configuration.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals for PCC are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.6.3.1A.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.3.2.1.2 Procedure

- 1) Configure SCC according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 for all downlink physical channels, except PHICH.
- 2) The SS shall configure SCC as per ETSI TS 136 508 [2], clause 5.2A.4.
- 3) SS activates SCC by sending the activation MAC-CE. Wait for at least 2 seconds.
- 4) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to table 6.6.3.1A.1.4.1-1 of ETSI TS 136 521-1 [1] on both PCC and SCC. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.
- 5) Send continuously uplink power control "up" commands in every uplink scheduling information to the UE; allow at least 200 ms for the UE to reach P_{UMAX} level.

6) For each applicable requirement in tables from 4.2.4.2.2-2 to 4.2.4.2.2-7; measure the power of the transmitted signal with a measurement filter of bandwidths for each component carrier. The centre frequency of the filter shall be stepped in contiguous steps according to the tables. The measured power shall be verified for each step. The measurement period shall capture the active time slots.

95

7) Repeat for applicable test frequencies, channel bandwidths, operating bands and environmental conditions.

Details of the test method can be found in ETSI TS 136 521-1 [1], clauses 6.6.3.1A.1 and 6.6.3.2A.1.

5.3.3.2.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.4.2.2 in order to show compliance.

5.3.3.2A Transmitter spurious emissions for inter-band Carrier Aggregation (DL CA and UL CA)

- 5.3.3.2A.1 Method of test
- 5.3.3.2A.1.1 Initial conditions

Test environment: normal; as specified in annex B.

Frequencies to be tested: low range for PCC and SCC, high range for PCC and SCC; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Test CC Combination setting (N_{RB_agg}): lowest N_{RB_agg} , highest N_{RB_agg} , as specified in ETSI TS 136 521-1 [1], clause 5.4.2A.1 for the CA Configuration.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals for PCC are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.6.3.1A.2.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.3.2A.1.2 Procedure

- 1) Configure SCC according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 for all downlink physical channels, except PHICH.
- 2) The SS shall configure SCC as per ETSI TS 136 508 [2], clause 5.2A.4.
- 3) SS activates SCC by sending the activation MAC-CE. Wait for at least 2 seconds.
- 4) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to table 6.6.3.1A.2.4.1-1 of ETSI TS 136 521-1 [1] on both PCC and SCC. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.
- 5) Send continuously uplink power control "up" commands in every uplink scheduling information to the UE; allow at least 200 ms for the UE to reach P_{UMAX} level.

- 6) For each applicable requirement in table 4.2.4.2.2-3A; measure the power of the transmitted signal with a measurement filter of bandwidths for each component carrier. The centre frequency of the filter shall be stepped in contiguous steps according to the tables. The measured power shall be verified for each step. The measurement period shall capture the active time slots.
- 7) Repeat for applicable test frequencies, channel bandwidths, operating bands and environmental conditions.

Details of the test method can be found in ETSI TS 136 521-1 [1], clauses 6.6.3.1A.2 and 6.6.3.2A.2.

5.3.3.2A.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.4.2.2 in order to show compliance.

5.3.3.3 Transmitter spurious emissions for UL-MIMO

- 5.3.3.3.1 Method of test
- 5.3.3.3.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

Frequencies to be tested: low range, mid-range, high range; see ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidth to be tested: lowest, 5 MHz and highest channel bandwidth as defined in ETSI TS 136 508 [2].

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.6.3B.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.3.3.1.2 Procedure

- 1) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 6.6.3B.1.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 2) Send continuously Up power control commands in the uplink scheduling information to the UE until the UE transmits at P_{UMAX} level.
- 3) For each applicable requirement in tables 4.2.4.1.2-2 and 4.2.4.1.2-3, measure the power of the transmitted signal with a measurement filter of bandwidths. The centre frequency of the filter shall be stepped in contiguous steps according to the tables. The measured power shall be verified for each step. The measurement period shall capture the active time slots.
- 4) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clauses 6.6.3B.1 and 6.6.3B.2.

5.3.3.3.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.4.3.2 in order to show compliance.

5.3.3.4 Transmitter spurious emissions for Multi-Cluster PUSCH within a component carrier

97

5.3.3.4.1 Method of test

5.3.3.4.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

Frequencies to be tested: low range, mid-range, high range; see ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidth to be tested: Highest channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.6.3.1_1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.3.4.1.2 Procedure

- 1) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 6.6.3.1_1.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 2) Send continuously Up power control commands in the uplink scheduling information to the UE until the UE transmits at P_{UMAX} level.
- 3) For the requirements in table 4.2.4.1.2-2, measure the power of the transmitted signal with a measurement filter of bandwidths. The centre frequency of the filter shall be stepped in contiguous steps according to the tables. The measured power shall be verified for each step. The measurement period shall capture the active time slots.
- 4) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.6.3.1.1.

5.3.3.4.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.4.4.2 in order to show compliance.

5.3.3.5 Transmitter spurious emissions for category NB1

5.3.3.5.1 Method of test

5.3.3.5.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

Frequencies to be tested: Frequency ranges defined in ETSI TS 136 521-1 [1], clause K.1.2, see ETSI TS 136 508 [2], clause 8.1.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors using only main UE Tx/Rx antenna.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 8.1.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1.1 and H.4.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.6.3F.1.4.1 and clause 6.6.3F.2.4.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 2A-NB with CP CIoT Optimization according to ETSI TS 136 508 [2], clause 8.1.5.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.3.5.1.2 Procedure

- SS sends uplink scheduling information via NPDCCH DCI format N0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 6.6.3F.1.4.1-1 or table 6.6.3F.2.4.1-1 (co-existence) and with the scheduling pattern according to ETSI TS 136 521-1 [1], clause A.2. Since the UE has no payload and no loopback data to send, the UE transmits uplink MAC padding bits on the UL RMC (UE should be already transmitting P_{UMAX} after Initial Conditions setting).
- 2) For each applicable requirement in tables 4.2.4.1.2-1 and 4.2.4.1.2-2, measure the power of the transmitted signal with a measurement filter of bandwidths. The centre frequency of the filter shall be stepped in contiguous steps according to the tables. The measured power shall be verified for each step. The measurement period shall capture the active time slots.
- NOTE: For configuration IDs applicable to UE depending on UE capability in Test Configuration Table with different UL sub-carrier spacing, the SS releases the connection through State 3A-NB and finally ensures that the UE is in State 2A-NB with CP CIoT Optimization according to ETSI TS 136 508 [2], clause 8.1.5 using the appropriate UL subcarrier spacing in Random Access Response message.

Details of the test method can be found in ETSI TS 136 521-1 [1], clauses 6.6.3F.1 and 6.6.3F.2.

5.3.3.5.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.4.5.2 in order to show compliance.

5.3.3.6 Transmitter spurious emissions for UE category 0

5.3.3.6.1 Method of test

5.3.3.6.1.1 Initial conditions

Same initial conditions as in clause 5.3.3.1.1.1 with following exceptions for HD-FDD:

- Instead of clause 6.6.3.1.4.1 in ETSI TS 136 521-1 [1] → use clauses 6.6.3E.1.4.1 or 6.6.3E.2.4.1 in ETSI TS 136 521-1 [1], as applicable.
- Connect SS to the UE antenna connectors as shown in ETSI TS 136 508 [2] annex A, figure A.3 using only main UE Tx/Rx antenna.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.3.6.1.2 Procedure

Same test procedure as in clause 5.3.3.1.1.2 with following exception for HD-FDD:

• In step 3), slots with transient periods are not under test. Half-duplex guard subframes are not under test.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.6.3E.

5.3.3.6.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.4.1.2 in order to show compliance.

5.3.3.7 Transmitter spurious emissions for UE category M1

- 5.3.3.7.1 Method of test
- 5.3.3.7.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

Frequencies to be tested: low range, mid-range, high range; see ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidth to be tested:

- for general requirements: lowest channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1; or
- for operating band specific requirements to protected bands: lowest, 5 MHz and highest channel bandwidths as defined in ETSI TS 136 508 [2], clause 4.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect SS to the UE antenna connectors as shown in ETSI TS 136 508 [2], annex A, figure A.7 using only main UE Tx/Rx antenna.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], use clauses 6.6.3EA.1.4.1 or 6.6.3EA.2.4.1, as applicable.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF-CE according to ETSI TS 136 508 [2], clause 5.2A.2AA.

NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.3.7.1.2 Procedure

- SS sends uplink scheduling information for each UL HARQ process via MPDCCH DCI format 6-0A for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], use tables 6.6.3EA.1.4.1-1 or 6.6.3EA.2.4.1-1, as applicable. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 2) Send continuously uplink power control "up" commands in the uplink scheduling information to the UE until the UE transmits at P_{UMAX} level.
- 3) For each applicable requirement in tables from 4.2.4.1.2-2 to 4.2.4.1.2-4 measure the power of the transmitted signal with a measurement filter of bandwidths. The centre frequency of the filter shall be stepped in contiguous steps according to the tables. The measured power shall be verified for each step. The measurement period shall capture the active time slots.
- 4) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.6.3EA.

5.3.3.7.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.4.1.2 in order to show compliance.

5.3.4 Transmitter Minimum Output Power

5.3.4.1 Transmitter minimum output power for Single Carrier

5.3.4.1.1 Method of test

5.3.4.1.1.1 Initial conditions

Test Environment: normal, TL/VL, TL/VH, TH/VL, TH/VH; as specified in annex B.

Frequencies to be tested: low range, mid-range and high range; see ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidths to be tested: lowest, 5 MHz and highest channel bandwidth, as specified in ETSI TS 136 508 [2], clause 4.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.3.2.4.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.4.1.1.2 Procedure

- 1) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 6.3.2.4.1-1. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.
- 2) Send continuous uplink power control "down" commands in the uplink scheduling information to the UE to ensure that the UE transmits at its minimum output power.
- 3) Measure the mean power of the UE in the associated measurement bandwidth specified in table 4.2.5.1.2-1 for the specific channel bandwidth under test. The period of measurement shall be the continuous duration of one sub-frame (1 ms). For TDD slots with transient periods are not under test.
- 4) Repeat for applicable test frequencies, channel bandwidths, operating bands and environmental conditions.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.3.2.

5.3.4.1.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.5.1.2 in order to show compliance.

5.3.4.2 Transmitter minimum output power for intra-band contiguous Carrier Aggregation (DL CA and UL CA)

5.3.4.2.1 Method of test

5.3.4.2.1.1 Initial conditions

Test environment: normal, TL/VL, TL/VH, TH/VL, TH/VH; as specified in annex B.

Frequencies to be tested: low range, high range; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Test CC Combination setting (N_{RB_agg}): lowest N_{RB_agg} , highest N_{RB_agg} , as specified in ETSI TS 136 521-1 [1], clause 5.4.2A.1 for the CA Configuration.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals for PCC are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.3.2A.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.4.2.1.2 Procedure

- 1) Configure SCC according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 for all downlink physical channels, except PHICH.
- 2) The SS shall configure SCC as per ETSI TS 136 508 [2], clause 5.2A.4.
- 3) SS activates SCC by sending the activation MAC-CE. Wait for at least 2 seconds.

4) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to table 6.3.2A.1.4.1-1 of ETSI TS 136 521-1 [1] on both PCC and SCC. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.

102

- 5) Send continuously uplink power control "down" commands in every uplink scheduling information to the UE to ensure that the UE transmits at its minimum output power.
- 6) Measure the mean transmitted power of each component carrier in the CA configuration of the radio access mode. The period of measurement shall be at least the continuous duration of one sub-frame (1 ms). For TDD slots with transient periods are not under test.
- 7) Repeat for applicable test frequencies, channel bandwidths, operating bands and environmental conditions.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.3.2A.1.

5.3.4.2.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.5.2.2 in order to show compliance.

5.3.4.2A Transmitter minimum output power for inter-band Carrier Aggregation (DL CA and UL CA)

- 5.3.4.2A.1 Method of test
- 5.3.4.2A.1.1 Initial conditions

Test environment: normal, TL/VL, TL/VH, TH/VL, TH/VH; as specified in annex B.

Frequencies to be tested: low range for PCC and SCC, high range for PCC and SCC; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Test CC Combination setting (N_{RB_agg}): lowest N_{RB_agg} , highest N_{RB_agg} , as specified in ETSI TS 136 521-1 [1], clause 5.4.2A.1 for the CA Configuration.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals for PCC are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.3.2A.2.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.4.2A.1.2 Procedure

- 1) Configure SCC according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 for all downlink physical channels, except PHICH.
- 2) The SS shall configure SCC as per ETSI TS 136 508 [2], clause 5.2A.4.
- 3) SS activates SCC by sending the activation MAC-CE. Wait for at least 2 seconds.

4) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to table 6.3.2A.2.4.1-1 of ETSI TS 136 521-1 [1] on both PCC and SCC. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.

103

- 5) Send continuously uplink power control "down" commands in every uplink scheduling information to the UE to ensure that the UE transmits at its minimum output power.
- 6) Measure the mean transmitted power of each component carrier in the CA configuration of the radio access mode. The period of measurement shall be at least the continuous duration of one sub-frame (1 ms). For TDD slots with transient periods are not under test.
- 7) Repeat for applicable test frequencies, channel bandwidths, operating bands and environmental conditions.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.3.2A.2.

5.3.4.2A.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.5.2.2 in order to show compliance.

5.3.4.3 Transmitter minimum output power for UL-MIMO

5.3.4.3.1 Method of test

5.3.4.3.1.1 Initial conditions

Test environment: normal, TL/VL, TL/VH, TH/VL, TH/VH; as specified in annex B.

Frequencies to be tested: low range, mid-range, high range; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidths to be tested: lowest, 5 MHz and highest channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.3.2B.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.4.1.1.2 Procedure

- 1) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to table 6.3.2B.1.4.1-1 of ETSI TS 136 521-1 [1]. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.
- 2) Send continuously uplink power control "down" commands in the uplink scheduling information to the UE to ensure that the UE transmits at its minimum output power.
- 3) Measure the sum of mean power of the UE at each UE antenna connector in the associated measurement bandwidth specified in table 4.2.5.3.2-1 for the specific channel bandwidth under test. The period of measurement shall be the continuous duration of one sub-frame (1 ms). For TDD slots with transient periods are not under test.

4) Repeat for applicable test frequencies, channel bandwidths, operating bands and environmental conditions.

104

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.3.2B.

5.3.4.3.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.5.3.2 in order to show compliance.

5.3.4.4 Transmitter minimum output power for category NB1

5.3.4.4.1 Method of test

5.3.4.4.1.1 Initial conditions

Test Environment: normal, TL/VL, TL/VH, TH/VL, TH/VH; as specified in annex B.

Frequencies to be tested: frequency ranges defined in ETSI TS 136 521-1 [1], clause K.1.1, see ETSI TS 136 508 [2], clause 8.1.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors using only the main UE Tx/Rx antenna.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 8.1.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.3.2F.4.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 2A-NB with CP CIoT Optimization according to ETSI TS 136 508 [2], clause 8.1.5.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.4.4.1.2 Procedure

- SS sends uplink scheduling information for each UL HARQ process via NPDCCH DCI format N0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 6.3.2F.4.1-1. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.
- 2) Measure the mean power of the UE in the channel bandwidth of the radio access mode. The period of measurement shall be at least the continuous duration of one sub-frame (1 ms) for sub-carrier spacing of 15 kHz or one slot (2 ms) excluding the 2 304Ts gap when UE is not transmitting for sub-carrier spacing of 3,75 kHz. Half-duplex guard subframes are not under test.
- NOTE: For configuration IDs applicable to UE depending on UE capability in Test Configuration Table with different UL sub-carrier spacing, the SS releases the connection through State 3A-NB and finally ensures that the UE is in State 2A-NB with CP CIoT Optimization according to ETSI TS 136 508 [2], clause 8.1.5 using the appropriate UL subcarrier spacing in Random Access Response message.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.3.2.

5.3.4.4.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.5.4.2 in order to show compliance.

5.3.4.5 Transmitter minimum output power for UE category 0

5.3.4.5.1 Method of test

5.3.4.5.1.1 Initial conditions

Same initial conditions as in clause 5.3.4.1.1.1 with following exceptions:

- Instead of clause 6.3.2.4.1 in ETSI TS 136 521-1 [1] → use clause 6.3.2E.4.1 in ETSI TS 136 521-1 [1].
- Connect SS to the UE antenna connectors as shown in ETSI TS 136 508 [2] annex A, figure A.3 using only main UE Tx/Rx antenna.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.4.5.1.2 Procedure

Same test procedure as in clause 5.3.4.1.1.2 with the following exception for HD-FDD:

• In step 3), slots with transient periods are not under test. Half-duplex guard subframes are not under test.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.3.2E.

5.3.4.5.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.5.1.2 in order to show compliance.

5.3.4.6 Transmitter minimum output power for UE category M1

5.3.4.6.1 Method of test

5.3.4.6.1.1 Initial conditions

Test Environment: normal, TL/VL, TL/VH, TH/VL, TH/VH; as specified in annex B.

Frequencies to be tested: low range, mid-range and high range; see ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidths to be tested: 5 MHz channel bandwidth, as specified in ETSI TS 136 508 [2], clause 4.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors as shown in ETSI TS 136 508 [2] annex A, figure A.3 using only main UE Tx/Rx antenna.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.3.2EA.4.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF-CE according to ETSI TS 136 508 [2], clause 5.2A.2AA.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.4.6.1.2 Procedure

- 1) SS sends uplink scheduling information for each UL HARQ process via MPDCCH DCI format 6-0A for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 6.3.2EA.4.1-1. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.
- 2) Send continuous uplink power control "down" commands in the uplink scheduling information to the UE to ensure that the UE transmits at its minimum output power.
- 3) Measure the mean power of the UE in the associated measurement bandwidth specified in table 4.2.5.1.2-1 for the specific channel bandwidth under test. The period of measurement shall be the continuous duration of one sub-frame (1 ms). For TDD slots with transient periods are not under test. For HD-FDD slots with transient periods and Half-duplex guard subframe are not under test.
- 4) Repeat for applicable test frequencies, channel bandwidths, operating bands and environmental conditions.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.3.2EA.

5.3.4.6.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.5.1.2 in order to show compliance.

5.3.5 Receiver Adjacent Channel Selectivity (ACS)

5.3.5.1 Receiver Adjacent Channel Selectivity (ACS) for Single Carrier

5.3.5.1.1 Method of test

5.3.5.1.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

Frequencies to be tested: mid-range see ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidth to be tested: lowest, 5 MHz and highest channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS and interfering source to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.1 and uplink signals according to clauses H.1 and H.3.1.
- 4) The UL and DL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], table 7.5.4.1-1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.5.1.1.2 Procedure

- 1) SS transmits PDSCH via PDCCH DCI format 1A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.5.4.1-1. The SS sends downlink MAC padding bits on the DL RMC.
- 2) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.5.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.

3) Set the Downlink signal level to the value as defined in table 4.2.6.1.2-2 (Case 1). Send Uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.6.1.2-2 (Case 1) for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < $f \le 4,2$ GHz, for at least the duration of the Throughput measurement (obtain correct UE output power as specified in ETSI TS 136 521-1 [1], table 7.5.3-2).

107

- 4) Set the Interferer signal level to the value as defined in table 4.2.6.1.2-2 (Case 1) and frequency below the wanted signal, using a modulated interferer as defined in ETSI TS 136 521-1 [1], annex D.
- 5) Measure the average throughput for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].
- 6) Set the Downlink signal level to the value as defined in table 4.2.6.1.2-3 (Case 2). Send Uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.6.1.2-3 (Case 2) for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < $f \le 4,2$ GHz, for at least the duration of the throughput measurement (obtain correct UE output power as specified in ETSI TS 136 521-1 [1], table 7.5.3-3.
- 7) Set the Interferer signal level to the value as defined in table 4.2.6.1.2-3 (Case 2) and frequency below the wanted signal, using a modulated interferer as defined in ETSI TS 136 521-1 [1], annex D.
- 8) Measure the average throughput for a duration sufficient to achieve statistical significance according to ETSI TS 136 521-1 [1], annex G.
- 9) Repeat for applicable channel bandwidths in both Case 1 and Case 2.
- 10) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.5.

5.3.5.1.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.6.1.2 in order to show compliance.

5.3.5.2 Receiver Adjacent Channel Selectivity (ACS) for Carrier Aggregation in DL-only bands

- 5.3.5.2.1 Method of test
- 5.3.5.2.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

Frequencies to be tested: mid-range see ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidth to be tested: Highest $N_{RB agg}$ for PCC and SCC.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS and interfering source to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL and DL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], table 7.5A.3.4.1-1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.

NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.5.2.1.2 Procedure

- 1) Configure SCC according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.1 for all downlink physical channels.
- 2) The SS shall configure SCC as per ETSI TS 136 508 [2], clause 5.2A.4.
- 3) SS activates SCC by sending the activation MAC-CE. Wait for at least 2 seconds.
- SS transmits PDSCH via PDCCH DCI format 1A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.5A.3.4.1-1 on both PCC and SCC. The SS sends downlink MAC padding bits on the DL RMC.
- 5) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.5A.3.4.1-1 on PCC. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.
- 6) Set the Downlink signal level for PCC and SCC to the value as defined in table 4.2.6.1.2-2 (Case 1), or table 4.2.6.2.2-2 for operating band 46. Send Uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.6.1.2-2 (Case 1) for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < f $\le 4,2$ GHz, for at least the duration of the Throughput measurement.
- 7) Set the Interferer signal level to the value as defined in table 4.2.6.1.2-2 (Case 1), or table 4.2.6.2.2-2 for operating band 46, and frequency below the wanted signal, using a modulated interferer bandwidth as defined in ETSI TS 136 521-1 [1], annex D.
- 8) Measure the average throughput of SCC for a duration sufficient to achieve statistical significance according to ETSI TS 136 521-1 [1], clause G.2.
- 9) Repeat steps from 6 to 8, using an interfering signal above the wanted signal in Case 1 at step 7).
- 10) Set the Downlink signal level for PCC and SCC to the value as defined in table 4.2.6.1.2-3 (Case 2), or table 4.2.6.2.2-3 for operating band 46. Send Uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.6.1.2-3 (Case 2) for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < f \le 4,2 GHz, for at least the duration of the Throughput measurement.
- 11) Set the Interferer signal level to the value as defined in table 4.2.6.1.2-3 (Case 2), or table 4.2.6.2.2-3 for operating band 46, and frequency below the wanted signal, using a modulated interferer bandwidth as defined in ETSI TS 136 521-1 [1], annex D.
- 12) Measure the average throughput of SCC for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].
- 13) Repeat steps from 10 to 12, using an interfering signal above the wanted signal in Case 2 at step 11).

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.5.

5.3.5.2.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.6.2.2 in order to show compliance.

5.3.5.3 Receiver Adjacent Channel Selectivity (ACS) for category NB1

- 5.3.5.3.1 Method of test
- 5.3.5.3.1.1 Initial conditions

Test Environment: normal, as specified in annex B.
Frequencies to be tested: Frequency ranges defined in ETSI TS 136 521-1 [1], clause K.1.1, see ETSI TS 136 508 [2], clause 8.1.3.1.

109

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors using only the main UE Tx/Rx antenna, and noting that the interference can be either GSM or E-UTRA.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 8.1.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.2 and NPUSCH Format 2 is used to carry ACK/NACK on the uplink.
- 4) The DL Reference Measurement channel is set according to ETSI TS 136 521-1 [1], table 7.5F.4.1-1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 2A-NB with CP CIoT Optimization according to ETSI TS 136 508 [2], clause 8.1.5. Message contents are defined according to ETSI TS 136 521-1 [1], clause 7.5F.4.3.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.5.3.1.2 Procedure

- SS transmits NPDSCH via NPDCCH DCI format N1 for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.5F.4.1-1. The SS sends downlink MAC padding bits on the DL RMC. The UE will send HARQ feedback based on information contained in DCI format N1.
- 2) Set the Downlink signal level to the value defined for ACS1, GSM in table 4.2.6.3.2-1.
- 3) Set the Interferer signal level to the value defined for ACS1, GSM in table 4.2.6.3.2-1, with frequency below the wanted signal according to table 4.2.6.3.2-1, using a modulated interferer bandwidth as defined in ETSI TS 136 521-1 [1], clause D.2.
- 4) Measure the average throughput for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].
- 5) Repeat steps 2) to 4), using an interfering signal above the wanted signal at step 3).
- 6) Set the Downlink signal level to the value defined for ACS1, E-UTRA in table 4.2.6.3.2-1.
- 7) Set the Interferer signal level to the value defined for ACS1, E-UTRA in table 4.2.6.3.2-1, with frequency below the wanted signal according to table 4.2.6.3.2-1, using a modulated interferer bandwidth as defined in ETSI TS 136 521-1 [1], clause D.2.
- 8) Measure the average throughput for a duration sufficient to achieve statistical significance according to ETSI TS 136 521-1 [1], clause G.2.
- 9) Repeat steps 6) to 8), using an interfering signal above the wanted signal at step 7).
- 10) Release the connection through State 3A-NB.
- 11) Modify system information elements according to ETSI TS 136 521-1 [1], table 7.5F.4.3-1 and notify the UE via paging message with SytemInformationModification included.
- 12) Ensure the UE is in State 2A-NB with CP CIoT Optimization according to ETSI TS 136 508 [2], clause 8.1.5 using the new UL power control setting.
- 13) SS transmits NPDSCH via NPDCCH DCI format N1 for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.5F.4.1-1. The SS sends downlink MAC padding bits on the DL RMC. The UE will send HARQ feedback based on information contained in DCI format N1.
- 14) Set the Downlink signal level to the value defined for ACS2, GSM in table 4.2.6.3.2-1. For steps 14 to 17 and 18 to 21, use message contents with the exceptions defined in ETSI TS 136 521-1 [1], table 7.5F.4.3-1.

15) Set the Interferer signal level to the value defined for ACS2, GSM in table 4.2.6.3.2-1, with frequency below the wanted signal according to table 4.2.6.3.2-1, using a modulated interferer of 5 MHz bandwidth as defined in ETSI TS 136 521-1 [1], clause D.2.

110

- 16) Measure the average throughput for a duration sufficient to achieve statistical significance according to ETSI TS 136 521-1 [1], clause G.2.
- 17) Repeat steps 14 to 16, using an interfering signal above the wanted signal in step 15).
- 18) Set the Downlink signal level to the value defined for ACS2, E-UTRA according to table 4.2.6.3.2-1.
- 19) Set the Interferer signal level to the value defined for ACS2, E-UTRA in table 4.2.6.3.2-1, with frequency below the wanted signal according to table 4.2.6.3.2-1, using a modulated interferer of 5 MHz bandwidth as defined in ETSI TS 136 521-1 [1], clause D.2.
- 20) Measure the average throughput for a duration sufficient to achieve statistical significance according to ETSI TS 136 521-1 [1], clause G.2.
- 21) Repeat steps 18 to 20, using an interfering signal above the wanted signal at step 19).

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.5F.4.

5.3.5.3.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.6.2.2 in order to show compliance.

5.3.5.4 Receiver Adjacent Channel Selectivity (ACS) for UE category 0

5.3.5.4.1 Method of test

5.3.5.4.1.1 Initial conditions

Same initial conditions as in clause 5.3.5.1.1.1 with the following exceptions:

- Instead of table 7.5.4.1-1 in ETSI TS 136 521-1 [1] → use table 7.5E.4.1-1 in ETSI TS 136 521-1 [1].
- Connect the SS to the UE antenna connectors as shown in ETSI TS 136 508 [2] annex A, figure A.4 using only main UE Tx/Rx antenna.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.5.4.1.2 Procedure

Same test procedure as in clause 5.3.5.1.1.2.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.5E.

5.3.5.4.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.6.1.2 in order to show compliance.

5.3.5.5 Receiver Adjacent Channel Selectivity (ACS) for UE category M1

- 5.3.5.5.1 Method of test
- 5.3.5.5.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

Frequencies to be tested: mid-range see ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidth to be tested: 1,4 MHz, 3 MHz and 5 MHz channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

111

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors as shown in ETSI TS 136 508 [2], annex A, figure A.4 using only main UE Tx/Rx antenna.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.1 and uplink signals according to clauses H.1 and H.3.1.
- 4) The UL and DL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], table 7.5EA.4.1-1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF-CE according to ETSI TS 136 508 [2], clause 5.2A.2AA.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.5.5.1.2 Procedure

- 1) SS transmits PDSCH via M-PDCCH DCI format 6-1A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.5EA.4.1-1. The SS sends downlink MAC padding bits on the DL RMC.
- 2) SS sends uplink scheduling information for each UL HARQ process via M-PDCCH DCI format 6-0A for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.5EA.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 3) Set the Downlink signal level to the value as defined in table 4.2.6.1.2-2 (Case 1). Send Uplink power control commands to the UE (less or equal to 1dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.6.1.2-2 (Case 1) for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < f \le 4,2 GHz, for at least the duration of the Throughput measurement.
- 4) Set the Interferer signal level to the value as defined in table 4.2.6.1.2-2 (Case 1) and frequency below the wanted signal, using a modulated interferer as defined in ETSI TS 136 521-1 [1], annex D.
- 5) Measure the average throughput for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].
- 6) Set the Downlink signal level to the value as defined in table 4.2.6.1.2-3 (Case 2). Send Uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.6.1.2-3 (Case 2) for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < f \le 4,2 GHz, for at least the duration of the throughput measurement.
- 7) Set the Interferer signal level to the value as defined in table 4.2.6.1.2-3 (Case 2) and frequency below the wanted signal, using a modulated interferer as defined in ETSI TS 136 521-1 [1], annex D.
- 8) Measure the average throughput for a duration sufficient to achieve statistical significance according to ETSI TS 136 521-1 [1], clause G.2.
- 9) Repeat for applicable channel bandwidths in both Case 1 and Case 2.
- 10) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.5EA.

5.3.5.5.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.6.1.2 in order to show compliance.

5.3.6 Receiver Blocking Characteristics

5.3.6.1 Receiver Blocking Characteristics for Single Carrier

- 5.3.6.1.1 Method of test
- 5.3.6.1.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

For In-band blocking, the frequencies to be tested are mid-range as defined in ETSI TS 136 508 [2], clause 4.3.1.

For out-of-band blocking, the frequency to be tested is low or high range as defined in ETSI TS 136 508 [2], clause 4.3.1.

For Narrow-band blocking, the frequencies to be tested are mid-range as defined in ETSI TS 136 508 [2], clause 4.3.1.

112

Channel bandwidth to be tested: lowest, 5 MHz and highest channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1. Range 3 of out-of-band blocking is tested only with the highest bandwidth.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.1 and uplink signals according to clauses H.1 and H.3.1.
- 4) The UL and DL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], table 7.6.1.4.1-1, or table 7.6.2.4.1-1, or table 7.6.2.4.1-1 as applicable.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.6.1.1.2 In-Band Procedure

- 1) SS transmits PDSCH via PDCCH DCI format 1A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.6.1.4.1-1. The SS sends downlink MAC padding bits on the DL RMC.
- 2) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.6.1.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 3) Set the parameters of the signal generator for an interfering signal below the wanted signal in Case 1 according to tables 4.2.7.1.2-1 and 4.2.7.1.2-2.
- 4) Set the downlink signal level according to the table 4.2.7.1.2-1. Send uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.7.1.2-1 for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < $f \le 4,2$ GHz, for at least the duration of the throughput measurement as specified in ETSI TS 136 521-1 [1].
- 5) Measure the average throughput for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].
- 6) Repeat steps from 3 to 5, using an interfering signal above the wanted signal in Case 1 at step 3).

7) Repeat steps from 3 to 6, using interfering signals in Case 2 at steps 3) and 6). The ranges of case 2 are covered in steps equal to the interferer bandwidth. The test frequencies are chosen in analogy to ETSI TS 136 521-1 [1], table 7.6.1.4.2-1.

113

8) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.6.1.

5.3.6.1.1.3 Out-Of-Band Procedure

- 1) SS transmits PDSCH via PDCCH DCI format 1A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.6.2.4.1-1. The SS sends downlink MAC padding bits on the DL RMC.
- 2) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.6.2.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 3) Set the parameters of the CW signal generator for an interfering signal according to table 4.2.7.1.2-4. The frequency step size is 1 MHz.
- 4) Set the downlink signal level according to the table 4.2.7.1.2-3. Send uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.7.1.2-3 for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < $f \le 4,2$ GHz, for at least the duration of the throughput measurement as specified in ETSI TS 136 521-1 [1].
- 5) Measure the average throughput for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].
- 6) For table 4.2.7.1.2-4 record the frequencies for which the throughput does not meet the requirements.
- 7) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.6.2.

5.3.6.1.1.4 Narrow-Band Procedure

- 1) SS transmits PDSCH via PDCCH DCI format 1A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.6.3.4.1-1. The SS sends downlink MAC padding bits on the DL RMC.
- 2) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.6.3.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 3) Set the parameters of the CW signal generator for an interfering signal below the wanted signal according to table 4.2.7.1.2-5.
- 4) Set the downlink signal level according to the table 4.2.7.1.2-5. Send uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.7.1.2-5 for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < $f \le 4,2$ GHz, for at least the duration of the throughput measurement as specified in ETSI TS 136 521-1 [1].
- 5) Measure the average throughput for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].
- 6) Repeat steps from 3 to 5, using an interfering signal above the wanted signal at step 3).
- 7) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.6.3.

5.3.6.1.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.7.1.2 in order to show compliance.

5.3.6.2 Receiver Blocking Characteristics for Carrier Aggregation in DL-only bands

114

5.3.6.2.1 Method of test

5.3.6.2.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

Frequencies to be tested: mid-range see ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidth to be tested: Highest N_{RB} agg for PCC and SCC.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.1 and uplink signals according to clauses H.1 and H.3.1.
- 4) The UL and DL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], table 7.6.1A.3.4.1-1 for in-band blocking and table 7.6.2A.3.4.1-1 for out-of-band blocking and table 7.6.3A.3.4.1-1 for narrow-band blocking.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.6.2.1.2 In-Band Procedure

- 1) Configure SCC according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.2 for all downlink physical channels except PHICH.
- 2) The SS shall configure SCC as per ETSI TS 136 508 [2], clause 5.2A.4.
- 3) SS activates SCC by sending the activation MAC-CE. Wait for at least 2 seconds.
- 4) SS transmits PDSCH via PDCCH DCI format 1A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.6.1A.3.4.1-1 on both PCC and SCC. The SS sends downlink MAC padding bits on the DL RMC.
- 5) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.6.1A.3.4.1-1 on PCC. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 6) Set the parameters of the signal generator for an interfering signal below the SCC in Case 1 according to tables 4.2.7.1.2-1 and 4.2.7.2.2-1, or 4.2.7.1.2-1 and 4.2.7.2.2-1 for operating bands without uplink band (as noted in table 1-1), or 4.2.7.2.2-1a and 4.2.7.2.2-1b for operating band 46 without uplink band.
- 7) Set the downlink signal level according to the table 4.2.7.1.2-1 or table 4.2.7.2.2-1a for operating band 46. Send uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.7.1.2-1 or table 4.2.7.2.2-1a for operating band 46 for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < $f \le 4,2$ GHz, for at least the duration of the throughput measurement.
- 8) Measure the average throughput of SCC for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].
- 9) Repeat steps from 6 to 8, using an interfering signal above the SCC in Case 1 at step 6).

10) Repeat steps from 6 to 9, using interfering signals in Case 2 at steps 6) and 9). The ranges of case 2 are covered in steps equal to the interferer bandwidth. The test frequencies are chosen in analogy to ETSI TS 136 521-1 [1], table 7.6.1.4.2-1.

115

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.6.1A.3.

5.3.6.2.1.3 Out-Of-Band Procedure

- 1) Configure SCC according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.2 for all downlink physical channels except PHICH.
- 2) The SS shall configure SCC as per ETSI TS 136 508 [2], clause 5.2A.4.
- 3) SS activates SCC by sending the activation MAC-CE. Wait for at least 2 seconds.
- 4) SS transmits PDSCH via PDCCH DCI format 1A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.6.2A.3.4.1-1 on both PCC and SCC. The SS sends downlink MAC padding bits on the DL RMC.
- 5) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.6.2A.3.4.1-1 on PCC. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 6) Set the parameters of the signal generator for an interfering signal below the SCC's operating band according to table 4.2.7.2.2-2 or table 4.2.7.2.2-2a for operating band 46. The frequency step size is 1 MHz.
- 7) Set the downlink signal level according to the table 4.2.7.1.2-3 for both carriers. Send uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.7.1.2-3 for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < f \le 4,2 GHz, for at least the duration of the throughput measurement.
- 8) Measure the average throughput of SCC for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].
- 9) Record the frequencies for which the throughput does not meet the requirements.
- 10) Repeat steps from 6 to 9, using an interfering signal above the SCC's operating band at step 6).

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.6.2A.3.

5.3.6.2.1.4 Narrow-Band Procedure

- 1) Configure SCC according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.1 for all downlink physical channels except PHICH.
- 2) The SS shall configure SCC as per ETSI TS 136 508 [2], clause 5.2A.4.
- 3) SS activates SCC by sending the activation MAC-CE. Wait for at least 2 seconds.
- 4) SS transmits PDSCH via PDCCH DCI format 1A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.6.3A.3.4.1-1 on both PCC and SCC. The SS sends downlink MAC padding bits on the DL RMC.
- 5) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.6.3A.3.4.1-1 on PCC. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 6) Set the parameters of the CW signal generator for an interfering signal below the SCC according to table 4.2.7.1.2-5.
- 7) Set the downlink signal level according to the table 4.2.7.1.2-5. Send uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.7.1.2-5 for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < $f \le 4,2$ GHz, for at least the duration of the throughput measurement.

8) Measure the average throughput of SCC for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].

116

9) Repeat steps from 6 to 8, using an interfering signal above the SCC at step 6).

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.6.3A.3.

5.3.6.2.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.7.2.2 in order to show compliance.

5.3.6.3 Receiver Blocking Characteristics for category NB1

5.3.6.3.1 Method of test

5.3.6.3.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

For both, in-band and out-of-band blocking, the frequencies to be tested are frequency ranges defined in ETSI TS 136 521-1 [1], clause K.1.1 as defined in ETSI TS 136 508 [2], clause 8.1.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connector using only the main UE Tx/Rx antenna.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 8.1.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.2 and NPUSCH Format 2 is used to carry ACK/NACK on the uplink.
- 4) The DL Reference Measurement channel is set according to ETSI TS 136 521-1 [1], table 7.6.1F.4.1-1 for in-band blocking and table 7.6.2F.4.1-1 for out-of-band blocking.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 2A-NB with CP CIoT Optimization according to ETSI TS 136 508 [2], clause 8.1.5.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.6.3.1.2 In-Band Procedure

- SS transmits NPDSCH via NPDCCH DCI format N1 for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.6.1F.4.1-1. The SS sends downlink MAC padding bits on the DL RMC. The UE will send HARQ feedback based on information contained in DCI format N1.
- 2) Set the downlink signal level according to the table 4.2.7.3.2-1.
- 3) Set the parameters of the signal generator for an interfering signal below the wanted signal in IBB1 according to table 4.2.7.3.2-1.
- 4) Measure the average throughput for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].
- 5) Repeat steps from 3 to 4, using an interfering signal above the wanted signal in IBB1 at step 3).
- 6) Repeat steps from 3 to 5, using interfering signals in IBB2 at steps 3) and 5). The ranges of IBB2 are covered in steps equal to the interferer bandwidth. The test frequencies are chosen in analogy to ETSI TS 136 521-1 [1], table 7.6.1F.4.2-1.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.6.1F.4.

5.3.6.3.1.3 Out-Of-Band Procedure

- SS transmits NPDSCH via NPDCCH DCI format N1 for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.6.2F.4.1-1. The SS sends downlink MAC padding bits on the DL RMC. The UE will send HARQ feedback based on information contained in DCI format N1.
- 2) Set the downlink signal level according to the table 4.2.7.3.2-2.
- 3) Set the parameters of the CW signal generator for an interfering signal below the wanted signal according to table 4.2.7.3.2-2. The frequency step size is 1 MHz.
- 4) Measure the average throughput for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].
- 5) Record the frequencies for which the throughput does not meet the requirements.
- 6) Repeat steps from 3 to 5, using an interfering signal above the wanted signal at step 3).

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.6.2F.4.

5.3.6.3.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.7.3.2 in order to show compliance.

5.3.6.4 Receiver Blocking Characteristics for UE category 0

5.3.6.4.1 Method of test

5.3.6.4.1.1 Initial conditions

Same initial conditions as in clause 5.3.6.1.1.1 with the following exceptions:

- Instead of table 7.6.1.4.1-1 in ETSI TS 136 521-1 [1] → use table 7.6.1E.4.1-1 in ETSI TS 136 521-1 [1], or instead of table 7.6.2.4.1-1 in ETSI TS 136 521-1 [1] → use table 7.6.2E.4.1-1 in ETSI TS 136 521-1 [1], or instead of table 7.6.3.4.1-1 in ETSI TS 136 521-1 [1] → use table 7.6.3E.4.1-1 in ETSI TS 136 521-1 [1], as applicable.
- Connect the SS to the UE antenna connectors as shown in ETSI TS 136 508 [2], annex A using only main UE Tx/Rx antenna.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.6.4.1.2 In-Band Procedure

Same test procedure as in clause 5.3.6.1.1.2.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.6.1E.

5.3.6.4.1.3 Out-Of-Band Procedure

Same test procedure as in clause 5.3.6.1.1.3.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.6.2E.

5.3.6.4.1.4 Narrow-Band Procedure

Same test procedure as in clause 5.3.6.1.1.4.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.6.3E.

5.3.6.4.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.7.1.2 in order to show compliance.

118

5.3.6.5 Receiver Blocking Characteristics for UE category M1

5.3.6.5.1 Method of test

5.3.6.5.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

Frequencies to be tested:

- for In-band blocking: mid-range as defined in ETSI TS 136 508 [2], clause 4.3.1; or
- for out-of-band blocking: low or high range as defined in ETSI TS 136 508 [2], clause 4.3.1;
- for Narrow-band blocking: mid-range as defined in ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidth to be tested: 5 MHz channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors as shown in ETSI TS 136 508 [2], annex A using only main UE Tx/Rx antenna.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.1 and uplink signals according to clauses H.1 and H.3.1.
- 4) The UL and DL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], table 7.6.1EA.4.1-1, 7.6.2EA.4.1-1 or 7.6.3EA.4.1-1, as applicable.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF-CE according to ETSI TS 136 508 [2], clause 5.2A.2AA.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.6.5.1.2 In-Band Procedure

- SS transmits PDSCH via MPDCCH DCI format 6-1A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.6.1EA.4.1-1. The SS sends downlink MAC padding bits on the DL RMC. The SS sends one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in ETSI TS 136 521-1 [1], clauses A.5.1.1 and A.5.2.1.
- 2) SS sends uplink scheduling information for each UL HARQ process via MPDCCH DCI format 6-0A for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.6.1EA.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 3) Set the parameters of the signal generator for an interfering signal below the wanted signal in Case 1 according to tables 4.2.7.1.2-1 and 4.2.7.1.2-2.
- 4) Set the downlink signal level according to the table 4.2.7.1.2-1. Send uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.7.1.2-1 for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < $f \le 4,2$ GHz, for at least the duration of the throughput measurement as specified in ETSI TS 136 521-1 [1].
- 5) Measure the average throughput for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].

- 6) Repeat steps from 3 to 5, using an interfering signal above the wanted signal in Case 1 at step 3).
- 7) Repeat steps from 3 to 6, using interfering signals in Case 2 at steps 3) and 6). The ranges of case 2 are covered in steps equal to the interferer bandwidth. The test frequencies are chosen in analogy to ETSI TS 136 521-1 [1], table 7.6.1EA.4.2-1.
- 8) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.6.1EA.

5.3.6.5.1.3 Out-Of-Band Procedure

- SS transmits PDSCH via MPDCCH DCI format 6-1A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.6.2EA.4.1-1. The SS sends downlink MAC padding bits on the DL RMC. The SS sends one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in ETSI TS 136 521-1 [1], clauses A.5.1.1 and A.5.2.1.
- 2) SS sends uplink scheduling information for each UL HARQ process via MPDCCH DCI format 6-0A for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.6.2EA.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 3) Set the parameters of the CW signal generator for an interfering signal according to table 4.2.7.1.2-4. The frequency step size is 1 MHz.
- 4) Set the downlink signal level according to the table 4.2.7.1.2-3. Send uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.7.1.2-3 for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < $f \le 4,2$ GHz, for at least the duration of the throughput measurement as specified in ETSI TS 136 521-1 [1].
- 5) Measure the average throughput for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].
- 6) For table 4.2.7.1.2-4 record the frequencies for which the throughput does not meet the requirements.
- 7) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.6.2EA.

5.3.6.5.1.4 Narrow-Band Procedure

- SS transmits PDSCH via MPDCCH DCI format 6-1A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.6.1EA.4.1-1. The SS sends downlink MAC padding bits on the DL RMC. The SS sends one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in ETSI TS 136 521-1 [1], clauses A.5.1.1 and A.5.2.1.
- 2) SS sends uplink scheduling information for each UL HARQ process via MPDCCH DCI format 6-0A for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.6.3EA.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 3) Set the parameters of the CW signal generator for an interfering signal below the wanted signal according to table 4.2.7.1.2-5.
- 4) Set the downlink signal level according to the table 4.2.7.1.2-5. Send uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.7.1.2-5 for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < $f \le 4,2$ GHz, for at least the duration of the throughput measurement as specified in ETSI TS 136 521-1 [1].
- 5) Measure the average throughput for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].
- 6) Repeat steps from 3 to 5, using an interfering signal above the wanted signal at step 3).
- 7) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.6.3EA.

5.3.6.5.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.7.1.2 in order to show compliance.

5.3.7 Receiver Spurious Response

5.3.7.1 Receiver Spurious Response for Single Carrier

5.3.7.1.1 Method of test

5.3.7.1.1.1 Initial conditions

The initial conditions shall be the same as for those in out-of-band blocking in clause 5.3.6.1.1 in order to test spurious responses obtained in clause 5.3.6.1.2 under the same conditions.

5.3.7.1.1.2 Procedure

- 1) SS transmits PDSCH via PDCCH DCI format 1A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.6.2.4.1-1. The SS sends downlink MAC padding bits on the DL RMC.
- 2) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.6.2.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 3) Set the parameters of the CW signal generator for an interfering signal according to table 4.2.8.1.2-2. The spurious frequencies are taken from step 5) records in clause 5.3.6.1.1.2.
- 4) Set the downlink signal level according to the table 4.2.8.1.2-1. Send uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.8.1.2-1 for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < $f \le 4,2$ GHz, for at least the duration of the throughput measurement as specified in ETSI TS 136 521-1 [1].
- 5) For the spurious frequency, measure the average throughput for a duration sufficient to achieve statistical significance.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.7.

5.3.7.1.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.8.1.2 in order to show compliance.

5.3.7.2 Receiver Spurious Response for Carrier Aggregation in DL-only bands

- 5.3.7.2.1 Method of test
- 5.3.7.2.1.1 Initial conditions

The initial conditions shall be the same as for those in out-of-band blocking in clause 5.3.6.2.1.1 in order to test spurious responses obtained in clause 5.3.6.2.1.3 under the same conditions.

5.3.7.2.1.2 Procedure

- 1) Configure SCC according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.2 for all downlink physical channels except PHICH.
- 2) The SS shall configure SCC as per ETSI TS 136 508 [2], clause 5.2A.4.

- 3) SS activates SCC by sending the activation MAC-CE. Wait for at least 2 seconds.
- 4) SS transmits PDSCH via PDCCH DCI format 1A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.6.2A.3.4.1-1 on both PCC and SCC. The SS sends downlink MAC padding bits on the DL RMC.

121

- 5) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.6.2A.3.4.1-1 on PCC. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 6) Set the parameters of the CW signal generator for an interfering signal according to table 4.2.8.1.2-2. The spurious frequencies are taken from step 5) records in clause 5.3.6.2.1.3.
- 7) Set the downlink signal level according to the table 4.2.8.1.2-1 for both carriers. Send uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.8.1.2-1 for carrier frequency $f \le 3.0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < f \le 4,2 GHz, for at least the duration of the throughput measurement.
- 8) For the spurious frequency, measure the average throughput on SCC for a duration sufficient to achieve statistical significance.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.7A.3.

5.3.7.2.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.8.1.2 in order to show compliance.

5.3.7.3 Receiver Spurious Response for category NB1

5.3.7.3.1 Method of test

5.3.7.3.1.1 Initial conditions

The initial conditions shall be the same as for those in out-of-band blocking in clause 5.3.6.3.1 in order to test spurious responses obtained in clause 5.3.6.3.2 under the same conditions.

5.3.7.3.1.2 Procedure

- SS transmits NPDSCH via NPDCCH DCI format N1 for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.6.2F.4.1-1. The SS sends downlink MAC padding bits on the DL RMC. The UE will send HARQ feedback based on information contained in DCI format N1.
- 2) Set the parameters of the CW signal generator for an interfering signal according to table 4.2.8.3.2-1. The spurious frequencies are taken from records in the final step of test procedures in clause 5.3.6.3.1.2.
- 3) Set the downlink signal level according to the table 4.2.8.3.2-1.
- 4) For the spurious frequency, measure the average throughput for a duration sufficient to achieve statistical significance according to ETSI TS 136 521-1 [1], clause G.2.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.7F.4.

5.3.7.3.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.8.3.2 in order to show compliance.

5.3.7.4 Receiver Spurious Response for UE category 0

5.3.7.4.1 Method of test

5.3.7.4.1.1 Initial conditions

The initial conditions shall be the same as for those in out-of-band blocking in clause 5.3.6.4.1 in order to test spurious responses obtained in clause 5.3.6.4.2 under the same conditions.

5.3.7.4.1.2 Procedure

Same test procedure as in clause 5.3.7.1.1.2.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.7E.

5.3.7.4.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.8.1.2 in order to show compliance.

5.3.7.5 Receiver Spurious Response for UE category M1

5.3.7.5.1 Method of test

5.3.7.5.1.1 Initial conditions

The initial conditions shall be the same as for those in out-of-band blocking in clause 5.3.6.5.1 in order to test spurious responses obtained in clause 5.3.6.5.2 under the same conditions.

5.3.7.5.1.2 Procedure

- SS transmits PDSCH via MPDCCH DCI format 6-1A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.6.2E.4.1-1. The SS sends downlink MAC padding bits on the DL RMC. The SS sends one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in ETSI TS 136 521-1 [1], clauses A.5.1.1 and A.5.2.1.
- 2) SS sends uplink scheduling information for each UL HARQ process via MPDCCH DCI format 6-0A for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.6.2E.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 3) Set the parameters of the CW signal generator for an interfering signal according to table 4.2.8.1.2-2. The spurious frequencies are taken from step 5) records in clause 5.3.6.5.1.2.
- 4) Set the downlink signal level according to the table 4.2.8.1.2-1. Send uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.8.1.2-1 for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < $f \le 4,2$ GHz, for at least the duration of the throughput measurement as specified in ETSI TS 136 521-1 [1].
- 5) For the spurious frequency, measure the average throughput for a duration sufficient to achieve statistical significance.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.7EA.

5.3.7.5.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.8.1.2 in order to show compliance.

5.3.8 Receiver Intermodulation Characteristics

5.3.8.1 Receiver Intermodulation Characteristics for Single Carrier

5.3.8.1.1 Method of test

5.3.8.1.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

Frequencies to be tested: mid-range; see ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidths to be tested: lowest, 5 MHz and highest channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

123

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS and interfering sources to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1, C.3.1 and uplink signals according to clauses H.1 and H.3.1.
- 4) The UL and DL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], table 7.8.4.1-1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.8.1.1.2 Procedure

- 1) SS transmits PDSCH via PDCCH DCI format 1A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.8.1.4.1-1. The SS sends downlink MAC padding bits on the DL RMC.
- 2) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.8.1.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 3) Set the Downlink signal level to the value as defined in table 4.2.9.1.2-1. Send uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.9.1.2-1 for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < $f \le 4,2$ GHz, for at least the duration of the throughput measurement as specified in ETSI TS 136 521-1 [1].
- 4) Set the Interfering signal levels to the values as defined in table 4.2.9.1.2-1, using a modulated interferer bandwidth as defined in annex D of ETSI TS 136 521-1 [1].
- 5) Measure the average throughput for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].
- 6) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.8.

5.3.8.1.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.9.1.2 in order to show compliance.

5.3.8.2 Receiver Intermodulation Characteristics for Carrier Aggregation in DL-only bands

124

5.3.8.2.1 Method of test

5.3.8.2.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

Frequencies to be tested: mid-range; see ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidths to be tested: Highest N_{RB} agg for PCC and SCC.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS and interfering sources to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1, C.3.1 and uplink signals according to clauses H.1 and H.3.1.
- 4) The UL and DL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], table 7.8.1A.3.4.1-1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.8.2.1.2 Procedure

- 1) Configure SCC according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.1 for all downlink physical channels except PHICH.
- 2) The SS shall configure SCC as per ETSI TS 136 508 [2], clause 5.2A.4.
- 3) SS activates SCC by sending the activation MAC-CE. Wait for at least 2 seconds.
- SS transmits PDSCH via PDCCH DCI format 2A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.8.1A.3.4.1-1 on both PCC and SCC. The SS sends downlink MAC padding bits on the DL RMC.
- 5) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.8.1A.3.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 6) Set the Downlink signal level to the value as defined in table 4.2.9.1.2-1 or table 4.2.9.2.2-1 for operating band 46. Send Uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.9.1.2-1 or table 4.2.9.2.2-1 for operating band 46 for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < $f \le 4,2$ GHz, for at least the duration of the Throughput measurement.
- 7) Set the Interfering signal levels to the values as defined in table 4.2.9.1.2-1 or table 4.2.9.2.2-1 for operating band 46, and frequency below the wanted signal, using a modulated interferer bandwidth as defined in annex D of ETSI TS 136 521-1 [1].
- 8) Measure the average throughput of SCC for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].
- 9) Repeat steps from 6 to 8, using an interfering signal above the wanted signal at step 4).

5.3.8.2.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.9.2.2 in order to show compliance.

5.3.8.3 Receiver Intermodulation Characteristics for category NB1

5.3.8.3.1 Test requirements

5.3.8.3.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

Frequencies to be tested: Frequency ranges defined in ETSI TS 136 521-1 [1], clause K.1.1, see ETSI TS 136 508 [2], clause 8.1.3.1.

125

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors using only the main UE Tx/Rx antenna.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 8.1.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1, C.2 and NPUSCH Format 2 is used to carry ACK/NACK on the uplink.
- 4) The DL Reference Measurement channel is set according to ETSI TS 136 521-1 [1], table 7.8.1F.4.1-1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 2A-NB with CP CIoT Optimization according to ETSI TS 136 508 [2], clause 8.1.5.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.8.3.1.2 Procedure

- SS transmits NPDSCH via NPDCCH DCI format N1 for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.8.1F.4.1-1. The SS sends downlink MAC padding bits on the DL RMC. The UE will send HARQ feedback based on information contained in DCI format N1.
- 2) Set the Downlink signal level to the value as defined in table 4.2.9.3.2-1.
- 3) Set the Interfering signal levels to the values as defined in table 4.2.9.3.2-1 and frequency below the wanted signal, using a modulated interferer bandwidth as defined in annex D of ETSI TS 136 521-1 [1].
- 4) Measure the average throughput for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].
- 5) Repeat steps from 2 to 4, using an interfering signal above the wanted signal in step 3).

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.8.1F.4.

5.3.8.3.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.9.3.2 in order to show compliance.

5.3.8.4 Receiver Intermodulation Characteristics for UE category 0

5.3.8.4.1 Method of test

5.3.8.4.1.1 Initial conditions

Same initial conditions as in clause 5.3.8.1.1.1 with the following exceptions:

- Instead of table 7.8.1.4.1-1 in ETSI TS 136 521-1 [1] → use table 7.8.1E.4.1-1 in ETSI TS 136 521-1 [1].
- Connect the SS to the UE antenna connectors as shown in ETSI TS 136 508 [2], annex A, figure A.6 using only main UE Tx/Rx antenna.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.8.4.1.2 Procedure

Same test procedure as in clause 5.3.8.1.1.2.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.8.1E.

5.3.8.4.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.9.1.2 in order to show compliance.

5.3.8.5 Receiver Intermodulation Characteristics for UE category M1

- 5.3.8.5.1 Method of test
- 5.3.8.5.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

Frequencies to be tested: mid-range; see ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidths to be tested:5 MHz channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

- 1) Connect the SS to the UE antenna connectors as shown in ETSI TS 136 508 [2], annex A, figure A.6 using only main UE Tx/Rx antenna.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1, C.3.1 and uplink signals according to clauses H.1 and H.3.1.
- 4) The UL and DL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], table 7.8.1EA.4.1-1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF-CE according to ETSI TS 136 508 [2], clause 5.2A.2AA.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.8.5.1.2 Procedure

- 1) SS transmits PDSCH via MPDCCH DCI format 6-1A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.8.1EA.4.1-1. The SS sends downlink MAC padding bits on the DL RMC.
- 2) SS sends uplink scheduling information for each UL HARQ process via MPDCCH DCI format 6-0A for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.8.1EA.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 3) Set the Downlink signal level to the value as defined in table 4.2.9.1.2-1. Send uplink power control commands to the UE (less or equal to 1 dB step size should be used), to ensure that the UE output power is within +0, -3,4 dB of the target level in table 4.2.9.1.2-1 for carrier frequency $f \le 3,0$ GHz or within +0, -4,0 dB of the target level for carrier frequency 3,0 GHz < $f \le 4,2$ GHz, for at least the duration of the throughput measurement as specified in ETSI TS 136 521-1 [1].
- 4) Set the Interfering signal levels to the values as defined in table 4.2.9.1.2-1, using a modulated interferer bandwidth as defined in annex D of ETSI TS 136 521-1 [1].
- 5) Measure the average throughput for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].
- 6) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.8.1EA.

5.3.8.5.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.9.1.2 in order to show compliance.

5.3.9 Receiver Spurious Emissions

- 5.3.9.1 Receiver Spurious Emissions for Single Carrier
- 5.3.9.1.1 Method of test
- 5.3.9.1.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

Frequencies to be tested: low range, mid-range and high range; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidth to be tested: highest channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

- 1) Connect a spectrum analyser (or other suitable test equipment) to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.1.
- 4) The DL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 7.9.4.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.9.1.1.2 Procedure

1) Sweep the spectrum analyser (or other suitable test equipment) over a frequency range from 30 MHz to 12,75 GHz and measure the average power of the spurious emissions.

128

- 2) Repeat step 1) for all E-UTRA Rx antennas of the UE.
- 3) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.9.

5.3.9.1.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.10.1.2 in order to show compliance.

5.3.9.2 Receiver Spurious Emissions in DL-only bands

5.3.9.2.1 Method of test

5.3.9.2.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

Frequencies to be tested: mid-range; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidth to be tested: Highest $N_{RB_{agg}}$ for PCC and SCC as defined in ETSI TS 136 508 [2], clause 4.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect a spectrum analyser (or other suitable test equipment) to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals for PCC are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.1.
- 4) The DL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 7.9A.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.9.2.1.2 Procedure

- 1) Configure SCC according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.1 for all downlink physical channels except PHICH.
- 2) The SS shall configure SCC as per ETSI TS 136 508 [2], clause 5.2A.4.
- 3) SS activates SCC by sending the activation MAC-CE. Wait for at least 2 seconds.
- SS transmits PDSCH via PDCCH DCI format 2A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.9A.4.1-1 on both PCC and SCC. The SS sends downlink MAC padding bits on the DL RMC.
- 5) Sweep the spectrum analyser (or equivalent equipment) over a frequency range and measure the average power of spurious emission. During measurement, SS sends no uplink scheduling information to the UE.
- 6) Repeat steps 1) to 5) for all E-UTRA DL-only band Rx antennas of the UE.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.9A.

5.3.9.2.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.10.2.2 in order to show compliance.

5.3.9.3 Receiver Spurious Emissions for UE category 0

5.3.9.3.1 Method of test

5.3.9.3.1.1 Initial conditions

Same initial conditions as in clause 5.3.9.1.1.1 with the following exceptions:

- Instead of clause 7.9.4.1 in ETSI TS 136 521-1 [1] → use clause 7.9E.4.1 in ETSI TS 136 521-1 [1].
- Connect the SS to the UE antenna connectors as shown in ETSI TS 136 508 [2] annex A, figure A.8 using only main UE Tx/Rx antenna.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.9.3.1.2 Procedure

Same test procedure as in clause 5.3.9.1.1.2.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.9E.

5.3.9.3.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.10.1.2 in order to show compliance.

5.3.9.4 Receiver Spurious Emissions for UE category M1

5.3.9.4.1 Method of test

5.3.9.4.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

Frequencies to be tested: low range, mid-range and high range; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidth to be tested: 5 MHz channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

- Connect a spectrum analyser (or other suitable test equipment) to the UE antenna connectors as shown in ETSI TS 136 508 [2], annex A, figure A.7 using only main UE Tx/Rx antenna.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.1.
- 4) The DL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 7.9EA.4.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF-CE according to ETSI TS 136 508 [2], clause 5.2A.2AA.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.9.4.1.2 Procedure

Same test procedure as in clause 5.3.9.1.1.2.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.9EA.

5.3.9.4.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.10.1.2 in order to show compliance.

130

5.3.9.5 Receiver Spurious Emissions for UE category NB1

5.3.9.5.1 Method of test

5.3.9.5.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

Frequencies to be tested: low range, mid-range and high range; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect a spectrum analyser (or other suitable test equipment) to the UE antenna connectors as shown in ETSI TS 136 508 [2], annex A, figure A.8 using only main UE Tx/Rx antenna.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.1.
- 4) The DL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 7.9F.4.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 2A-NB with CP CIoT Optimization according to ETSI TS 136 508 [2], clause 8.1.5. Message contents are defined in clause 7.9F.4.3.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.9.5.1.2 Procedure

Same test procedure as in clause 5.3.9.1.1.2.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.9F.

5.3.9.5.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.10.1.2 in order to show compliance.

5.3.10 Transmitter Adjacent Channel Leakage Power Ratio

5.3.10.1 Transmitter adjacent channel leakage power ratio for Single Carrier

5.3.10.1.1 Method of test

5.3.10.1.1.1 Initial conditions

Test Environment: normal, TL/VL, TL/VH, TH/VL and TH/VH, as specified in annex B.

Frequencies to be tested: low range, mid-range and high range; see ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidth to be tested: lowest, 5 MHz, 10 MHz and highest channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1, C.3.0 and uplink signals according to clauses H.1 and H.3.0.

131

- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1].
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.10.1.1.2 Procedure

- 1) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 6.6.2.3.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 2) Send continuous uplink power control "up" commands in the uplink scheduling information to the UE to ensure that the UE transmits at P_{UMAX} level.
- 3) Measure the mean power of the UE in the channel bandwidth of the radio access mode according to the test configuration, which shall meet the requirements described in tables 4.2.11.1.2-1 and 4.2.11.1.2-2. The period of the measurement shall be at least the continuous duration of one sub-frame (1 ms). For TDD slots with transient periods are not under test.
- 4) Measure the filtered mean power for E-UTRA.
- 5) Measure the filtered mean power of the first E-UTRA adjacent channel.
- 6) Measure the RRC filtered mean power of the first and the second UTRA adjacent channel.
- 7) Calculate the ratio of the power between the values measured in step 4) over step 5) for E-UTRA_{ACLR}.
- Calculate the ratio of the power between the values measured in step 4) over step 6) for UTRA_{ACLR1}, UTRA_{ACLR2}.
- 9) Repeat for applicable test frequencies, channel bandwidths, operating bands and environmental conditions.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.6.2.3.

5.3.10.1.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.11.1.2 in order to show compliance.

5.3.10.2 Transmitter adjacent channel leakage power ratio for intra-band contiguous Carrier Aggregation (DL CA and UL CA)

132

- 5.3.10.2.1 Method of test
- 5.3.10.2.1.1 Initial conditions

Test environment: normal, TL/VL, TL/VH, TH/VL and TH/VH, as specified in annex B.

Frequencies to be tested: low range, high range; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Test CC Combination setting (N_{RB_agg}): lowest N_{RB_agg} , highest N_{RB_agg} , as specified in ETSI TS 136 521-1 [1], clause 5.4.2A.1 for the CA Configuration.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals for PCC are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1, C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.6.2.3A.1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.10.2.1.2 Procedure

- 1) Configure SCC according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 for all downlink physical channels, except PHICH.
- 2) The SS shall configure SCC as per ETSI TS 136 508 [2], clause 5.2A.4.
- 3) SS activates SCC by sending the activation MAC-CE. Wait for at least 2 seconds.
- 4) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to table 6.6.2.3A.1.4.1-1 of ETSI TS 136 521-1 [1] on both PCC and SCC. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.
- 5) Send continuously uplink power control "up" commands in every uplink scheduling information to the UE; allow at least 200 ms for the UE to reach P_{UMAX} level.
- 6) Measure the mean power over all component carriers of the UE in the CA configuration of the radio access mode according to the test configuration, which shall meet the requirements described in tables 4.2.11.2.2-1 and 4.2.11.2.2-2. The period of the measurement shall be at least the continuous duration of one sub-frame (1 ms). For TDD slots with transient periods are not under test.
- 7) Measure the rectangular filtered mean power for CA E-UTRA.
- 8) Measure the rectangular filtered mean power of the first CA E-UTRA adjacent channel on both lower and upper side of the CA E-UTRA channel, respectively.
- 9) Measure the RRC filtered mean power of the first and the second UTRA for CA adjacent channel on both lower and upper side of the CA E-UTRA channel, respectively.
- 10) Calculate the ratio of the power between the values measured in step 7) over step 8) for CA E-UTRA_{ACLR}.

- Calculate the ratio of the power between the values measured in step 7) over step 9) for UTRA_{ACLR1}, UTRA_{ACLR2}.
- 12) Repeat for applicable test frequencies, channel bandwidths, operating bands and environmental conditions.

133

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.6.2.3A.1.

5.3.10.2.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.11.2.2 in order to show compliance.

5.3.10.2A Transmitter adjacent channel leakage power ratio for inter-band Carrier Aggregation (DL CA and UL CA)

- 5.3.10.2A.1 Method of test
- 5.3.10.2A.1.1 Initial conditions

Test environment: normal, TL/VL, TL/VH, TH/VL and TH/VH, as specified in annex B.

Frequencies to be tested: low range for PCC and SCC, high range for PCC and SCC; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Test CC Combination setting (N_{RB_agg}): lowest N_{RB_agg} , highest N_{RB_agg} , as specified in ETSI TS 136 521-1 [1], clause 5.4.2A.1 for the CA Configuration.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals for PCC are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1, C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.6.2.3A.2.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.10.2A.1.2 Procedure

- 1) Configure SCC according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 for all downlink physical channels, except PHICH.
- 2) The SS shall configure SCC as per ETSI TS 136 508 [2], clause 5.2A.4.
- 3) SS activates SCC by sending the activation MAC-CE. Wait for at least 2 seconds.
- 4) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to table 6.6.2.3A.2.4.1-1 of ETSI TS 136 521-1 [1] on both PCC and SCC. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.
- 5) Send continuously uplink power control "up" commands in every uplink scheduling information to the UE; allow at least 200 ms for the UE to reach P_{UMAX} level.

6) Measure the mean power over all component carriers of the UE in the CA configuration of the radio access mode according to the test configuration, which shall meet the requirements described in table 4.2.2.2.2.2. The period of the measurement shall be at least the continuous duration of one sub-frame (1 ms). For TDD slots with transient periods are not under test.

134

- 7) Measure the rectangular filtered mean power for CA E-UTRA in the first band of the CA configuration.
- 8) Measure the rectangular filtered mean power of the first CA E-UTRA adjacent channel on both lower and upper side of the CA E-UTRA channel in the first band of the CA configuration, respectively.
- 9) Measure the RRC filtered mean power of the first and the second UTRA adjacent channel on both lower and upper side of the CA E-UTRA channel in the first band of the CA configuration, respectively.
- 10) Calculate the ratio of the power between the values measured in step 7) over step 8) for CA E-UTRA_{ACLR}.
- Calculate the ratio of the power between the values measured in step 7) over step 9) for UTRA_{ACLR1}, UTRA_{ACLR2}.
- 12) Repeat steps 6) to 11) for the second band of the CA configuration.
- 13) Repeat for applicable test frequencies, channel bandwidths, operating bands and environmental conditions.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.6.2.3A.2.

5.3.10.2A.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.11.1.2 in order to show compliance.

5.3.10.3 Transmitter adjacent channel leakage power ratio for UL-MIMO

- 5.3.10.3.1 Method of test
- 5.3.10.3.1.1 Initial conditions

Test environment: normal, TL/VL, TL/VH, TH/VL, TH/VH; as specified in annex B.

Frequencies to be tested: low range, mid-range, high range; as specified in ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidths to be tested: lowest, 5 MHz, 10 MHz and highest channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1, C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.6.2.3B.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.10.3.1.2 Procedure

- 1) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to table 6.6.2.3B.1.4.1-1 of ETSI TS 136 521-1 [1]. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.
- 2) Send continuously uplink power control "up" commands in every uplink scheduling information to the UE; allow at least 200 ms for the UE to reach P_{UMAX} level.
- 3) Measure the sum of the mean power at each antenna connector of UE in the channel bandwidth of the radio access mode according to the test configuration, which shall meet the requirements described in tables 4.2.11.3.2-1 and 4.2.11.3.2-2. The period of the measurement shall be at least the continuous duration of one sub-frame (1 ms). For TDD slots with transient periods are not under test.
- 4) Measure the rectangular filtered mean power for E-UTRA at each antenna connector of UE.
- 5) Measure the rectangular filtered mean power of the first E-UTRA adjacent channel at each antenna connector of UE.
- 6) Measure the RRC filtered mean power of the first and the second UTRA adjacent channel at each antenna connector of UE.
- 7) Calculate the ratio of the power between the values measured in step 4) over step 5) for E-UTRA_{ACLR}.
- 8) Calculate the ratio of the power between the values measured in step 4) over step 6) for UTRA_{ACLR1}, UTRA_{ACLR2}.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.6.2.3B.

5.3.10.3.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.11.3.2 in order to show compliance.

5.3.10.4 Transmitter adjacent channel leakage power ratio for Multi-Cluster PUSCH within a component carrier

5.3.10.4.1 Method of test

5.3.10.4.1.1 Initial conditions

Test Environment: normal, TL/VL, TL/VH, TH/VL and TH/VH, as specified in annex B.

Frequencies to be tested: low range, mid-range and high range; see ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidth to be tested: Highest channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

- 1) Connect the SS to the UE to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1, C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.6.2.3_2.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.10.4.1.2 Procedure

- 1) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 6.6.2.3_2.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 2) Send continuous uplink power control "up" commands in the uplink scheduling information to the UE to ensure that the UE transmits at P_{UMAX} level.
- 3) Measure the mean power of the UE in the channel bandwidth of the radio access mode according to the test configuration, which shall meet the requirements described in tables 4.2.11.1.2-1 and 4.2.11.1.2-2. The period of the measurement shall be at least the continuous duration of one sub-frame (1 ms). For TDD slots with transient periods are not under test.
- 4) Measure the filtered mean power for E-UTRA.
- 5) Measure the filtered mean power of the first E-UTRA adjacent channel.
- 6) Measure the RRC filtered mean power of the first and the second UTRA adjacent channel.
- 7) Calculate the ratio of the power between the values measured in step 4) over step 5) for E-UTRA_{ACLR}.
- Calculate the ratio of the power between the values measured in step 4) over step 6) for UTRA_{ACLR1}, UTRA_{ACLR2}.
- 9) Repeat for applicable test frequencies, channel bandwidths, operating bands and environmental conditions.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.6.2.3.2.

5.3.10.4.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.11.4.2 in order to show compliance.

5.3.10.5 Transmitter adjacent channel leakage power ratio for category NB1

5.3.10.5.1 Method of test

5.3.10.5.1.1 Initial conditions

Test Environment: normal, as specified in annex B.

Frequencies to be tested: frequency ranges defined in ETSI TS 136 521-1 [1], clause K.1.2, see ETSI TS 136 508 [2], clause 8.1.3.1.

- 1) Connect the SS to the UE to the UE antenna connectors using only main Tx/Rx antenna.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 8.1.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1, C.3.0 and uplink signals according to clauses H.1.1 and H.4.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], clause 6.6.2.3F.4.1-1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 2A-NB with CP CIoT Optimization according to ETSI TS 136 508 [2], clause 8.1.5.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.10.5.1.2 Procedure

- SS sends uplink scheduling information for the UL HARQ process via NPDCCH DCI format N0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 6.6.2.3F.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC (UE should be ready transmitting P_{UMAX} after Initial Conditions setting).
- 2) Measure the mean power of the UE in the channel bandwidth of the radio access mode according to the test configuration, which shall meet the requirements described in table 4.2.11.5.2-1. The measurement period is at least one sub-frame (1 ms) for 15 kHz channel spacing, and at least a 2 ms slot (excluding the 2 304Ts gap when UE is not transmitting) respectively for the 3,75 kHz channel spacing.
- 3) Measure the rectangular filtered mean power for category NB1 UE channel.
- 4) Measure the rectangular filtered mean power of the GSM adjacent channel on both lower and upper side of the category NB1 UE channel, respectively.
- 5) Measure the RRC filtered mean power of UTRA adjacent channel on both lower and upper side of the category NB1 UE channel, respectively.
- 6) Calculate the ratios of the power between the value measured in step 3) over step 4) for lower and upper GSM_{ACLR}.
- Calculate the ratio of the power between the value measured in step 3) over step 5) for lower and upper UTRA_{ACLR}.
- NOTE: For configuration IDs applicable to UE depending on UE capability in Test Configuration Table with different UL sub-carrier spacing, the SS releases the connection through State 3A-NB and finally ensures that the UE is in State 2A-NB with CP CIoT Optimization according to ETSI TS 136 508 [2], clause 8.1.5 using the appropriate UL subcarrier spacing in Random Access Response message.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.6.2.3F.

5.3.10.5.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.11.5.2 in order to show compliance.

5.3.10.6 Transmitter adjacent channel leakage power ratio for UE category 0

5.3.10.6.1 Method of test

5.3.10.6.1.1 Initial conditions

Same initial conditions as in clause 5.3.10.1.1.1 with following exceptions:

- Connect SS to the UE antenna connectors as shown in ETSI TS 136 508 [2], annex A, figure A.3 using only main UE Tx/Rx antenna.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.10.6.1.2 Procedure

Same test procedure as in clause 5.3.10.1.1.2 with following exception for HD-FDD:

- In step 3), slots with transient periods are not under test. Half-duplex guard subframes are not under test.
- Instead of table 6.6.2.3.4.1-1 in ETSI TS 136 521-1 [1] → use table 6.6.2.3E.4.1-1 in ETSI TS 136 521-1 [1].

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.6.2.3E.

5.3.10.6.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.11.1.2 in order to show compliance.

5.3.10.7 Transmitter adjacent channel leakage power ratio for UE category M1

138

5.3.10.7.1 Method of test

5.3.10.7.1.1 Initial conditions

Test Environment: normal, TL/VL, TL/VH, TH/VL and TH/VH, as specified in annex B.

Frequencies to be tested: low range, mid-range and high range; see ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidth to be tested: lowest, 5 MHz, 10 MHz and highest channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect SS to the UE antenna connectors as shown in ETSI TS 136 508 [2], annex A, figure A.3 using only main UE Tx/Rx antenna.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1, C.3.0 and uplink signals according to clauses H.1 and H.3.0.
- 4) The UL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], table 6.6.2.3EA.4.1-1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF-CE according to ETSI TS 136 508 [2], clause 5.2A.2AA.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.10.7.1.2 Procedure

- SS sends uplink scheduling information for each UL HARQ process via MPDCCH DCI format 6-0A for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 6.6.2.3EA.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 2) Send continuously uplink power control "up" commands in the uplink scheduling information to the UE until the UE transmits at P_{UMAX} level.
- 3) Measure the mean power of the UE in the channel bandwidth of the radio access mode according to the test configuration, which shall meet the requirements described in tables 4.2.11.1.2-1 and 4.2.11.1.2-2. The period of the measurement shall be at least the continuous duration of one sub-frame (1 ms). For TDD slots with transient periods are not under test. For HD-FDD slots with transient periods and Half-duplex guard subframe are not under test.
- 4) Measure the filtered mean power for E-UTRA.
- 5) Measure the filtered mean power of the first E-UTRA adjacent channel.
- 6) Measure the RRC filtered mean power of the first and the second UTRA adjacent channel.
- 7) Calculate the ratio of the power between the values measured in step 4) over step 5) for E-UTRA_{ACLR}.
- Calculate the ratio of the power between the values measured in step 4) over step 6) for UTRA_{ACLR1}, UTRA_{ACLR2}.
- 9) Repeat for applicable test frequencies, channel bandwidths, operating bands and environmental conditions.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 6.6.2.3EA.

5.3.10.7.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.11.1.2 in order to show compliance.

139

5.3.11 Receiver Reference Sensitivity Level

5.3.11.1 Receiver Reference Sensitivity Level for Single Carrier

5.3.11.1.1 Method of test

5.3.11.1.1.1 Initial conditions

Test Environment: normal, TL/VL, TL/VH, TH/VL, TH/VH; as specified in annex B.

Frequencies to be tested: low range, mid-range, high range see ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidth to be tested: lowest, 5 MHz and highest channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1, C.3.1 and uplink signals according to clauses H.1 and H.3.1.
- 4) The UL and DL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], table 7.3.4.1-1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.11.1.1.2 Procedure

- 1) SS transmits PDSCH via PDCCH DCI format 1A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.3.4.1-1. The SS sends downlink MAC padding bits on the DL RMC.
- 2) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.3.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 3) Set the Downlink signal level to the appropriate REFSENS value defined in table 4.2.12.1.2-1. Send continuously uplink power control "up" commands in the uplink scheduling information to the UE to ensure the UE transmits P_{UMAX} level for at least the duration of the Throughput measurement. (obtain correct UE output power as specified in ETSI TS 136 521-1 [1]).
- 4) Measure the average throughput for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].
- 5) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.3.

5.3.11.1.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.12.1.2 in order to show compliance.

5.3.11.2 Receiver Reference Sensitivity Level for Carrier Aggregation in DL-only bands

140

- 5.3.11.2.1 Method of test
- 5.3.11.2.1.1 Initial conditions

Test Environment: normal, TL/VL, TL/VH, TH/VL, TH/VH; as specified in annex B.

Frequencies to be tested: mid-range as defined ETSI TS 136 508 [2], clause 4.3.1, unless otherwise specified in the tables 7.3A.3.4.1-2 to 7.3A.3.4.1-29 in ETSI TS 136 521-1 [1].

Channel bandwidth to be tested: Highest $N_{RB_{agg}}$ as defined in ETSI TS 136 508 [2], clause 4.3.1, unless otherwise specified in tables 7.3A.3.4.1-2 to 7.3A.3.4.1-29 in ETSI TS 136 521-1 [1].

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors as shown in ETSI TS 136 508 [2] annex A, figure group A.32 as appropriate if the UE supports 2 Rx antennas or figure group A.80 if the UE supports 4 Rx antennas in the band.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals for PCC are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1, C.3.1, and uplink signals according to ETSI TS 136 521-1 [1], clauses H.1 and H.3.0.
- 4) The UL and DL Reference Measurement channels are set according to table 7.3A.3.4.1-1 in ETSI TS 136 521-1 [1].
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF according to ETSI TS 136 508 [2], clause 5.2A.2.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.11.2.2 Procedure

- 1) Configure SCC according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.1 for all downlink physical channels.
- 2) The SS shall configure SCC as per ETSI TS 136 508 [2], clause 5.2A.4.
- 3) SS activates SCC by sending the activation MAC-CE. Wait for at least 2 seconds.
- 4) SS transmits PDSCH via PDCCH DCI format 1A for C_RNTI to transmit the DL RMC according to table 7.3A.3.4.1-1 of ETSI TS 136 521-1 [1] on both PCC and SCC. The SS sends downlink MAC padding bits on the DL RMC.
- 5) SS sends uplink scheduling information for each UL HARQ process via PDCCH DCI format 0 for C_RNTI to schedule the UL RMC according to table 7.3A.3.4.1-1 of ETSI TS 136 521-1 [1] on PCC. Since the UE has no payload and no loopback data to send the UE sends uplink MAC padding bits on the UL RMC.
- 6) Set the Downlink signal level for PCC and SCC to the appropriate REFSENS value defined in table 4.2.12.2.2-1. Send continuously uplink power control "up" commands in the uplink scheduling information to the UE to ensure the UE transmits P_{UMAX} level for at least the duration of the Throughput measurement.
- 7) Measure the average throughput for each component carrier for a duration sufficient to achieve statistical significance according to ETSI TS 136 521-1 [1], clause G.2A.

5.3.11.2.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.12.2.2 in order to show compliance.

5.3.11.3 Receiver Reference Sensitivity Level for category NB1

5.3.11.3.1 Method of test

5.3.11.3.1.1 Initial conditions

Test Environment: normal, TL/VL, TL/VH, TH/VL, TH/VH; as specified in annex B.

Frequencies to be tested: Frequency ranges defined in ETSI TS 136 521-1 [1], clause K.1.2, see ETSI TS 136 508 [2], clause 4.3.1.

141

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors using only the main UE Tx/Rx antenna.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 8.1.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.0 and NPUSCH Format 2 is used to carry ACK/NACK on the uplink.
- 4) The DL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], table 7.3F.1.4.1-1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 2A-NB with CP CIoT Optimization according to ETSI TS 136 508 [2], clause 8.1.5.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.11.3.1.2 Procedure

- SS transmits NPDSCH via NPDCCH DCI format N1 for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.3F.1.4.1-1. The SS sends downlink MAC padding bits on the DL RMC. The UE will send the HARQ feedback based on information contained in DCI format N1.
- 2) Set the Downlink signal level to the REFSENS value defined in table 4.2.12.2.2-1.
- 3) Measure the average throughput for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.3F.1.4.

5.3.11.3.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.12.2.2 in order to show compliance.

5.3.11.4 Receiver Reference Sensitivity Level for UE category 0

5.3.11.4.1 Method of test

5.3.11.4.1.1 Initial conditions

Same initial conditions as in clause 5.3.11.1.1.1 with the following exceptions:

- Instead of table 7.3.4.1-1 in ETSI TS 136 521-1 [1] → use table 7.3E.4.1-1 in ETSI TS 136 521-1 [1].
- Connect the SS to the UE antenna connectors as shown in ETSI TS 136 508 [2] annex A, figure A.3 using only main UE Tx/Rx antenna.

NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.11.4.1.2 Procedure

Same test procedure as in clause 5.3.11.1.1.2 with the following exceptions:

• Instead of table 4.2.12.1.2-1 \rightarrow use tables 4.2.12.4.2-1 and 4.2.12.4.2-2.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.3E.

5.3.11.4.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.12.3.2 in order to show compliance.

5.3.11.5 Receiver Reference Sensitivity Level for UE category M1

5.3.11.5.1 Method of test

5.3.11.5.1.1 Initial conditions

Test Environment: normal, TL/VL, TL/VH, TH/VL, TH/VH; as specified in annex B.

Frequencies to be tested: low range, mid-range, high range see ETSI TS 136 508 [2], clause 4.3.1.

Channel bandwidth to be tested: 5 MHz channel bandwidth as defined in ETSI TS 136 508 [2], clause 4.3.1.

Uplink/Downlink configurations: as specified in ETSI TS 136 521-1 [1]:

- 1) Connect the SS to the UE antenna connectors as shown in ETSI TS 136 508 [2], annex A, figure A.3 using only main UE Tx/Rx antenna.
- 2) The parameter settings for the cell are set up according to ETSI TS 136 508 [2], clause 4.4.3.
- 3) Downlink signals are initially set up according to ETSI TS 136 521-1 [1], clauses C.0, C.1 and C.3.1 and uplink signals according to clauses H.1 and H.3.1.
- 4) The UL and DL Reference Measurement channels are set according to ETSI TS 136 521-1 [1], table 7.3EA.4.1-1.
- 5) Propagation conditions are set according to ETSI TS 136 521-1 [1], clause B.0.
- 6) Ensure the UE is in State 3A-RF-CE according to ETSI TS 136 508 [2], clause 5.2A.2AA.
- NOTE: When reference is made to test set up, call set up and test mode, guidance on the applicability of these can be found in ETSI TS 136 521-1 [1], ETSI TS 136 508 [2] and ETSI TS 136 509 [i.10] respectively.

5.3.11.5.1.2 Procedure

- SS transmits PDSCH via M-PDCCH DCI format 6-1A for C_RNTI to transmit the DL RMC according to ETSI TS 136 521-1 [1], table 7.3EA.4.1-1. The SS sends downlink MAC padding bits on the DL RMC. The SS sends one sided dynamic OCNG Pattern OP.1 FDD/TDD for the DL-signal as described in ETSI TS 136 521-1 [1], clauses A.5.1.1 and A.5.2.1.
- 2) SS sends uplink scheduling information for each UL HARQ process via MPDCCH DCI format 6-0A for C_RNTI to schedule the UL RMC according to ETSI TS 136 521-1 [1], table 7.3EA.4.1-1. Since the UE has no payload data to send, the UE transmits uplink MAC padding bits on the UL RMC.
- 3) Set the Downlink signal level to the appropriate REFSENS value defined in table 4.2.12.4.2-1 for FDD and TDD and in table 4.2.12.4.2-2 for HD-FDD. Send continuously uplink power control "up" commands in the uplink scheduling information to the UE to ensure the UE transmits P_{UMAX} level for at least the duration of the Throughput measurement.

4) Measure the average throughput for a duration sufficient to achieve statistical significance according to clause G.2 of ETSI TS 136 521-1 [1].

143

5) Repeat for applicable test frequencies, channel bandwidths and operating bands.

Details of the test method can be found in ETSI TS 136 521-1 [1], clause 7.3EA.

5.3.11.5.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.12.4.2 in order to show compliance.

5.3.12 Receiver Total Radiated Sensitivity (TRS)

5.3.12.1 Method of test

5.3.12.1.1 Initial conditions

Initial conditions are described in ETSI TS 137 544 [6], clause 7.1.5.4.1 for FDD and clause 7.1.6.4.1 for TDD systems.

5.3.12.1.2 Procedure

Procedure is described in ETSI TS 137 544 [6], clause 7.1.5.4.2 for FDD and clause 7.1.6.4.2 for TDD systems.

In case devices support adaptive features that dynamically tune the RF front end and adjust TX power for optimum performance in its region of operation, the device being measured should be representative of the device configuration used by a consumer in that region. This could include setting the MCC value or another parameter to one used within the region.

5.3.12.1.3 Procedure, reverberation chamber method

Refer to ETSI TS 137 544 [6], clause 7.1.5.4.3 for FDD and clause 7.1.6.4.3 for TDD systems.

5.3.12.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.13.2 in order to show compliance.

5.3.13 Total Radiated Power (TRP)

5.3.13.1 Method of test

5.3.13.1.1 Initial conditions

Initial conditions are described in ETSI TS 137 544 [6], clause 6.1.5.4.1 for FDD and clause 6.1.6.4.1. for TDD systems.

5.3.13.1.2 Procedure

Procedure is described in ETSI TS 137 544 [6], clause 6.1.5.4.2 for FDD and clause 6.1.6.4.2 for TDD systems.

In case devices support adaptive features that dynamically tune the RF front end and adjust TX power for optimum performance in its region of operation, the device being measured should be representative of the device configuration used by a consumer in that region. This could include setting the MCC value or another parameter to one used within the region.

For devices supporting transmit antenna switching using multiple TX antennas, the TRP should be measured for each transmit antenna individually. The antenna with the greater TRP should be used to determine the pass/fail compliance.

5.3.13.1.3 Procedure, reverberation chamber method

Refer to ETSI TS 137 544 [6], clause 6.1.5.4.3 for FDD and clause 6.1.6.4.3. for TDD systems.

5.3.13.2 Test requirements

The results obtained shall be compared to the limits in clause 4.2.14.2 in order to show compliance.
Annex A (informative): Relationship between the present document and the essential requirements of Directive 2014/53/EU

The present document has been prepared under the Commission's standardisation request C(2015) 5376 final [i.9] to provide one voluntary means of conforming to the essential requirements of Directive 2014/53/EU on the harmonisation of the laws of the Member States relating to the making available on the market of radio equipment and repealing Directive 1999/5/EC [i.2].

Once the present document is cited in the Official Journal of the European Union under that Directive, compliance with the normative clauses of the present document given in table A-1 confers, within the limits of the scope of the present document, a presumption of conformity with the corresponding essential requirements of that Directive and associated EFTA regulations.

Harmonised Standard ETSI EN 301 908-13						
Requirement				R	Requirement Conditionality	
No	Description	Essential requirements of Directive	Clause(s) of the present document	U/C	Condition	
1	Transmitter Maximum Output Power	3.2	4.2.2	U		
2	Transmitter Spectrum Emission Mask	3.2	4.2.3	U		
3	Transmitter Spurious Emissions	3.2	4.2.4	U		
4	Transmitter Minimum Output Power	3.2	4.2.5	U		
5	Receiver Adjacent Channel Selectivity (ACS)	3.2	4.2.6	U		
6	Receiver Blocking Characteristics	3.2	4.2.7	U		
7	Receiver Spurious Response	3.2	4.2.8	U		
8	Receiver Intermodulation Characteristic	3.2	4.2.9	U		
9	Receiver Spurious Emissions	3.2	4.2.10	U		
10	Transmitter Adjacent Channel Leakage Power Ratio	3.2	4.2.11	U		
11	Receiver Reference Sensitivity Level	3.2	4.2.12	U		
12	Receiver Total Radiated Sensitivity (TRS)	3.2	4.2.13	U		
13	Total Radiated Power (TRP)	3.2	4.2.14	U		

Table A-1: Relationship between the present document and the essential requirements of Directive 2014/53/EU

Key to columns:

Requirement:

No A unique identifier for one row of the table which may be used to identify a requirement.

Description A textual reference to the requirement.

Essential requirements of Directive

Identification of article(s) defining the requirement in the Directive.

Clause(s) of the present document

Identification of clause(s) defining the requirement in the present document unless another document is referenced explicitly.

Requirement Conditionality:

U/C Indicates whether the requirement is unconditionally applicable (U) or is conditional upon the manufacturer's claimed functionality of the equipment (C).

146

Condition Explains the conditions when the requirement is or is not applicable for a requirement which is classified "conditional".

Presumption of conformity stays valid only as long as a reference to the present document is maintained in the list published in the Official Journal of the European Union. Users of the present document should consult frequently the latest list published in the Official Journal of the European Union.

Other Union legislation may be applicable to the product(s) falling within the scope of the present document.

Annex B (normative): Environmental profile

B.1 General

B.1.1 Introduction

This annex specifies the environmental profile of the UE.

B.1.2 Temperature

The UE shall fulfil all the requirements in the full temperature range of:

Table B.1.2-1: Temperature Test Environment

147

+15 °C to +35 °C	For normal conditions (with relative humidity up to 75 %)
-10 °C to +55 °C	For extreme conditions (see IEC 60068-2-1 [4] and IEC 60068-2-2 [5])

The normative reference for this requirement is ETSI TS 136 101 [3], clause E.1.

Some tests are performed also in extreme temperature conditions. These test conditions are denoted as TL (Temperature Low, -10 $^{\circ}$ C) and TH (Temperature High, +55 $^{\circ}$ C).

B.1.3 Voltage

The UE shall fulfil all the requirements in the full voltage range, i.e. the voltage range between the extreme voltages.

For the equipment that can be operated from one or more of the power sources listed below, the lower extreme voltage shall not be higher and the higher extreme voltage shall not be lower than that specified below.

Power source	Lower extreme Voltage	Higher extreme voltage	Normal conditions voltage
AC mains	0,9 × nominal	$1,1 \times nominal$	nominal
Regulated lead acid battery	0,9 × nominal	$1,3 \times nominal$	1,1 × nominal
Non regulated batteries:			
Leclanché	$0,85 \times nominal$	Nominal	Nominal
Lithium	$0,95 \times nominal$	$1,1 \times Nominal$	1,1 × Nominal
Mercury/nickel and cadmium	0,90 imes nominal		Nominal

 Table B.1.3-1: Voltage Test Environment

Outside this voltage range, the UE if powered on, shall not make ineffective use of the radio frequency spectrum. In no case shall the UE exceed the transmitted levels as defined in ETSI TS 136 101 [3] for extreme operation. In particular, the UE shall inhibit all RF transmissions when the power supply voltage is below the manufacturer declared shutdown voltage.

The normative reference for this requirement is ETSI TS 136 101 [3], clause E.2.

Some tests are performed also in extreme voltage conditions. These test conditions are denoted as Lower extreme Voltage (VL) and Higher extreme Voltage (VH).

B.1.4 Test environment

Where a normal environment is required then the normal conditions shown in clauses B.1.2 and B.1.3 shall be applied.

148

Where an extreme environment is required then the various combinations of extreme temperatures together with the extreme voltages shown in clauses B.1.2 and B.1.3 shall be applied. The combinations are:

- Low extreme Temperature/Low extreme Voltage (TL/VL);
- Low extreme Temperature/High extreme Voltage (TL/VH);
- High extreme Temperature/Low extreme Voltage (TH/VL);
- High extreme Temperature/High extreme Voltage (TH/VH).

Annex C (informative): Maximum Measurement Uncertainty

The measurements described in the present document are based on the following assumptions:

- the measured value related to the corresponding limit is used to decide whether an equipment meets the requirements of the present document;
- the value of the measurement uncertainty for the measurement of each parameter is included in the test report.

Table C.1 shows the recommended values for the maximum measurement uncertainty figures.

Parameter	Uncertainty				
Transmitter maximum output power	±0,7 dB				
Transmitter spectrum emissions mask	±1,5 dB				
Transmitter spurious emissions	±2,0 dB				
9 kHz < f \leq 4 GHz	±4,0 dB				
4 GHz < f ≤ 12,75 GHz					
Transmitter Minimum output power	±1,0 dB				
Receiver Adjacent Channel Selectivity (ACS)	±1,1 dB				
Receiver Blocking characteristics	±1,3 dB				
1 MHz < f _{interferer} ≤ 3 GHz	±3,2 dB				
3 GHz < f _{interferer} ≤ 12,75 GHz					
Receiver spurious response	±1,3 dB				
1 MHz < f _{interferer} ≤ 3 GHz	±3,2 dB				
3 GHz < f _{interferer} ≤ 12,75 GHz					
Receiver intermodulation characteristics	±1,4 dB				
30 MHz ≤ f ≤ 4,0 GHz	±2,0 dB				
4 GHz < f ≤ 12,75 GHz	±4,0 dB				
Transmitter adjacent channel leakage power ratio	±0,8 dB				
Receiver Reference Sensitivity Level	±0,7 dB				
f ≤ 4,0 GHz	±1,0 dB				
4 GHz < f ≤ 12,75 GHz					
NOTE 1: For RF tests it should be noted that the	uncertainties in table C-1				
apply to the test system operating into a nominal 50 Ω load and					
do not include system effects due to mi	smatch between the EUI				
and the test system.					
NOTE 2: If the test system for a test is known to	nave a measurement				
uncertainty greater than that specified in table C-1, this					
equipment can still be used provided that an adjustment is made					
as follows.					
specified in table C-1 should be used to tighten the test					
requirements - making the test harder to pass (for some tests					
e.g. receiver tests, this may require modification of stimulus					
signals). This procedure will ensure that a test system not					
compliant with table C-1 does not increase the probability of					
passing an EUT that would otherwise have failed a test if a test					
system compliant with table C-1 had been used.					

Annex D (informative): Bibliography

- Directive 2004/108/EC of the European Parliament and of the Council of 15 December 2004 on the approximation of the laws of the Member States relating to electromagnetic compatibility and repealing Directive 89/336/EEC (EMC Directive).
- CEPT/ERC/REC 74-01 (Siófok 1998, Nice 1999, Sesimbra 2002, Hradec Kralove 2005, Cardiff 2011): "Unwanted Emissions in the Spurious Domain".
- Directive 2006/95/EC of the European Parliament and of the Council of 12 December 2006 on the harmonisation of the laws of Member States relating to electrical equipment designed for use within certain voltage limits (LV Directive).
- Commission Decision (EU) 2015/750 of 8 May 2015 on the harmonisation of the 1 452-1 492 MHz frequency band for terrestrial systems capable of providing electronic communications services in the Union.
- Directive 98/34/EC of the European Parliament and of the Council of 22 June 1998 laying down a procedure for the provision of information in the field of technical standards and regulations.
- Directive 98/48/EC of the European Parliament and of the Council of 20 July 1998 amending Directive 98/34/EC laying down a procedure for the provision of information in the field of technical standards and regulations.
- Commission Decision 2005/513/EC of 11 July 2005 on the harmonised use of radio spectrum in the 5 GHz frequency band for the implementation of wireless access systems including radio local area networks (WAS/RLANs).
- Commission Decision 2007/90/EC of 12 February 2007 amending Decision 2005/513/EC on the harmonised use of radio spectrum in the 5 GHz frequency band for the implementation of Wireless Access Systems including Radio Local Area Networks (WAS/RLANs).
- Commission Implementing Decision (EU) 2018/637 amending Decision 2009/766/EC on the harmonisation of the 900 MHz and 1800 MHz frequency bands for terrestrial systems capable of providing pan-European electronic communications services in the Community as regards relevant technical conditions for the Internet of Things.
- Commission Implementing <u>Decision 2012/688/EU</u> on the harmonisation of the frequency bands 1920-1980 MHz and 2110-2170 MHz for terrestrial systems capable of providing electronic communications services in the Union <u>Press Release</u>.
- ETSI TR 103 288 (V1.1.1) (05-2015): "Electromagnetic compatibility and Radio spectrum Matters (ERM); Report of the CENELEC/ETSI Joint Working Group in response to the EC letter ENTRP/F5/DP/MM/entr.f5.(2013)43164 to the ESOs".

Version	Information about changes
11.1.4	Stable draft with new features up to release 13 March 2018
11.1.7	Final draft for approval with resolved comments received from EC early assessment December 2018
11.1.8	Minor editorial change in 5.3.2.1.1 and Ready for second Around Assessment January 2019
11.1.9	Final draft with resolved comments from second around EC early assessment
11.1.10	Band 65 for NB1 feature is added and an error/typo for protecting emission from Band 28 to Band 32 in table 4.2.4.1.2-3 is corrected, July 2019
13.2.0	Included Over The Air (OTA) antenna performance requirements in terms of Receiver Total Radiated Sensitivity (TRS) and Total Radiated Power (TRP)

151

· ····································					
Document history					
V4.2.1	March 2010	Publication			
V5.2.1	May 2011	Publication			
V6.2.1	October 2013	Publication			
V7.1.1	December 2015	Publication			
V11.1.1	July 2016	Publication			
V11.1.2	July 2017	Publication			
V13.1.1	November 2019	Publication			
V13.2.1	February 2022	Publication			
V13.3.0	July 2024	SRdAP	EV 20241001:	03/07/2024 to 01/10/2024	

152

History